MINUTES BOX ELDER COUNTY COMMISSION JANUARY 05, 2022

The Board of County Commissioners of Box Elder County, Utah met in an Administrative / Operational Session at the Historic County Courthouse, 1 South Main Street in Brigham City, Utah at 4:45 p.m. on **January 05, 2022**. The following members were present:

Stan Summers	Chairman
Jeff Scott	Commissioner
Jeff Hadfield	Commissioner
Marla Young	Clerk

The following items were discussed:

- 1. Agenda
- 2. Commissioners' Correspondence
- 3. Staff Reports Agenda Related
- 4. Correspondence

The Administrative / Operational Session adjourned at 4:51 p.m.

The regular session was called to order by Chairman Summers at 5:00 p.m. with the following members present, constituting a quorum:

Stan Summers	Chairman
Jeff Scott	Commissioner
Jeff Hadfield	Commissioner
Marla Young	Clerk

The prayer was offered by Commissioner Scott.

The Pledge of Allegiance was led by Clerk Marla Young.

APPROVAL OF MINUTES

THE MINUTES OF THE REGULAR MEETING OF DECEMBER 15, 2021 WERE APPROVED AS WRITTEN ON A MOTION BY COMMISSIONER HADFIELD, SECONDED BY COMMISSIONER SCOTT, AND UNANIMOUSLY CARRIED.

ATTACHMENT NO. 1 - AGENDA

<u>ADMINISTRATIVE REVIEW / REPORTS / FUTURE AGENDA ITEMS - COMMISSION</u>

There were no Administrative Review Reports given.

FORMER AGENDA ITEMS, FOLLOW UP - COMMISSIONERS

There were no Former Agenda Items discussed.

EMERGENCY MANAGEMENT ISSUES

Commissioner Scott gave a brief update regarding Covid-19. He stated the numbers continue to rise with the omicron variant. He encouraged everyone to be careful and courteous to others.

ARPA

There were no ARPA items discussed.

COMMISSIONERS

2022 Commissioner Chair Appointment and Commissioner Assignments - Commissioners

The Commissioners are keeping the same assignments for 2022.

MOTION: Commissioner Hadfield made a motion to appoint Commissioner Scott as the 2022 Commission Chairman. The motion was seconded by Commissioner Summers and unanimously carried.

Chairman Scott conducted the remainder of the meeting.

Rocky Mountain Home Fund - Commissioner Summers

Steve Waldrip of Rocky Mountain Home Fund gave a presentation of their private program that helps in funding home ownership opportunities for professionals. They help with financing using a split equity program. He showed a video explaining their program and services.

<u>AT&T Thank You Letter For Expanding Connectivity in Our Communities - Commissioner Summers</u>

Commissioner Summers acknowledged the Commissioners received a nice letter from AT&T for their support in broadband services in the county.

Confirm New Health Department Appointment - Commissioner Scott

Chairman Scott thanked Chief Deputy Sheriff Dale Ward and Jim Abel for serving on the Health Department Board. They have served on the board for seven years. He recommended Cade Palmer and Kevin Hall be appointed to serve on the board.

MOTION: Commissioner Hadfield made a motion to appoint Cade Palmer and Kevin Hall to the Bear River Health Department Board. The motion was seconded by Commissioner Summers and unanimously carried.

PUBLIC INTERESTS / PRESENTATIONS / CONCERNS

Representation for Beaver Dam and Collinston on the Bear River Water Conservancy Board of Directors - Richard Garrett

Richard Garrett of Collinston requested to have a member from their community on the Bear River Water Conservancy District Board of Directors. He would like to have representation in decisions that affect their area.

Chairman Scott stated they would need to look at the bylaws of when the district was formed and see what it takes to add a board member.

Input on the Developing Agricultural Restrictions in Box Elder County - Laura Holmgren

Laura Holmgren and Junior Goring voiced concerns regarding the proposed ordinance that will set restrictions on agricultural industries. She encouraged the Commission not to act in haste as the county has a rich history of agriculture. The ag industry deserves due process and should be involved in decisions in the future.

Permission to Use Courthouse Grounds for Rosary - George Fuller

George Fuller thanked the Commissioners for their service and requested to use the courthouse grounds on January 22, 2022 for a public rally and prayer at 12:00 noon on the front steps of the courthouse. He stated approximately 40 people participate and it will last an estimated half hour.

MOTION: Commissioner Summers made a motion to allow the use of the courthouse grounds for the rosary and rally on January 22, 2022. The motion was seconded by Commissioner Hadfield and unanimously carried.

<u>Letter from the Box Elder County Farm Bureau Representing over 2000 Farm Bureau</u> <u>Members - Calee Garn</u>

Calee Garn representing Box Elder County Farm Bureau read a letter expressing unified opposition to the land use plan regarding SB130 and the ordinance that is to be passed by February 1st.

Commissioner Hadfield stated all the Commissioners have received calls and agriculture is the biggest thing in the county and they want to protect it.

Commissioner Summers stated they are hoping changes to the date will be made in the legislature on the bill.

Chairman Scott said he appreciated the input given as it shows people care. He is hopeful of changes being made from the legislature.

ATTACHMENT NO. 2 - Letter from Farm Bureau

AUDITOR'S OFFICE

Ag Lease Agreement #22-01 With Clark Land and Livestock LLC - Parcel 04-068-0016 - Tom Kotter

This item was canceled.

COMMUNITY DEVELOPMENT

Rodney and Lindy Jenson Subdivision - Scott Lyons

Community Development Director Scott Lyons explained the Rodney and Lindy Jensen Subdivision is a one-lot subdivision located at 10000 W 11444 N in the Bothwell area. It has been through all applicable department reviews and meets the minimum zoning requirements.

MOTION: Commissioner Hadfield made a motion to approve the Rodney and Lindy Jensen Subdivision. The motion was seconded by Commissioner Summers and unanimously carried.

Ordinance #543 - Rezone from A-20 ro RR-1 - Scott Lyons

Community Development Director Scott Lyons explained Ordinance #543 is regarding a rezone request of 8.96 acres located at 11200 N 3600 W. The original request was for the entire acreage and went through the Planning Commission. The Planning Commission recommended approval for smaller acreage. It is currently zoned A-20 and the request is to be changed to RR-1.

MOTION: Commissioner Summers made a motion to approve Ordinance #543. THe motion was seconded by Commissioner Hadfield and unanimously carried.

ATTACHMENT NO. 3 - ORDINANCE #543

Ordinance #548 Zoning Map Amendment from C-G to C-E - Scott Lyons

Community Development Director Scott Lyons explained Ordinance #548 regards a request for a zoning map amendment for 5.11 acres to be changed from the C-G zone to the C-E zone. The property is located at approximately 9000 S Hwy 89 in South Willard. A public hearing was held at the Planning Commission meeting and they have forwarded a recommendation of approval.

MOTION: Commissioner Hadfield made a motion to approve Ordinance #548. The motion was seconded by Commissioner Summers and unanimously carried.

ATTACHMENT NO. 5 - ORDINANCE #548

BUILDINGS & GROUNDS

Contract #22-02 for Renovation of the America First Building for DMV - Codey Illum

Chief Building Official Codey Illum stated they sent out for bid for the remodel of the DMV building. They received five bids and the low bid was from Patriot Construction.

MOTION: Commissioner Summers made a motion to award the bid for the DMV renovation to Patriot Construction. The motion was seconded by Commissioner Hadfield and unanimously carried.

ATTACHMENT NO. 5 - CONTRACT #22-02

HUMAN RESOURCES

Employee Early HSA Payment - Jenica Stander

Human Resource Manager Jenica Stander explained the early HSA payment request is for an employee that is having a medical hardship.

MOTION: Commissioner Hadfield made a motion to approve the early HSA distribution for an employee. The motion was seconded by Commissioner Summers and unanimously carried.

Policy 25 Update - Jenica Stander

Human Resource Manager Jenica Stander explained Policy 25 regards the covid pay. The last revision expired December 31, 2021 and asked to have it extended to June.

MOTION: Commissioner Summers made a motion to approve the updated Policy 25. The motion was seconded by Commissioner Hadfield and unanimously carried.

Section 5 Disciplinary Section Updates - Anne Hansen

Deputy Attorney Anne Hansen explained the update to Section 5 fixed some grammar issues and made changes to reprimand warnings.

MOTION: Commissioner Hadfield made a motion to approve the changes to Section 5. The motion was seconded by Commissioner Summers and unanimously carried.

<u>Policy to Comply with OSHA's Emergency Temporary Standard for Employers - Stephen</u> <u>Hadfield</u>

Attorney Stephen Hadfield explained that President Biden mandated vaccines or weekly testing for COVID, but there was a lawsuit filed and is currently being reviewed by the Supreme Court. It affects employers over a certain threshold and significant penalties are imposed for violation. He stated Resolution #22-01 addresses the requirements to comply with OSHA's emergency standards.

The Commissioners did not agree with the mandate and reluctantly decided upon the advice of the County Attorney to pass the policy to avoid the penalties.

MOTION: Commissioner Summers made a motion that under extreme duress the Commission approve Resolution #22-01 in protest. The motion was seconded by Commissioner Hadfield and unanimously carried.

ATTACHMENT NO. 6 - RESOLUTION #22-01

WARRANT REGISTER - COMMISSIONERS

The Warrant Register was signed and the following claims were approved. Claim numbers 116918 through 116957 in the amount of \$557,314.00 and claim numbers 116,875 through 116917 in the amount of \$1,334.738.11 with voided check numbers 116,709, 116770, and 116379. Claim numbers 116824 through 116874 in the amount of \$290,701.48 and claim numbers 116764 through 116768 in the amount of \$487,501.92.

PERSONNEL ACTIONS / VOLUNTEER ACTION FORMS - COMMISSIONERS

Employee Name:	Department:	PA Type:	Effective Date:
BUCKLEY, COLTON	SHERIFF'S OFFICE	PROMOTION	12/20/2021
WARD, JOHN KEVIN	SHERIFF'S OFFICE	PROMOTION	12/20/2021

			r
CLUFF, MICHELLE	SHERIFF'S OFFICE	SEPARATION	1/1/2022
HURD, RYAN	SHERIFF'S OFFICE	SEPARATION	12/31/2021
LOVELAND, RICKEY	SHERIFF'S OFFICE	SEPARATION	12/31/2021
MACFARLANE, TERRY	ROAD DEPARTMENT	SEPARATION	11/17/2021
CHRISTENSEN, KELSEY	BUILDINGS/GROUNDS	NEW HIRE	12/27/2021
THOMPSON, DOUGLAS	SHERIFF'S OFFICE	PROMOTION	12/20/2021
MILLER, ZACHARY	SHERIFF'S OFFICE	PROMOTION	12/20/2021
STICKNEY, WENDI	ATTORNEY'S OFFICE	SEPARATION	12/9/2021
WALLACE, KINLEY	ATTORNEY'S OFFICE	NEW HIRE	01/03/2022
MILLER, COLTON	LANDFILL	NEW HIRE 01/03/20	
JENSEN, COOPER	LANDFILL	NEW HIRE	01/03/2022
KREBS, JOSEPH	ASSESSOR'S OFFICE	COMPENSATION CHANGE	01/02/2022

WILDE, KORY	ASSESSOR'S OFFICE	COMPENSATION CHANGE	01/02/2022
COOK, KELLEN	LANDFILL	COMPENSATION CHANGE	01/02/2022
GUTIERREZ, DAMIEN	SHERIFF'S OFFICE	COMPENSATION CHANGE	01/29/2022
JENSEN, JARON	LANDFILL	COMPENSATION CHANGE	01/02/2022
LIMB, TRAVIS	LANDFILL	COMPENSATION CHANGE	01/02/2022
RISER, KARSON	LANDFILL	COMPENSATION CHANGE	01/02/2022
SATTERTHWAITE, CASEY	JUSTICE COURT	COMPENSATION CHANGE	12/07/2021
VAN TASSELL, ROMAN	LANDFILL	COMPENSATION CHANGE	01/02/2022
MYERS, CACHE	EMERGENCY MANAGEMENT	VOLUNTEER	01/05/2022
GABRIEL, TYLER	SHERIFF'S OFFICE	COMPENSATION CHANGE	12/25/2021

ANDERSON, MARK	WEED DEPARTMENT	COMPENSATION CHANGE	01/01/2022
KEENE, STEVEN K	EMERGENCY MANAGEMENT	VOLUNTEER	01/05/2022

CLOSED SESSION

There was not a closed session.

ADJOURNMENT

A motion was made by Commissioner Scott to adjourn. Commissioner Hadfield seconded the motion, and the meeting adjourned at 6:06 p.m.

ADOPTED AND APPROVED in regular session this 2nd day of February 2022.

Jen Scor, Chairman

Stan Summers, Commissioner

Jeff Hadfield, Commissioner

ATTEST:

10



COUNTY COMMISSION MEETING

Commission Chambers, 01 South Main Street, Brigham City, Utah 84302 Wednesday, January 05, 2022 at 5:00 PM

AGENDA

NOTICE: Public notice is hereby given that the Box Elder County Board of County Commissioners will hold an Administrative/Operational Session commencing at 4:45 P.M. and a regular Commission Meeting commencing at 5:00 P.M. on Wednesday January 5, 2022 in the Commission Chambers of the Box Elder County Courthouse, 01 South Main Street, Brigham City, Utah.

1. ADMINISTRATIVE / OPERATIONAL SESSION

- A. Agenda Review / Supporting Documents
- B. Commissioners' Correspondence
- C. Staff Reports

2. CALL TO ORDER 5:00 P.M.

- A. Invocation Given by: Commissioner Scott
- B. Pledge of Allegiance Given by: Clerk Marla Young
- C. Approval Of Minutes 12-15-2021

3. ADMINISTRATIVE REVIEW / REPORTS / FUTURE AGENDA ITEMS

- 4. FORMER AGENDA ITEMS
- 5. EMERGENCY MANAGEMENT ISSUES
- 6. ARPA

7. COMMISSIONERS

- A. 5:10 2022 Commission Chairman Appointment, and Commissioner Assignments-Commissioners
- B. 5:15 Rocky Mountain Home Fund-Commissioner Summers
- C. 5:20 Thank You Letter From AT&T to Expand Connectivity in our Communities-Commissioner Summers
- D. 5:22 Confirm New Health Department Appointment-Commissioner Scott

8. PUBLIC INTERESTS / PRESENTATIONS / CONCERNS

- A. 5:24 Representation for Beaver Dam and Collinston on the Bear River Conservancy Board of Directors-Richard Garrett
- B. 5:29 Input On the Developing Agricultural Restrictions In Box Elder County-Laura Holmgren
- C. 5:32 Permission to use Grounds for Public Square Rosary-George Fuller
- D. 5:34 Reading a Letter From Box Elder County Farm Bureau Representing over 2000 Farm Bureau Members-Calee Garn

9. AUDITOR'S OFFICE

A. 5:39 Ag Lease Agreement #22-01 with Clark Land & Livestock LLC- Parcel 04-068-0016-Tom Kotter

10. COMMUNITY DEVELOPMENT

- A. 5:41 Lindy and Rodney Jenson Subdivision-Scott Lyons
- B. 5:43 Ordinance #543- Rezone from A-20 to RR-1-Scott Lyons
- C. 5:45 Ordinance #548 Zoning Map Amendment from C-G to C-E-Scott Lyons

11. BUILDINGS & GROUNDS

A. 5:47 Contract #22-02 for Renovation of the America First Building for the DMV-Codey Illum

12. HUMAN RESOURCES

- A. 5:49 Employee Early H.S.A. Payment-Jenica Stander
- B. 5:51 Policy 25 Update-Jenica Stander
- C. 5:53 Section 5 Disciplinary Section Updates-Anne Hansen
- D. 5:55 Resolution # 22-01 to Comply with OSHA's Emergency Temporary Standards for Employers with 100 or More Employees-Steve Hadfield

13. WARRANT REGISTER

14. PERSONNEL ACTIONS / VOLUNTEER ACTION FORMS / CELL PHONE ALLOWANCE

- 15. CLOSED SESSION
- 16. ADJOURNMENT

Prepared and posted this 30th day of December, 2021. Mailed to the Box Elder News Journal and the Leader on the 30th of December, 2021. These assigned times may vary depending on the length of discussion, cancellation of scheduled agenda times and agenda alteration. Therefore, the times are estimates of agenda items to be discussed. If you have any interest in any topic you need to be in attendance at 5:00 p.m.

Marla R. Young - County Clerk

Box Elder County

NOTE: Please turn off or silence cell phones and pagers during public meetings. This facility is wheelchair accessible and accessible parking spaces are available. Requests for accommodations or interpretive services must be made three (3) working days prior to this meeting. Please contact the Commission Secretary's office at 734-3347 or FAX 734-2038 for information or assistance.



To: Commissioner Summers, Commissioner Hadfield, Commissioner Scott, Box Elder County Planning Department, and Box Elder County Planning Commission.

We, as members of the Utah Farm Bureau and Box Elder County Farm Bureau are writing to you to voice our unified opposition to any amendment to The Box Elder County Land Use Plan related to Animal Feeding Operations or Concentrated Animal Feeding Operations.

The proposal to amend Article 4 Chapter 4-5 related to Animal Feeding Operations and Concentrated Animal Feeding Operations (AFO/CAFO) presents detrimental consequences to the future of agriculture in Box Elder County. After conversations with the planning commission and Commissioner Stan Summers, this amendment is being quickly pushed through the county in response to Utah S.B. 130. Line 101-103 of this bill states, "the legislative body of a county desiring to restrict siting of large concentrated animal feeding operations shall adopt a county large concentrated animal feeding operation land use ordinance in accordance with this part by no later than February 1, 2022." We, as farmers and ranchers in Box Elder County DO NOT "desire restrictions" as we are already regulated on a state and federal level. Action is not required by the bill, nor is it desired by Box Elder County Residents.

Next, the proposed amendment acts contrary to what Box Elder County Residents have voiced. During 2021, the county participated in a process focusing on development priorities of the county. The overwhelming consensus of several surveys throughout the county was to preserve agriculture. Any changes relating to AFO/CAFO restrictions acts in opposition to the counties consensus. Furthermore, approximately 100 farmers and ranchers gathered at the planning commission meeting where the proposal was discussed on December 16, 2021. Of those in attendance, 40+ stood and for public record voiced their opposition to additional restrictions being put in place. We, as farmers and ranchers in Box Elder County do not support restrictions of any kind being placed on AFO/CAFO.

Finally, the proposal to restrict AFO/CAFO in Box Elder County is a death sentence to agriculture in our communities. If the proposal allows current operations to be "grandfathered in" eventually, these operations will cease to exist. Restrictions will prevent growth for those grandfathered in and will abolish any opportunity for new agriculture opportunities within the county. On paper, this proposal impacts only those involved in feeding livestock, but in reality, the proposal will impact all sectors of agriculture within the county.

In conclusion, we, as members of Farm Bureau cannot support restrictions being placed on the future of agriculture in Box Elder County.

Sincerely,

Chair

North Box Elder

Farm Bureau

Chair

South Box Elder

Farm Bureau

State Board

Riggin Holmgrem

Member

Utah Farm Bureau

2,393 Box Elder

County Farm

Bureau Members

CONTRACTOR'S AGREEMENT

FOR: Box Elder County America First Building DMV Remodel

WITNESSETH: WHEREAS, the **Owner** intends to have Work performed at 35 East 100 South, Brigham City , UT 84302

WHEREAS, Contractor agrees to perform the Work for the sum stated herein.

NOW, THEREFORE, Owner and Contractor for the consideration provided in this Contractor's Agreement, agree as follows:

ARTICLE 1. SCOPE OF WORK. The Work to be performed shall be in accordance with:

- A. The Bid Proposal prepared by Patriot Construction, LLC dated **December 15, 2021**, attached hereto.
- B. The Bidder Instructions provided to the Contractor by the Box Elder County Facilities Management Division, attached hereto.
- C. The following general scope:

 Demolition and remodel of existing building (See Bid Schedule and Project Drawings).

The Box Elder County Facilities Management Division General Conditions ("General Conditions") dated February 2021 and all Supplemental General Conditions ("also referred to as General Conditions") on file at the office of Box Elder County, are hereby incorporated by reference as part of this Agreement and are included in the specifications for this Project. All terms used in this Contractor's Agreement shall be as defined in the Contract Documents referenced in **Article 1** and defined in **Article 4** of this agreement, and in particular, the General Conditions.

The Contractor Agrees to furnish labor, materials and equipment to complete the Work as required in the Contract Documents which are hereby incorporated by reference. It is understood and agreed by the parties hereto that all Work shall be performed as required in the Contract Documents and shall be subject to inspection and approval of the Owner or its authorized representative. The relationship of the Contractor to the Owner hereunder is that of an independent Contractor.

ARTICLE 2. CONTRACT SUM. The Owner agrees to pay and the Contractor agrees to accept in full performance of this Contractor's Agreement, the sum of **Five Hundred and One Thousand Five Hundred and Seventy Four Dollars and Ninety-Four Cents (\$501,574.94)**, which is the base bid, and which sum also includes the cost of a 100% Performance Bond and a 100% Payment Bond as well as all insurance requirements of the Contractor. Said bonds have already been posted by the Contractor pursuant to

State law. The required proof of insurance certificates have been delivered to the Owner in accordance with the General Conditions before the execution of this Contractor's Agreement.

ARTICLE 3. TIME OF COMPLETION AND DELAY REMEDY. The Work shall be Substantially Complete by May 1, 2022. Contractor agrees to pay liquidated damages in the amount of \$500 per day for each day after expiration of the Contract Time until the Contractor achieves Substantial Completion in accordance with the Contract Documents, if Contractor's delay makes the damages applicable. The provision for liquidated damages is: (a) to compensate the Owner for delay only; (b) is provided for herein because actual damages cannot be readily ascertained at the time of execution of this Contractor's Agreement; (c) is not a penalty; and (d) shall not prevent the Owner from maintaining Claims for other non-delay damages, such as costs to complete or remedy defective Work.

No action shall be maintained by the Contractor, including its or Subcontractor or suppliers at any tier, against the Owner for damages or other claims due to losses attributable to hindrances or delays from any cause whatsoever, including acts and omissions of the Owner or its officers, employees or agents, except as expressly provided in the General Conditions. The Contractor may receive a written extension of time, signed by the Owner, in which to complete the Work under this Contractor's Agreement in accordance with the General Conditions.

ARTICLE 4. CONTRACT DOCUMENTS. The Contract Documents consist of this Contractor's Agreement, the Conditions of the Contract (Documents referenced in Article 1, Box Elder County Facilities Management Division General Conditions, Supplementary and other Conditions), the Drawings, Specifications, Addenda and Modifications. The Contract Documents shall also include the bidding documents, including the Invitation to Bid, Instructions to Bidders/Proposers and the Bid/Proposal, to the extent not in conflict therewith and other documents and oral presentations that are documented as an attachment to the contract.

All such documents are hereby incorporated by reference herein. Any reference in this Contractor's Agreement to certain provisions of the Contract Documents shall in no way be construed as to lessen the importance or applicability of any other provisions of the Contract Documents.

ARTICLE 5. PAYMENT. The Owner agrees to pay the Contractor from time to time as the Work progresses, but not more than once each month after the date of Notice to Proceed, and only upon Certificate of an authorized representative of the Box Elder County Facilities Management Division for Work performed during the preceding calendar month, ninety-five percent (95%) of the value of the labor performed and ninety-five percent (95%) of the value of materials furnished in place or on the site. The Contractor agrees to furnish to the Owner invoices for materials purchased and on the site but not installed, for which the Contractor requests payment and agrees to safeguard and protect such equipment or materials and is responsible for safekeeping thereof and if such be stolen, lost or destroyed, to replace same.

Such evidence of labor performed and materials furnished as the Owner may reasonably require shall be supplied by the Contractor at the time of request for Certificate of Payment on account. Materials for which payment has been made cannot be removed from the job site without Owner's written approval. Five percent (5%) of the earned amount shall be retained from each monthly payment. The retainage, including any additional retainage imposed and the release of any retainage, shall be in accordance with current Utah State Code. Contractor shall also comply with the requirements of said code, including restrictions of retainage regarding subcontractors and the distribution of interest earned on the

retention proceeds. The Owner shall not be responsible for enforcing the Contractor's obligations under State law in fulfilling the retention law requirements with subcontractors at any tier.

ARTICLE 6. INDEBTEDNESS. Before final payment is made, the Contractor must submit evidence satisfactory to the Owner that all payrolls, materials bills, subcontracts at any tier and outstanding indebtedness in connection with the Work have been properly paid. Final Payment will be made after receipt of said evidence, final acceptance of the Work by the Owner as well as compliance with the applicable provisions of the General Conditions.

Contractor shall respond immediately to any inquiry in writing by Owner as to any concern of financial responsibility and Owner reserves the right to request any waivers, releases or bonds from Contractor in regard to any rights of Subcontractors (including suppliers) at any tier or any third parties prior to any payment by Owner to Contractor.

ARTICLE 7. ADDITIONAL WORK. It is understood and agreed by the parties hereto that no money will be paid to the Contractor for additional labor or materials furnished unless a new contract in writing or a Modification hereof in accordance with the General Conditions and the Contract Documents for such additional labor or materials has been executed. The Owner specifically reserves the right to modify or amend this Contractor's Agreement and the total sum due hereunder either by enlarging or restricting the scope of the Work.

ARTICLE 8. INSPECTIONS. The Work shall be inspected for acceptance in accordance with the General Conditions.

ARTICLE 9. DISPUTES. Any dispute, Preliminary Resolution Effort or Claim between the parties shall be subject to the provisions of Article 7 of the General Conditions. Owner reserves all rights to pursue its rights and remedies as provided in the General Conditions.

ARTICLE 10. TERMINATION, SUSPENSION OR ABANDONMENT. This Contractor's Agreement may be terminated, suspended or abandoned in accordance with the General Conditions.

ARTICLE 11. OWNER'S RIGHT TO WITHHOLD CERTAIN AMOUNT AND MAKE USE THEREOF. The Owner may withhold from payment to the Contractor such amount as, in Owner's judgment, may be necessary to pay just claims against the Contractor or Subcontractor at any tier for labor and services rendered and materials furnished in and about the Work. The Owner may apply such withheld amounts for the payment of such claims in Owner's discretion. In so doing, the Owner shall be deemed the agent of Contractor and payment so made by the Owner shall be considered as payment made under this Contractor's Agreement by the Owner to the Contractor. Owner shall not be liable to the Contractor for any such payment made in good faith. Such withholdings and payments may be made without prior approval of the Contractor and may be also prior to any determination as a result of any dispute, Preliminary Resolution Effort, Claim or litigation.

ARTICLE 12. INDEMNIFICATION. The Contractor shall comply with the indemnification provisions of the General Conditions.

ARTICLE 13. SUCCESSORS AND ASSIGNMENT OF CONTRACT. The Owner and Contractor, respectively bind themselves, their partners, successors, assigns and legal representatives to the other party to this Agreement, and to partners, successors, assigns and legal representatives of such other party with

respect to all covenants, provisions, rights and responsibilities of this Contractor's Agreement. The Contractor shall not assign this Contractor's Agreement without the prior written consent of the Owner, nor shall the Contractor assign any moneys due or to become due as well as any rights under this Contractor's Agreement, without prior written consent of the Owner.

ARTICLE 14. RELATIONSHIP OF THE PARTIES. The Contractor accepts the relationship of trust and confidence established by this Contractor's Agreement and covenants with the Owner to cooperate with the Owner and its representatives and use the Contractor's best skill, efforts and judgment in furthering the interest of the Owner; to furnish efficient business administration and supervision; to make best efforts to furnish at all times an adequate supply of workers and materials; and to perform the Work in the best and most expeditious and economic manner consistent with the interests of the Owner.

ARTICLE 15. AUTHORITY TO EXECUTE AND PERFORM AGREEMENT. Contractor and Owner each represent that the execution of this Contractor's Agreement and the performance thereunder is within their respective duly authorized powers.

ARTICLE 16. ATTORNEY FEES AND COSTS. Except as otherwise provided in the dispute resolution provisions of the General Conditions, the prevailing party shall be entitled to reasonable attorney fees and costs incurred in any action in the District Court and/or appellate body to enforce this Contractor's Agreement or recover damages or any other action as a result of a breach thereof.

IN WITNESS WHEREOF, the parties hereto have executed this Contractor's Agreement on the day and year stated hereinabove.

CONTRACTOR:		
	Signature	Date
	Title:	
itate of)) ss.		
County of)	Please type/print name clearly	
On this day of, 2021, po		
vhose identity is personally known to me (or pr vho by me duly sworn (or affirmed), did say tha		•
office) of the firm and that said document was s		
	Notary Public	
SEAL)	My Commission Expires	
WNER:	Signature of Mary	1-5-2022 Date
tate of <u>Mah</u>)	Title: Commission (Jeffran D. Scot)	Chairma.
ounty of Box Elder) ss.	Please type/print name clearly	/
on this <u>5</u> day of <u>January</u> 202 2 , perhose identity is personally known to me (or provided by me duly sworn (or affirmed), did say that fire) of the firm and that said document was s	oved to me on the basis of satisfactory e t he (she) is the <i>Commission Che</i>	evidence) and airmaktitle or
MARLA YOUNG Notary Public State of Utah My Commission Expires October 15, Commission Number 721029	Notary Public My Commission Expires	ung 1. 15 2026

Bidder Name Patriot Construction, LLC

DOCUMENT 00 41 23 BID FORM

AMERICA FIRST BUILDING DMV REMODEL

ARTICLE 1 - BID RECIPIENT

1.01 This Bid is submitted via email to seths@jonescivil.com, who receives it on behalf of:

Box Elder County 1 South Main Street Brigham City, UT 84302

1.02 The undersigned Bidder proposes and agrees, if this Bid is accepted, to enter into a Contract with Owner in the form included in the Bidding Documents to perform all Work as specified or indicated in the Bidding Documents for the prices and within the times indicated in this Bid and in accordance with the other terms and conditions of the Bidding Documents.

ARTICLE 2 - BIDDER'S ACKNOWLEDGEMENTS

2.01 Bidder accepts all of the terms and conditions of the Instructions to Bidders, including without limitation those dealing with the disposition of Bid security. This Bid will remain subject to acceptance for 30 days after the Bid opening, or for such longer period of time that Bidder may agree to in writing upon request of Owner.

ARTICLE 3 - BIDDER'S REPRESENTATIONS

- 3.01 In submitting this Bid, Bidder represents that:
 - A. Bidder has examined and carefully studied the Bidding Documents, and any data and reference items identified in the Bidding Documents, and hereby acknowledges receipt of the following Addenda:

Addendum No.	Addendum, Date	
1	12/1/21	
2	12/14/21	
3	12/15/21	

- B. Bidder has visited the Site, conducted a thorough, alert visual examination of the Site and adjacent areas, and become familiar with and satisfied itself as to the general, local, and Site conditions that may affect cost, progress, and performance of the Work.
- C. Bidder is familiar with and has satisfied itself as to all Laws and Regulations that may affect cost, progress, and performance of the Work.
- D. Bidder has considered the information known to Bidder itself; information commonly known to contractors doing business in the locality of the Site; information and observations

- obtained from visits to the Site; the Bidding Documents; and any Site-related reports and drawings identified in the Bidding Documents, with respect to the effect of such information, observations, and documents on (1) the cost, progress, and performance of the Work; (2) the means, methods, techniques, sequences, and procedures of construction to be employed by Bidder; and (3) Bidder's safety precautions and programs.
- E. Bidder agrees, based on the information and observations referred to in the preceding paragraph, that no further examinations, investigations, explorations, tests, studies, or data are necessary for the determination of this Bid for performance of the Work at the price bid and within the times required, and in accordance with the other terms and conditions of the Bidding Documents.
- F. Bidder is aware of the general nature of work to be performed by Owner and others at the Site that relates to the Work as indicated in the Bidding Documents.
- G. Bidder has given Engineer written notice of all conflicts, errors, ambiguities, or discrepancies that Bidder has discovered in the Bidding Documents, and confirms that the written resolution thereof by Engineer is acceptable to Bidder.
- H. The Bidding Documents are generally sufficient to indicate and convey understanding of all terms and conditions for the performance and furnishing of the Work.
- I. The submission of this Bid constitutes an incontrovertible representation by Bidder that Bidder has complied with every requirement of this Article, and that without exception the Bid and all prices in the Bid are premised upon performing and furnishing the Work required by the Bidding Documents.

ARTICLE 4 - BIDDER'S CERTIFICATION

4.01 Bidder certifies that:

- A. This Bid is genuine and not made in the interest of or on behalf of any undisclosed individual or entity and is not submitted in conformity with any collusive agreement or rules of any group, association, organization, or corporation;
- Bidder has not directly or indirectly induced or solicited any other Bidder to submit a false or sham Bid;
- C. Bidder has not solicited or induced any individual or entity to refrain from bidding; and
- D. Bidder has not engaged in corrupt, fraudulent, collusive, or coercive practices in competing for the Contract. For the purposes of this Paragraph 4.01.D:
 - 1. "corrupt practice" means the offering, giving, receiving, or soliciting of anything of value likely to influence the action of a public official in the bidding process;
 - 2. "fraudulent practice" means an intentional misrepresentation of facts made (a) to influence the bidding process to the detriment of Owner, (b) to establish bid prices at artificial non-competitive levels, or (c) to deprive Owner of the benefits of free and open competition;
 - 3. "collusive practice" means a scheme or arrangement between two or more Bidders, with or without the knowledge of Owner, a purpose of which is to establish bid prices at artificial, non-competitive levels; and

4. "coercive practice" means harming or threatening to harm, directly or indirectly, persons or their property to influence their participation in the bidding process or affect the e execution of the Contract.

BASIS OF BID

4.02 Bidder will complete the Work in accordance with the Contract Documents for the following price(s):

Bid Schedule on the following page(s)

BID SCHEDULE

	Description	Qty.	Units	Unit Price	Amount
Land	dscape				
1	Sprinkler repair & rock scape with shrubs	1	LS	10,1637.00	10,637.00
2	Concrete flatwork repair	200	sf	8 60	1,720.00
Arch	itectural				.
3	Demolition	6,700	sf	3.13	20971.00
4	Carpentry	6,700	sf	15.94	106,798.00
5	Flooring	4,700	sf	6.42	30,174.00
6	DMV / BRAG Outdoor Programmable	1	LS	12,340.00	12,340.00
	Message LED Display Sign Screen				
7	Panel Billboard Ceiling	3,300	sf	4.00	13, 200 .00
8	Millwork (Cabinets)	150	If	253.00	37, 950.00
9	Transaction window	5	ea	5,606.00	28,030.00
10	Doors and hardware	21	ea	946.90	19,884.90
11	Paint Interior	6,699	sf	00	4,421,34
12	Paint Exterior	31,000	sf	٠ الما	20,460.00
13	Fiber cement vertical panel	800	sf	7.20	6, 160 00
14	Standing seam metal roof	500	sf	27.50	13,750.00
Struc	ctural			,	#
15	Commercial ADA passenger elevator	1	LS	59,018 00	59,018.00
16	Standard size brick	260	sf	49.6	12,766.00
17	CMU block (elevator shaft)	1	LS	42,790.00	42,790.00
18	Footings (elevator)	1	LS	2,81600	2,816.00
19	Excavation (elevator)	1	LS	7,202.00	7,202.00
20	Saw cut opening for doorway	4	ea	1,650.00	6,600.00
Mech	nanical				
21	Demo and replace ductwork	300	lf	25.00	7,500.00
22	Ceiling supply 3-zone system unit	2	ea	7,735.00	15,470.00
23	Ceiling supply (24"x 24")	17	ea	125.00	2,125.00
24	Ceiling return perforated (24"x24")	5	ea	125.00	625.00
25	Ceiling exhaust system complete	4	ea	V80.00	2,720.50
6	Repair pneumatic diverter operator unit	1	LS	2,500.00	2,500.00

Plum	bing				40
27	Demo and install pex pipe	2,000	lf	16.65	31,300.00
28	Fire sprinklers heads	1,100	sf	3.74	4,114.00
29	Wall-hung lavatory (ADA)	4	ea	L25.00	2,500.00
30	Elongated one-piece toilet (ADA)	5	ea	580.00	2,900.00
31	Double basin stainless steel kitchen	2	ea	680.00	1,360.60
32	sink Wall mount drinking fountain and bottle filler (ADA)	1	ea	5,800.00	5,800.00
33	Specialties items	1	LS	2,850.00	2,850.00
34	Wall mount urinal	1	ea	580.00	5.80.00
Elect	rical				
35	Interior lighting fixtures	80	ea	198.13	15,850.40
36	Exterior lighting fixtures	27	ea	252.60	6,820.20
37	Fire alarm systems	1	LS	13,200.00	13,200.00
38	Conduit, wiring and fittings	3,000	lf	16.54	49,620.00
39	Data communications and networking	3,000	lf	8.60	25,800.00
40	Disconnects	2	ea	207.50	415.00
41	GFI receptacle	80	ea	34.25	2,740.00
42	Panelboard	1	ea	1,280.00	1,280.00
43	Pneumatic diverter control box and	1	ea	2,750.00	2,750.00
	power supply			\$ 658,507.84	

Bidder acknowledges that (1) each Bid Unit Price includes an amount considered by Bidder to be adequate to cover Contractor's overhead and profit for each separately identified item, and (2) estimated quantities are not guaranteed, and are solely for the purpose of comparison of Bids, and final payment for all unit price Bid items will be based on actual quantities, determined as provided in the Contract Documents.

ARTICLE 5 - TIME OF COMPLETION

- 5.01 Bidder agrees that the Work will be substantially complete and will be completed and ready for final payment in accordance with Article 4 of the Contract on or before the dates or within the number of calendar days indicated in the Contract.
- 5.02 Bidder accepts the provisions of the Contract as to liquidated damages.

ARTICLE 6 – ATTACHMENTS TO THIS BID

- 6.01 The following documents are submitted with and made a condition of this Bid:
 - A. Required Bid security;
 - B. List of Proposed Subcontractors (see Document 00 43 36 for form);
 - C. Copy of current business license;
 - D. Copy of current Utah contractor's license; and
 - E. E-Verify Form
- 6.02 The following documents shall be submitted upon request and made a condition of this Bid:
 - A. List of Proposed Suppliers;
 - B. List of Project References; and
 - C. Required Bidder Qualification Statement with supporting data, upon request.

Standard forms will be provided.

ARTICLE 7 – DEFINED TERMS

7.01 The terms used in this Bid with initial capital letters have the meanings stated in the Instructions to Bidders, and the Contract.

[CONTINUED ON NEXT PAGE]

ARTICLE 8 - BID SUBMITTAL

BIDDER: [Indicate	correct	name	of bidding	entity]
-------------------	---------	------	------------	---------

Patriot Construction, LCO
By: [Signature]
[Printed name] Pall Groberg (If Bidder is a corporation, a limited liability company, a partnership, or a joint venture, attach evidence of authority to sign.)
Attest: [Signature]
[Printed name] Kaylee Lee
Title: Office Manager
Submittal Date: 12 [15 [2]
Address for giving notices:
7485 N 3600 W
Honeyville, VT 84314
Telephone Number: 811-731-5240 (11: 436-237-959)
Fax Number: 888 - 867 - 1845
Contact Name and e-mail address: Paul Groberg
office@patriotcon.net
Bidder's License No.: S178262-5501 (where applicable)

END OF BID FORM

SECTION 00 43 36 LIST OF PROPOSED SUBCONTRACTORS

Subcontractor Company Name and Contact Person	Type of Work to be Performed	Estimated Percentage of Work
Thompson & son	Labinek	
Uintah Drywall	Francing Drywall	
Tremonton Heating	HVAC	
Strive Co	Plymbing	
Flooring Services	Flooring	
RKC	rooting	
Nicholls Bro Painting	Painting	
Alble Access Elector	Elevator	

Additional information to be provided upon Owner's request.

Trade mark Musonry	Musoary
Larry's Kle	Tile
Luck W. Elebertical	Electric
Deno Patriot	Demo
Partriot	Framing
Patriot	Concrete
Patriot	excavation
landscaping	Scenic solution

Private Employer Affidavit of Compliance Pursuant To Utah Code 63G-12-302

By executing this affidavit, the undersigned private employer verifies its compliance with Utah Code 63G-12-302, stating affirmatively that the individual, firm or corporation has registered with and utilizes the federal work authorization program commonly known as E-Verify, or other authorized Status Verification System, in accordance with the applicable provisions and deadlines established in Utah Code. Furthermore, the undersigned private employer hereby attests that its federal work authorization user identification number and date of authorization are as follows:

14-1843144 Federal Tax ID#
Name of Private Employer on File with E-Verify
I hereby declare under penalty of perjury that the foregoing is true and correct. I also acknowledge that the company will indemnify Box Elder County from all fines, penalties, and costs associated with the company's non-compliance with Utah Code 63G-12-302. Signature of Authorized Officer or Agent Date
Printed Name and Title of Authorized Officer or Agent
STATE OF UTAH) :SS County of Box Elder)
On December, 15, 2021, personally appeared before me Paul Groberg the signer of the within instrument, who duly
the signer of the within instrument, who duly acknowledged to me that she executed the same.





STATE OF UTAH DEPARTMENT OF COMMERCE DIVISION OF OCCUPATIONAL & PROFESSIONAL LICENSING

ACTIVE LICENSE

EFFECTIVE DATE:

10/07/2002

EXPIRATION DATE:

11/30/2023

ISSUED TO:

PATRIOT CONSTRUCTION, LLC

7485 N 3600 W

HONEYVILLE UT 84314



REFERENCE NUMBER(S), CLASSIFICATION(S) & DETAIL(S)

5178262-5501

Contractor With LRF

DBAs: None Associated

B100, E100, S350

Form #1

BOX ELDER COUNTY
BUSINESS LICENSE MMVI—

STATE OF UTAH NO. 2021 - 5

THE SUM OF TWENTY DOLLARS AND HAVING COMPLIED WITH THE ORDINANCE IN FORCE RELATING TO LICENSE, IS HEREBY AUTHORIZED TO TRANSACT THE BUSINESS OF Paul Groberghaving Paid INTO THE CITY TREASURY

PATRIOT CONSTRUCTION LLC AT

7485 N 3600 W HONEYVILLE, UT-84314 FOR THE TERM OF TWELVE MONTHS, ENDING ON DECEMBER 31, 2021,

N TESTIMONY WHEREOF, I, BOYD BINGHAM, MAYOR OF HONEYVILLE CITY, HAVE HEREUNTO SET MY HAND THIS 5th DAY OF JANUARY 2021

HONEYVILLE CITY MAYOR

Stating / harger HONEYVILLE CITY RECORDER

BOX ELDER COUNTY FACILITIES MANAGEMENT DIVISION

GENERAL CONDITIONS

FEBRUARY 2021

Table of Contents

ARTICLE 1. GENERAL PROVISIONS

1.1 Basic Definitions.

ARTICLE 2. OWNER

- 2.1 Information and Services Required of the Owner.
 - 2.1.1 Owner's Representative.
 - 2.1.2 Specialists and Inspectors.
 - 2.1.3 Surveys and Legal Description.
 - 2.1.4 Prompt Information and Services.
 - 2.1.5 Copies of Drawings and Project Manuals (For Construction).
 - 2.1.6 Other Duties.
- 2.2 Construction By Owner or By Separate Contractors
 - 2.2.1 Owner's Right To Perform Construction and to Award Separate Contracts.
 - (1) In General.
 - (2) Coordination and Revisions.
 - 2.2.2 Mutual Responsibility.
 - (1) Contractor Coordination.
 - (2) Reporting Problems to Owner.
 - (3) Costs.
 - (4) Contractor Remedial Work.

ARTICLE 3. A/E

- 3.1 A/E's Administration of the Contract.
 - 3.1.1 In General.
 - 3.1.2 Site Visits.
 - 3.1.3 Communications Facilitating Contract Administration.
 - 3.1.4 A/E May Reject Work, Order Inspection, Tests.
 - 3.1.5 A/E Review Contractor's Submittals.

3.2 Ownership and Use of A/E's Drawings, Specifications and Other Documents.

ARTICLE 4. CONTRACTOR

- 4.1 Review of Contract Documents and Field Conditions by Contractor.
 - 4.1.1 Reviewing Contract Documents, Information, Reporting Errors, Inconsistencies or Omissions.
 - 4.1.2 Field Conditions.
 - 4.1.3 Perform in Accordance with Contract Documents and Submittals.
 - 4.1.4 Performance to Produce the Complete System and Intended Results.
 - 4.1.5 Intent and Hierarchy.
 - 4.1.6 Dividing Work and Contractor Representation.
 - 4.1.7 Planning and Priority.
- 4.2 Supervision and Construction Procedures.
 - 4.2.1 Supervision and Control.
 - 4.2.2 Responsibility.
 - 4.2.3 Not Relieved of Obligations.
 - 4.2.4 Inspections and Approvals.
- 4.3 Labor and Materials.
 - 4.3.1 Payment by Contractor.
 - 4.3.2 Discipline and Competence.
- 4.4 Taxes and Other Payments to Government.
- 4.5 Permits, Fees, Notices, Labor and Materials.
 - 4.5.1 Permits and Fees.
 - 4.5.2 Compliance With Public Authorities, Notices.
 - 4.5.3 Correlation of Contract Documents and Enactments.
 - 4.5.4 Failure to Give Notice.
- 4.6 Superintendent.
- 4.7 Time and Contractor's Construction Schedules.

- 4.7.1 Progress and Completion.
 - (1) Time is of the Essence; Complete Within Contract Time.
 - (2) Notice to Proceed and Insurance.
- 4.7.2 Schedule Preparation.
- 4.7.3 Initial Contract Time.
- 4.7.4 Interim Completion Dates and Milestones.
- 4.7.5 Schedule Content Requirements.
- 4.7.6 Owner's Right to Take Exceptions.
- 4.7.7 Float Time.
- 4.7.8 Initial Schedule Submission.
- 4.7.9 Updates.
- 4.7.10 Schedule of Submittals.
- 4.7.11 Schedule Recovery.
- 4.7.12 Schedule Changes and Modifications.
 - (1) Contract Time Change Requires Modification.
 - (2) Contractor Reordering, Resequencing and Changing Activity Durations.
 - (3) Changes in Contract Time.
- 4.7.13 Excusable Delay.
 - (1) In General.
 - (2) Weather-Related Excusable Delays.
- 4.7.14 Compensable Delay, Suspension or Interruption.
 - (1) Basic Conditions.
 - (2) Compensable Delay Formula.
 - (3) Period of Compensable Delay, Suspension or Interruption
 - (4) Concurrent Delay.
- 4.7.15 Time Extension Requests.
- 4.7.16 Liquidated Damages.
 - (1) In General.
 - (2) No Waiver of Owner's Rights.
- 4.8 Documents and Samples at the Site, Certifying "As-Builts".
- 4.9 Shop Drawings, Product Data and Samples.
 - 4.9.1 Not Contract Documents.
 - 4.9.2 Promptness.
 - 4.9.3 Not Perform Until A/E Approves.
 - 4.9.4 Representations by Contractor.
 - 4.9.5 Contractor's Liability.
 - 4.9.6 Direct Specific Attention to Revisions.
 - 4.9.7 Informational Submittals.
 - 4.9.8 Reliance on Professional Certification.
- 4.10Use of Site.
 - 4.10.1 In General.
 - 4.10.2 Access to Neighboring Properties.
- 4.11Access to Work.
- 4.12Royalties and Patents.
- 4.13Indemnification.

4.13.1 In General.

ARTICLE 5. SUBCONTRACTORS

- 5.1 Award of Subcontracts and Other Contracts for Portions of the Work.
 - 5.1.1 Approval Required.
 - 5.1.2 Business and Licensing Requirements.
 - 5.1.3 Subsequent Changes.
 - 5.1.4 Bonding of Subcontractors.
- 5.2 Subcontractual Relations.
 - 5.2.1 Comply With Contract Documents.
 - 5.2.2 Rights.
 - 5.2.3 Sub-Subcontractors.
 - 5.2.4 Document Copies.
- 5.3 Contingent Assignment of Subcontracts.
 - 5.3.1 Conditions for Assignment to Owner.

ARTICLE 6. PROTECTION OF PERSONS AND PROPERTY

- 6.1 Safety of Persons and Property.
 - 6.1.1 Contractor Responsibility.
 - 6.1.2 Safety Program, Precautions.
 - 6.1.3 Compliance with Laws.
 - 6.1.4 Erect and Maintain Safeguards.
 - 6.1.5 Utmost Care.
 - 6.1.6 Prompt Remedy.
 - 6.1.7 Safety Designee.
 - 6.1.8 Load Safety.
 - 6.1.9 Off-Site Responsibility.
 - 6.1.10 Emergencies.
- 6.2 Hazardous Materials.
- 6.3 Historical and Archeological Considerations.
- 6.4 Contractor Liability.

ARTICLE 7. MODIFICATIONS, REQUEST FOR INFORMATION, PROPOSED CHANGE ORDER, PRELIMINARY RESOLUTION EFFORTS AND CLAIMS PROCESS

- 7.1 Modifications: In General.
 - 7.1.1 Types of Modifications and Limitations.
 - 7.1.2 By Whom Issued.
 - 7.1.3 Contractor to Proceed Unless Otherwise Stated.
 - 7.1.4 Adjusting Unit Prices.
 - 7.1.5 Special Notices Required In Order to
- 7.2 Contractor Initiated Requests.
 - 7.2.1 The Request for Information, RFI, Process and Time to File.
 - 7.2.2 Proposed Change Order ("PCO").
- 7.3 Proposal Request Initiated by Owner.
 - 7.3.1 If Agreement, Change Order Issued. Process.
 - 7.3.2 If Disagreement.

- 7.4 Evaluation of Proposal for Issuing Change Orders.
 - 7.4.1 Adjusting Sum Based Upon Agreement.
 - 7.4.2 Owner Resolution of Sum and Standards in the Absence of an Agreement under Paragraph 7.4.1.
 - 7.4.3 Credits.
- 7.5 Construction Change Directives.
 - 7.5.1 When Used and Contractor's Right to Challenge.
 - 7.5.2 Proceed With Work and Notify Owner About Adjustment Method.
 - 7.5.3 Interim Payments by Owner.
- 7.6 A/E's Supplemental Instruction
- 7.7 Procedure for Preliminary Resolution Efforts.
 - 7.7.1 Request for Preliminary Resolution Effort (Pre).
 - 7.7.2 Time for Filing.
 - 7.7.3 Content Requirement.
 - 7.7.4 Supplementation.
 - 7.7.5 Subcontractors.
 - 7.7.6 Pre-Resolution Procedure.
 - 7.7.7 Contractor Required to Continue Performance.
 - 7.7.8 Decision.
 - 7.7.9 Decision Final Unless Claim Submitted.
 - 7.7.10 Extension Requires Mutual Agreement.
 - 7.7.11 If Decision Not Issued.
 - 7.7.12 Payment for Performance.
- 7.8 Resolution of Claim.
 - 7.8.1 Claim.
 - 7.8.2 Subcontractors.
 - 7.8.3 Time for Filing.
 - 7.8.4 Content Requirement.
 - 7.8.5 Extension of Time to Submit Documentation.
 - 7.8.6 Contractor Required to Continue
 - 7.8.7 Agreement of Claimant on Method
 - 7.8.8 The Evaluation Process,
 - 7.8.9 Timeframes of Evaluator(s), Director's Determination, Administrative Appeal to the Executive Director and Judicial Review.
- 7.9 Payment of Claim.
- 7.10 Allocation of Costs of Claim Resolution Process.
- 7.11Alternative Procedures.
- 7.12Impact on Future Selections.
- 7.13Owner's Right to Have Issues, Disputes or Claims
 Considered

ARTICLE 8. PAYMENTS AND COMPLETION

- 8.1 Schedule of Values.
- 8.2 Applications for Payment.

- 8.2.1 In General.
- 8.2.2 Payment for Material and Equipment.
- 8.2.3 Warranty of Title.
- 8.2.4 Holdback by Owner.
- 8.3 Certificates for Payment.
 - 8.3.1 Issued by A/E.
 - 8.3.2 A/E's Representations.
- 8.4 Decisions to Withhold Certification.
 - 8.4.1 When Withheld.
 - 8.4.2 Certification Issued When Reasons For Withholding Removed.
 - 8.4.3 Continue Work Even if Contractor Disputes A/E's Determination.
 - 8.4.4 Owner Not in Breach.
- 8.5 Progress Payments.
 - 8.5.1 In General, Interest on Late Payments.
 - 8.5.2 Contractor and Subcontractor Responsibility.
 - 8.5.3 Information Furnished by A/E or Owner to Subcontractor.
 - 8.5.4 Owner and A/E Not Liable.
 - 8.5.5 Certificate, Payment or Use Not Acceptance of Improper Work.
- 8.6 Payment upon Substantial Completion.
- 8.7 Partial Occupancy or Use.
 - 8.7.1 In General.
 - 8.7.2 Inspection.
 - 8.7.3 Not Constitute Acceptance.
- 8.8 Final Payment.
 - 8.8.1 Certificate for Payment.
 - 8.8.2 Conditions for Final Payment.
 - 8.8.3 Waiver of Claims: Final Payment.
 Performance.
 - 8.8.4 Delays Not Contractor's Fault.
 - 8.8.5 Waiver by Accepting Final Payment.

ARTICLE 9. TESTS AND INSPECTIONS, SUBSTANTIAL AND FINAL COMPLETION, UNCOVERING, CORRECTION OF WORK AND GUARANTY PERIOD

- 9.1 Tests and Inspections.
 - 9.1.1 In General.
 - 9.1.2 Failure of an Inspector to Appear.
 - 9.1.3 Nonconforming Work.
 - 9.1.4 Certificates.
 - 9.1.5 A/E Observing.
 - 9.1.6 Promptness.
- 9.2 Inspections: Substantial and Final.
 - 9.2.1 Substantial Completion Inspection.
 - 9.2.2 Final Completion Inspection.
- 9.3 Uncovering of Work.
 - 9.3.1 Uncover Uninspected Work.
 - 9.3.2 Observation Prior to Covering.

- 9.3.3 When an Inspector Fails to Appear or A/E or Owner Did Not Make Prior Request.
- 9.4 Correction of Work and Guaranty Period.
 - 9.4.1 Contractor Correct the Work.
 - 9.4.2 Guaranty and Correction after Substantial Completion.
 - 9.4.3 Removal of Work.
 - 9.4.4 Not Limit Other Obligations.
- 9.5 Additional Warranties.
 - 9.5.1 In General.
 - 9.5.2 Exclusion.
 - 9.5.3 Furnish Evidence on Request.
- 9.6 Acceptance of Nonconforming Work.

ARTICLE 10. INSURANCE AND BONDS

- 10.1Liability Insurance.
 - 10.1.1 In General.
 - 10.1.2 Configurations.
 - 10.1.3 Contractor Liability.
 - 10.1.4 Certificate, Notice Requirements, Additional Insured.
 - 10.1.5 Deductible Liability.
 - 10.1.6 Additional Requirements:
- 10.2"Builder's Risk" Property Insurance.
 - 10.2.1 In General.
 - 10.2.2 Inspections, Recommendations.
 - 10.2.3 Deductible.
 - 10.2.4 Adjusted With and Payable to Risk Manager as Trustee.
 - 10.2.5 Waiver.
 - 10.2.6 Special Hazards.
- 10.3Performance Bond and Payment Bond.

ARTICLE 11. MISCELLANEOUS PROVISIONS

- 11.1A/E's Responsibilities.
- 11.2Successors and Assigns.
- 11.3Written Notice.
 - 11.3.1 Personal Delivery and Registered or Certified Mail.
 - 11.3.2 Fax.
- 11.4Rights and Remedies.
 - 11.4.1 Not Limit.
 - 11.4.2 Not Waiver.
- 11.5Commencement of Statutory Limitation Period.

- 11.5.1 Before Substantial Completion.
- 11.5.2 Between Substantial Completion and Final Certification for Payment.
- 11.5.3 After Final Certification for Payment.
- 11.5.4 Exception.
- 11.6 Not Discriminate, No Sexual Harassment.
- 11.7 Applicable Laws.
- 11.8 Interpretation.
- 11.9 Venue.
- 11.10 Severability.
- 11.11 Construction of Words.
- 11.12 No Third Party Rights.

ARTICLE 12. TERMINATION OR SUSPENSION OF THE CONTRACT

- 12.1Termination by Contractor.
 - 12.1.1 In General.
 - 12.1.2 Notice.
- 12.2Termination by the Owner for Cause.
 - 12.2.1 In General.
 - 12.2.2 Owner's Right to Carry Out the Work.
 - 12.2.3 Items Required to be Transferred or Delivered.
 - 12.2.4 Payment.
 - 12.2.5 Owner Protection if Lienable.
 - 12.2.6 Credits and Deficits.
 - 12.2.7 If Contractor Found Not in Default or Excusable.
 - 12.2.8 Rights and Remedies Not Exclusive.
- 12.3Suspension, Delay or Interruption of Work by the Owner for Convenience.
 - 12.3.1 By Owner in Writing.
 - 12.3.2 Time Period for Claims.
 - 12.3.3 Adjustments.
- 12.4Termination for Convenience of the Owner.
 - 12.4.1 In General.
 - 12.4.2 Contractor Obligations.
 - 12.4.3 Termination Claim.
 - 12.4.4 Agreed Upon Payment.
 - 12.4.5 Payment Not Agreed Upon.
 - 12.4.6 Deductions.
 - 12.4.7 Partial Termination.
 - 12.4.8 Partial Payments.
 - 12.4.9 Preserve and Make Available Records.
- 12.50wner's Right to Stop the Work.

BOX ELDER COUNTY FACILITIES MANAGEMENT DIVISION

GENERAL CONDITIONS

FEBURARY 2021

ARTICLE 1. GENERAL PROVISIONS.

1.1 BASIC DEFINITIONS.

A/E (including all design professionals). "A/E" means the person lawfully licensed to practice architecture or engineering or an entity lawfully practicing architecture or engineering identified as such in the A/E's Agreement and is referred to throughout the Contract Documents as if singular in number. The term "A/E" also means the A/E's representative and its subconsultants. When these General Conditions are part of a Contract in which the design professional is an interior designer, landscape subconsultant or other design professional, the term "A/E" as used in these General Conditions shall be deemed to refer to such design professional. A license is not required when the type of design professional is one which is not subject to a professional license, but such professional must meet the prevailing standards in the State of Utah for such practice. For projects where there is no A/E hired by Owner, the references in the General Conditions to A/E shall be deemed to refer to Owner as may be practicably applied.

ADDENDA. "Addenda" means the written or graphic instruments issued prior to the opening of Bids which clarify, correct or change the bidding documents or the Contract Documents.

ASI. "ASI" shall mean a Supplemental Instruction issued by the A/E to the Contractor which may result in clarifications or minor changes in the Work and does not affect the contract time or the contract amount.

BID. "Bid" means the offer or proposal of the bidder submitted on the prescribed form setting forth the prices for the Work to be performed.

BONDS. "Bonds" mean the bid bond, performance and payment bonds and other instruments of security.

CHANGE ORDER. "Change Order" means a written instrument signed by the Owner and Contractor, stating their agreement for changes of the Contract as specified on the required Owner's change order form.

CLAIM. "Claim" means a dispute, demand, assertion or other matter submitted by the Contractor, including a Subcontractor at any tier subject to the provisions of these General Conditions. The claimant may seek, as a matter adjustment of right. modification, interpretation of Contract terms, payment of money, extension of time or other relief with respect to the terms of the Contract. A request for Preliminary Resolution Effort (PRE) shall not be considered a "Claim." A requested amendment, requested change order, or a Construction Change Directive (CCD) is not a PRE or Claim unless agreement cannot be reached and the procedures of these General Conditions are followed.

CONSTRUCTION CHANGE DIRECTIVE. A

"Construction Change Directive" means a written order signed by the Owner, directing a change in the Work and stating a proposed basis for adjustment, if any, in the Contract Sum

or Contract Time, or both. The Owner may by Construction Change Directive, without invalidating the Contract, order changes in the Work within the general scope of the Contract consisting of additions, deletions or other revisions; even if it may impact the Contract Sum and Contract Time.

CONTRACT. The Contract Documents form the Contract for Construction. The term "Contract" represents the entire and integrated agreement between the parties hereto and supersedes prior negotiations, representations or agreements, either written or oral. The Contract Documents shall not be construed to create a contractual relationship of any kind (1) between the A/E and Contractor, (2) between the Owner and a Subcontractor or (3) between any persons or entities other than the Owner and Contractor.

CONTRACT DOCUMENTS. The term "Contract Documents" means the Contractor's Agreement between the Owner and Contractor (hereinafter referred to as "Contractor's Agreement"), the Conditions of the Contract (General, Supplementary and other Conditions), the Drawings, Specifications, Addenda, other documents listed in the Contractor's Agreement and Modifications issued after execution of the Contractor's Agreement. The Contract Documents shall also include bidding/proposal documents, including the Instructions to Bidders/Proposers, Notice to Contractors and the Bid/Proposal Form, to the extent not in conflict with the other abovestated Contract Documents and other documents and oral presentations as part of the Selection which are documented as an attachment to the Contract.

CONTRACT SUM. The term "Contract Sum" means the Contract Sum as stated in the Contractor's Agreement and, including authorized and signed adjustments to this agreement (modifications), is the total amount payable by the Owner to the Contractor for performance of the Work under the Contract Documents.

CONTRACT TIME. "Contract Time", unless otherwise provided in the Contract Documents, means the period of time, including authorized and signed adjustments (modifications), stated in the Contract Documents for Substantial Completion of the Work.

CONTRACTOR. The Contractor is the person or entity identified as such in the Owner Contractor's Agreement and is referred to throughout the Contract Documents as if singular in number. The term "Contractor" means the Contractor or the Contractor's authorized representative. When separate contracts are awarded for different portions of the Project or other construction or operations on the site, the term "Contractor" in the Contract Documents in each case, shall mean the Contractor who executes each separate Owner Contractor Agreement.

CONTRACTOR'S AGREEMENT. "Contractor's Agreement" means, unless the context requires otherwise, the agreement executed by the Contractor and Owner for the Project.

DAY. The term "day" or "days" as used in the Contract Documents shall mean calendar day unless otherwise specifically defined.

DEFECTIVE. "Defective" is an adjective which when modifying the word "Work" refers to Work that does not conform to the Contract Documents, or does not meet the requirements of any inspection, referenced standard, code, test or approval referred to in the Contract Documents, or has been damaged.

OWNER REPRESENTATIVE.

"Owner Representative" means the Facilities Management Division person directly assigned to work with the Contractor on a regular basis. Unless the context requires otherwise, the "Owner Representative" is the "Owner's Representative."

DIRECTOR. "Director" means the Facilities Management Division Manager unless the

context requires otherwise. Director may include a designee selected by the Director for the particular function referred to in the General Conditions.

OWNER. "Owner" means Box Elder County. Unless the context requires otherwise, Owner is the "Owner" as that term is commonly referred to in the construction industry.

DRAWINGS. The "Drawings" are the graphic and pictorial portions of the Contract Documents, wherever located and whenever issued, showing the design, location and dimensions of the Work, and generally include the drawings, elevations, sections, details, schedules and diagrams.

EXECUTIVE DIRECTOR. "Executive Director" means the Box Elder County Commission, including unless otherwise stated, his/her duly authorized designee.

INSPECTION. The word "inspection" or its derivatives shall mean a review of the Project, including but not limited to a visual review of the Work completed to date to ascertain if the Work is in accordance with the Contract Documents, including all applicable building codes and construction standards.

MODIFICATION. A "Modification" is (1) a Change Order (2) Construction Change Directive or (3) ASI. The Contract may be amended or modified only by (1) a written amendment executed by both the Owner and Contractor, or (2) by a Modification.

NOTICE TO PROCEED. A "Notice to Proceed" is a document prepared by the Owner and by its terms authorizes the Contractor to commence Work on the Project. It is deemed issued upon being sent by the Owner to the Contractor's specified address within the bid or proposal.

PARTIAL USE. "Partial Use" means placing a portion of the Work in service for the purpose for which it is intended (or a related purpose) before reaching Substantial Completion for all the

Work. This partial use does not constitute "substantial completion".

PRELIMINARY RESOLUTION EFFORT.

"Preliminary Resolution Effort" or "PRE" means the processing of a request for preliminary resolution or any similar notice about a problem that could potentially lead to a Claim and is prior to reaching the status of a Claim.

PRODUCT DATA. "Product Data" means illustrations, standard schedules, performance charts, instructions, brochures, diagrams and other information furnished by the Contractor to illustrate materials or equipment for some portion of the Work.

PROJECT. The "Project" means the total construction of the Work performed under the Contract Documents.

PROJECT MANUAL (FOR CONSTRUCTION). The "Project Manual" is the volume assembled for the Work and may include the bidding/proposal requirements, sample forms, General or Supplementary Conditions of the Contract and Specifications.

PROPOSAL REQUEST OR "PR". A "Proposal Request" or "PR" is a proposal request filed with the Contractor for the purposes of seeking a proposal in order to resolve an issue as part of the Change Order or Contract Modification process.

PROPOSED CHANGE ORDER. A "Proposed Change Order" ("PCO"), is an informal request by the Contractor filed with the Owner Representative, in an effort to commence the Contract Modification Process. It shall not be considered a "PRE" or a "Claim." The PCO may be related to any potential, or actual delay, disruption, unforeseen condition or materials or any other matter in which the Contractor intends to seek additional monies or time.

REQUEST FOR INFORMATION or RFI. A

"Request for Information" or "RFI" is a request filed by the Contractor with the A/E regarding any request for information, direction or clarification related to the Contract Documents, plans or specifications.

RESOLUTION OF THE CLAIM. "Resolution of the Claim" means the final resolution of the Claim by the Director, but does not include any administrative appeal, judicial review or judicial appeal thereafter.

RULE. "Rule," unless the context requires otherwise, shall mean a Rule of the Utah Administrative Code.

SALES TAX and/or USE TAX. Sales Tax and/or Use Tax, unless the context requires otherwise, shall mean the sales tax and/or use tax collected or to be collected by the Utah State Tax Commission and shall include any sales and/or use tax that the Utah State Tax Commission collects on behalf of any special district, local government or political subdivision.

SAMPLES. "Samples" mean physical examples, which illustrate materials, equipment or workmanship and establishes standards by which the Work will be judged.

SHOP DRAWINGS. "Shop Drawings" means drawings, diagrams, schedules and other data specially prepared for the Work by the Contractor or a Subcontractor, Subsubcontractor, manufacturer, supplier or distributor to illustrate some portion of the Work.

SPECIFICATIONS. The "Specifications" are that portion of the Contract Documents consisting of the written requirements for materials, equipment, construction systems, standards, installation and workmanship for the Work, and performance of related systems and services.

SUBCONTRACTOR. "Subcontractor" means the person or entity that has a direct contract with the Contractor, including any trade contractor or

specialty contractor. or with another Subcontractor at any tier to provide labor or materials for the work but does not include suppliers who provide only materials, equipment or supplies to a contractor or subcontractor. Notwithstanding the foregoing, the text in which the term is used may provide for the exclusion of Subcontractors of other Subcontractors or the exclusion of suppliers. The term "Subcontractor" is referred to throughout the Contract Documents as if singular in number and means a Subcontractor or authorized representative of the Subcontractor. The Term "Subcontractor" does not include a separate contractor or subcontractors of a separate contractor.

SUBSTANTIAL COMPLETION. "Substantial

Completion" is the date certified in accordance with Article 9.2 and means the date the Work or designated portion thereof is sufficiently complete, and any lack of completion or performance does not reasonably interfere with the Owner's intended use of the Project, in accordance with the Contract Documents so that the Owner can occupy and use the Work for its intended use. Owner's "intended use" or "occupy" as used in this definition, shall include any intended use or occupation by any agency or entity for which Owner has intended to so occupy the Project.

SUPPLEMENTARY CONDITIONS OR SUPPLEMENTARY GENERAL CONDITIONS.

"Supplementary Conditions" or "Supplementary General Conditions" means the part of the Contract Documents which amends or supplements these General Conditions.

WORK. The term "Work" means the construction and services required by the Contract Documents, whether completed or partially completed, and includes all labor, materials, equipment and services provided, or to be provided, by the Contractor to fulfill the Contractor's obligations.

ARTICLE 2. OWNER.

2.1 INFORMATION AND SERVICES REQUIRED OF THE OWNER.

2.1.1 OWNER'S REPRESENTATIVE. The Owner shall designate an Owner Representative authorized to act in the Owner's behalf with respect to the Project. The Owner or such authorized representative shall render decisions within a reasonable time pertaining to documents submitted by the A/E and/or Contractor in order to avoid a compensable delay in the orderly and sequential progress of the Project.

2.1.2 SPECIALISTS AND INSPECTORS. The

Owner will provide certified building inspection services in accordance with the adopted Building Codes. This includes 'routine' and 'special' inspections unless otherwise noted in the A/E Agreement. The Owner may assign an inspector or specialist to note deviations from, or necessary adjustments to, the Contract Documents or to report deficiencies or defects in the Work. The inspector or specialist's activities in no way relieves the Contractor of the responsibilities set forth in the Contract Documents.

- **2.1.3 SURVEYS AND LEGAL DESCRIPTION.** The Owner shall furnish surveys describing physical characteristics, legal limitations and utility locations for the site of the Project, and a legal description of the site. The Contractor shall review this information, including the surveys and any provided soils tests, and compare such information with observable physical conditions and the Contract Documents.
- **2.1.4 PROMPT INFORMATION AND SERVICES.** Upon receipt of a written request from the Contractor, the Owner shall furnish information or services under the Owner's control with reasonable promptness to avoid delay in the orderly progress of the Work.
- 2.1.5 COPIES OF DRAWINGS AND PROJECT MANUALS (FOR CONSTRUCTION). Unless

otherwise provided in the Contract Documents, the Contractor will be furnished, free of charge, such copies of Drawings and Project Manuals (for construction) as are reasonably necessary for execution of the Work. Owner's Web Page may also provide referenced documents for the Project.

- **2.1.6 OTHER DUTIES.** The foregoing is in addition to other duties and responsibilities of the Owner enumerated herein and especially those in respect to Article 2.2 (Construction by Owner or by Separate Contractors), Article 8 (Payments and Completion) and Article 10 (Insurance and Bonds).
- 2.2 CONSTRUCTION BY OWNER OR BY SEPARATE CONTRACTORS.
- 2.2.1 OWNER'S RIGHT TO PERFORM CONSTRUCTION AND TO AWARD SEPARATE CONTRACTS.
- (1) IN GENERAL. The Owner reserves the right to perform construction or operations related to the Project with the Owner's own forces, and to award separate contracts in connection with other portions of the Project or other construction or operations on the site under Conditions of the Contract identical or substantially similar to these including those portions related to insurance and waiver of subrogation.
- (2) COORDINATION AND REVISIONS. The Owner shall provide for coordination of the activities of the Owner's own forces and of each separate Contractor with the Work of the Contractor, who shall cooperate with them. The Contractor shall participate with other separate contractors and the Owner in reviewing their construction schedules when directed to do so. The Contractor shall make any revisions to the construction schedule and Contract Sum deemed necessary after a joint review and agreement by the Owner. The construction schedules shall then constitute the schedules to

be used by the Contractor, separate contractors and the Owner until subsequently revised.

2.2.2 MUTUAL RESPONSIBILITY.

- (1) CONTRACTOR COORDINATION. The Contractor shall afford the Owner and separate contractor(s) a reasonable opportunity for delivery and storage of their materials and equipment and performance of their activities and shall connect and coordinate the Contractor's construction and operations with theirs as required by the Contract Documents.
- (2) REPORTING PROBLEMS TO OWNER. If part of the Contractor's Work depends on work by the Owner or a separate contractor, the Contractor shall, prior to proceeding with that portion of the Work, promptly report in writing to the Owner apparent defects in workmanship that would render it unsuitable for proper execution. Failure of the Contractor to make said report shall constitute an acknowledgment that the Owner's or separate contractor's completed or partially completed construction is fit and proper to receive the Contractor's Work, except as to defects in workmanship not then reasonably discoverable.
- (3) COSTS. Costs caused by delays or by improperly timed activities or defective construction shall be borne by the responsible party in accordance with the procedures and provisions of the Contract Documents.
- (4) CONTRACTOR REMEDIAL WORK. The Contractor shall promptly remedy damage caused by the Contractor to completed or partially completed Work or to property of the Owner or separate contractors and subcontractors as provided in Article 6.

ARTICLE 3. A/E.

3.1 A/E'S ADMINISTRATION OF THE CONTRACT.

3.1.1 IN GENERAL. The A/E assists the Owner with the administration of the Contract as described in the Contract Documents. The A/E shall have the authority to act on behalf of the Owner only to the extent provided in the Contract Documents or A/E's Agreement.

3.1.2 SITE VISITS.

- (1) Site visits or inspections by the A/E, the Owner or any Owner representative shall in no way limit or affect the Contractor's responsibility to comply with all the requirements and the overall design concept of the Contract Documents as well as all applicable laws, statutes, ordinances, resolutions, codes, rules, regulations, orders and decrees.
- **(2) WRITTEN REPORT.** The A/E shall promptly submit to the Owner a written report subsequent to each site visit.
- COMMUNICATIONS **FACILITATING** 3.1.3 CONTRACT ADMINISTRATION. Except authorized by the Owner Representative or as otherwise provided in the Contract Documents, including these General Conditions, the A/E and Contractor shall communicate through the Owner Representative on issues regarding the timing of the Work, cost of the Work or scope of the Work. Contractor shall comply with communication policies agreed upon at any preconstruction meeting with the Owner. Communications by and with the A/E subconsultants shall be through the A/E. Communications by and with Subcontractors shall be through the Contractor. Communications by and with separate contractors shall be through the Owner.
- **3.1.4** A/E MAY REJECT WORK, ORDER INSPECTION, TESTS. The A/E shall have the responsibility and authority to reject Work which, based upon the A/E's knowledge or what may be reasonably inferred from the A/E's site observations and review of data, does not conform to the Contract Documents. Whenever the A/E considers it necessary or advisable for implementation of the intent of the Contract

Documents, the A/E shall have the responsibility and authority to require additional inspections or testing of the Work in accordance with the provisions of the Contract Documents, whether or not such Work is fabricated, installed or completed, provided, however, the A/E must obtain the Owner's prior written approval of any such additional inspections or testing. However, neither this authority of the A/E nor a decision made in good faith either to exercise or not to exercise such authority shall give rise to a duty or responsibility of the A/E to the Contractor, Subcontractors, their agents or employees or other persons performing portions of the Work, including separate contractors. If the Contractor disputes the rejection of any Work and the correction thereof shall involve additional cost or time, it shall be the Owner's option to accept such Work whether it be conforming or nonconforming.

3.1.5 A/E REVIEW CONTRACTOR'S SUBMITTALS.

- (1) Contractor shall submit shop drawings, product data, and samples and other submittals required by the Contract Documents to the A/E as required by the approved submittal schedule.
- (2) The A/E shall review and approve or take other appropriate action upon Contractor's submittals such as Shop Drawings, Product Data and Samples, but only for the purpose of checking for conformance with the information and design concepts expressed in the Contract Documents. A/E action taken on a submittal shall not constitute a Modification of this Agreement.
- (3) The A/E's action shall be taken no later than 15 days following A/E's receipt of the submittal, unless agreed to otherwise by Contractor and Owner, in order to avoid a delay in the Work of the Contractor or of separate contractors while allowing sufficient time in the A/E's professional judgment to permit adequate review.
- (4) Review of such submittals is not conducted for the purpose of determining the accuracy and

- completeness of other details such as dimensions and quantities or for substantiating instructions for installation or performance of equipment or systems, all of which remain the responsibility of the Contractor as required by the Contract Documents.
- **(5)** The A/E's review of the Contractor's submittals shall not relieve the Contractor of the obligations under the Contract Documents.
- **(6)** The A/E's review shall not constitute approval of safety precautions or, unless otherwise specifically stated by the A/E, of any construction means, methods, techniques, sequences or procedures.
- (7) The A/E's approval of a specific item shall not indicate approval of an assembly of which the item is a component.
- **(8)** When professional certification of performance characteristics of materials, systems or equipment is required by the Contract Documents, the A/E shall be entitled to rely upon such certifications to establish that the materials systems or equipment will meet the performance criteria required by the Contract Documents.
- 3.2 OWNERSHIP AND USE OF A/E'S DRAWINGS, SPECIFICATIONS AND OTHER DOCUMENTS. All Drawings, Specifications and other documents prepared by the A/E are and shall remain the property of the Owner, and Owner shall retain all common law, statutory and other reserved rights with respect thereto. Said documents were prepared and are intended for use as an integrated set for the Project which subject of this Contractor's Agreement. The Contractor shall not modify or use Contract Documents on any other project without the prior written consent of the Owner and A/E. Any such non-permissive use or modification, by Contractor, the Contractor's Subcontractors at any tier or anyone for whose acts the Contractor is liable, shall be at Contractor's sole risk. Contractor shall hold harmless and indemnify

the Owner from and against any and all claims, actions, suits, costs, damages, loss, expenses and attorney fees arising out of such non-permissive use or modification by the Contractor. Contractor and Subcontractors are granted a limited license to use and reproduce applicable portions of the Drawings, Specifications and other documents prepared by the A/E appropriate to and for use in the execution of their Work under the Contract Documents. All copies made under this license shall bear the statutory copyright notice, if any, shown on the Drawings, Specifications and other documents prepared by the A/E. Submittals or distributions necessary to meet official regulatory requirements or for other purposes relating to completion of the Project are not to be construed as a publication in derogation of the Owner's copyright or other reserved rights.

ARTICLE 4. CONTRACTOR

4.1 REVIEW OF CONTRACT DOCUMENTS AND FIELD CONDITIONS BY CONTRACTOR.

4.1.1 REVIEWING CONTRACT DOCUMENTS.

INFORMATION. REPORTING ERRORS. **INCONSISTENCIES** OR OMISSIONS. The Contractor shall carefully study and compare the Contract Documents with each other and with information furnished by the Owner pursuant to Article 2.1 hereinabove and shall at once report to the Owner and A/E errors, inconsistencies or omissions discovered. The Contractor shall not be liable to the Owner or A/E for damage resulting from errors, inconsistencies or omission in the Contract Documents, unless the Contractor recognized such error, inconsistency or omission or a Contractor of ordinary skill and expertise for the type of Work involved would have readily so recognized such inconsistency or omission, and the Contractor failed to report such to the Owner and A/E. If the Contractor performs any construction activity without such notice to the Owner and A/E and prior to the resolution of the error, inconsistency or omission, the Contractor shall assume

appropriate responsibility for such performance and shall bear an appropriate amount of the attributable costs for correction.

- **4.1.2 FIELD CONDITIONS.** The Contractor shall take field measurements and verify field conditions and shall carefully compare such field measurements and conditions and other information known to the Contractor, or information which a Contractor of ordinary skill and expertise for the type of Work involved would have known, before commencing activities. Errors, inconsistencies or omissions discovered shall be reported to the Owner and A/E at once. If the Contractor performs any construction activity without such notice to the Owner and A/E and prior to the resolution of the error, inconsistency or omission, the Contractor shall assume appropriate responsibility for such performance and shall bear an appropriate amount of the attributable costs for correction.
- **4.1.3 PERFORM IN ACCORDANCE WITH CONTRACT DOCUMENTS AND SUBMITTALS.** The Contractor shall perform the Work in accordance with the Contract Documents and submittals approved in accordance with the Contract Documents.
- **4.1.4 PERFORMANCE TO PRODUCE THE COMPLETE SYSTEM AND INTENDED RESULTS.** Performance by the Contractor shall be required

to the extent consistent with the Contract Documents and reasonably inferable from the Contract Documents as being necessary to allow the system to function within its intended use.

4.1.5 INTENT AND HIERARCHY. The Contract Documents should be read as a whole and wherever possible, the provisions should be construed in order that all provisions are operable. The intent of the Contract Documents is to include all items necessary for the proper execution and completion of the Work by the Contractor. The Contract Documents are complimentary, and what is required by one Document or provisions thereof shall be as binding as if required by all the Documents or

provisions thereof. In case of an irreconcilable conflict between provisions within a Contract Document or between Contract Documents, the following priorities shall govern as listed below:

- (1) A particular Modification shall govern over all Contract Document provisions or Modifications issued prior to said particular Modification.
- (2) Attachments to the Contractor's Agreement resulting from the Selection process including any management plan or documented interview information shall govern over addenda, the General Conditions, plans and specifications.
- (3) A particular Addendum shall govern over all other Contract Document provisions issued prior to said particular Addendum. Subsequent Addenda shall govern over all prior Addenda.
- (4) The Supplementary General Conditions shall govern over the General Conditions.
- (5) These General Conditions shall govern over all other Contract Documents except for the Supplementary General Conditions, Addenda, Modifications and Attachments resulting from the selection process.
- **(6)** The drawings and specifications shall not govern over any of the documents listed above.
- (7) In case of a conflict or ambiguity within the same level of hierarchy of described documents, Owner reserves the right to select the most stringent requirement unless the preponderance of the contract indicates the less stringent requirement.
- **4.1.6 DIVIDING WORK AND CONTRACTOR REPRESENTATION.** Organization of the specifications into divisions, sections and articles, and arrangement of Drawings, shall not control the Contractor in dividing the Work among Subcontractors or in establishing the extent of Work to be performed by any trade. Contractor represents that the Subcontractors,

Sub-subcontractors, manufacturers and suppliers engaged or to be engaged by it are and will be familiar with the requirements for performance by them of their obligations.

4.1.7 PLANNING AND PRIORITY. The Contractor shall plan and schedule its work to facilitate the Project and shall maintain a work schedule to place proper priority to sequence work to complete the project timely.

4.2 SUPERVISION AND CONSTRUCTION PROCEDURES.

4.2.1 SUPERVISION AND CONTROL. The Contractor shall supervise and direct the Work, using the Contractor's best skill and attention. The Contractor shall be solely responsible for and have control over the construction means, methods, techniques, sequences and procedures and for coordinating all portions of the Work under the Contract, except to the extent that the Contract Documents expressly and specifically state otherwise.

4.2.2 **RESPONSIBILITY**.

The Contractor shall be responsible to the Owner for acts and omissions of the Contractor's employees, Subcontractors, and their agents and employees, and other persons performing portions of the Work under a contract with the Contractor or on behalf of the Contractor.

4.2.3 NOT RELIEVED OF OBLIGATIONS. The Contractor shall not be relieved of obligations to perform the Work in accordance with the Contract Documents either by activities or duties of the Owner or its agents in the Owner's administration of the Contract, or by tests,

inspections or approvals required or performed by persons other than the Contractor or for those that the Contractor is liable.

4.2.4 INSPECTIONS AND APPROVALS.

- (1) The Contractor is responsible for requesting inspections for various stages and portions of the Work required under the Contract Documents in a timely manner.
- (2) If any of the Work is required to be inspected or approved by the terms of the Contract Documents by any public authority, the Contractor shall timely request such inspection or approval to be performed in accordance with Article 9. Except as provided in Article 9, work shall not proceed without any required inspection and the associated authorization to proceed. Contractor shall promptly notify Owner if the inspector fails to appear at the site.

4.3 LABOR AND MATERIALS.

4.3.1 PAYMENT BY CONTRACTOR. Except to the extent it is otherwise stated in the Contract Documents, the Contractor shall provide and pay for labor, materials, equipment, tools, construction equipment and machinery, water, heat, utilities, transportation, and other facilities, supplies, consumables and services necessary for the proper execution and completion of the Work, whether temporary or

permanent and whether or not incorporated or to be incorporated in the Work.

4.3.2 DISCIPLINE AND COMPETENCE. The Contractor shall enforce strict discipline and good order among the Contractor's employees, its Subcontractors, agents, representatives and other persons performing under the Contract Documents. The Contractor shall not permit employment of unfit persons or persons not skilled in tasks assigned to them.

4.4 TAXES AND OTHER PAYMENTS **GOVERNMENT.** The Contractor shall pay sales, consumer, use, employment-related and similar taxes related to the Work or portions thereof provided by the Contractor which are legally enacted when bids are received or negotiations concluded, whether or not yet effective or merely scheduled to go into effect. The Contractor shall comply with the laws and regulations regarding the payment of Sales and/or Use Tax and any exemptions. The procurement documents may have a provision regarding specific items which are exempt from State of Utah Sales Tax and/or Use Tax. Any such exemption shall be used only for the items and the project specified in the procurement documents. Any such exemption does not apply to taxes levied by the federal government or any taxing entity outside of the State of Utah. If a Contractor properly relies upon a provision(s) of the bidding or proposal documents, and if State of Utah Sales and/or Use Tax subsequently becomes due, then the Contractor shall be paid such tax amount not included in the bid/proposal amount due to the reliance upon such provision.

4.5 PERMITS, FEES, NOTICES, LABOR AND MATERIALS.

4.5.1 PERMITS AND FEES. Unless required in the Supplementary General Conditions or an Addendum, it will not be necessary for the Contractor to obtain or pay for local building permits, plan check fees, electrical permits, plumbing permits, connection fees, or impact

fees, nor will it be necessary to pay fees for inspections pertaining thereto.

- **4.5.2 COMPLIANCE** WITH PUBLIC AUTHORITIES, NOTICES. The Contractor shall comply with and give notices required by laws, ordinances, resolutions, rules, regulations and lawful orders of public authorities bearing on the performance of the Work.
- CORRELATION OF 4.5.3 **CONTRACT DOCUMENTS AND ENACTMENTS.** It is not the Contractor's responsibility to ascertain that the Contract Documents are in accordance with applicable statutes, laws. ordinances, resolutions, building codes, and rules and regulations. Notwithstanding this, if the Contractor observes, or if such is readily observable to a Contractor of ordinary skill and expertise for the type of Work involved, that a portion of the Contract Documents is at variance therewith, the Contractor shall promptly notify the A/E and Owner in writing, and necessary changes shall be accomplished appropriate Modification.
- **4.5.4 FAILURE TO GIVE NOTICE.** If the Contractor, or any Subcontractor thereof performs Work without complying with the requirements of this Article 4.5 hereinabove, the Contractor shall assume appropriate responsibility for such Work and shall bear the appropriate amount of the attributable costs.
- **4.6 SUPERINTENDENT.** The Contractor shall employ a competent superintendent and necessary assistants who shall be in attendance at the Project site during performance of the Work. The superintendent shall represent the Contractor, and communications given to the superintendent shall be as binding as if given to the Contractor. Important communications shall be confirmed in writing. Other communications shall be similarly confirmed on written request in each case.
- **4.7 TIME AND CONTRACTOR'S CONSTRUCTION SCHEDULES.**

4.7.1 PROGRESS AND COMPLETION.

- (1) TIME IS OF THE ESSENCE; COMPLETE WITHIN CONTRACT TIME. Time is of the essence. By executing the Contractor's Agreement, the Contractor confirms that the Contract Time is adequate to perform the Work. The Contractor shall proceed expeditiously with adequate forces to achieve Substantial Completion within the Contract Time.
- (2) NOTICE TO PROCEED AND INSURANCE. The Contractor shall not prematurely commence operations on the site or elsewhere prior to the issuance of a Notice to Proceed by the Owner or prior to the effective date of insurance required by Article 10 to be furnished by the Contractor, whichever is the latter.
- 4.7.2 SCHEDULE PREPARATION. The Contractor, promptly after being awarded the Contract, shall prepare and submit for the Owner's and A/E's review, a reasonably detailed CPM schedule for the Work. The schedule shall indicate the order. sequence, interdependence of all items known to be necessary to complete the Work including construction, procurement, fabrication, and delivery of materials and equipment, submittals and approvals of samples, shop drawings, procedures, or other documents. Work items of the Owner, other Contractors, utilities and other third parties that may affect or be affected by the Contractor shall be included. If the Owner is required, by the Contract Documents, to furnish any materials, equipment, or the like, to be incorporated into the Work by the Contractor, Contractor shall submit, with the first schedule submittal, a letter clearly indicating the dates that such items are required at the Project Site. The critical path should be identified, including the critical paths for interim completion dates and milestones. The Contractor's schedule shall be updated at least once a month.
- **4.7.3 INITIAL CONTRACT TIME.** Unless otherwise specified in the bidding documents,

the initial Contract Time is the time identified in the Contractor's Agreement.

4.7.4 INTERIM COMPLETION DATES AND MILESTONES. The schedule must include contractually specified interim completion dates and milestones. The milestone completion dates indicated are considered essential to the satisfactory performance of this Contract and to the coordination of all Work on the Project. The milestone dates listed are not intended to be a complete listing of all Work under this Contract or of interfaces with other Project Contractors.

4.7.5 SCHEDULE CONTENT REQUIREMENTS.

The schedule shall indicate an early completion date for the Project that is no later than the Project's required completion date. The schedule, including all activity duration's shall be given in calendar days. The Schedule shall also indicate all of the following as required by the Owner:

- (1) Interfaces with the work of outside contractors (e.g., utilities, power and with any separate Contractor);
- **(2)** Description of activity including activity number/numbers;
- (3) Estimated duration time for each activity;
- (4) Early start, late start, early finish, late finish date, and predecessor/successors including stop-start relationships with lead and lag time for each activity;
- (5) Float available to each path of activities;
- (6) Actual start date for each activity begun;
- (7) Actual finish date for each activity completed;
- **(8)** The percentage complete of each activity in progress or completed;
- (9) Identification of all critical path activities;

- (10) The critical path for the Project, with said path of activities being clearly and easily recognizable on the time-scaled network diagram. The path(s) with the least amount of float must be identified. Unless otherwise authorized by the Owner Representative, no more than 40% of all activities may be identified as critical path items. The relationship between non-critical activities and activities on the critical path shall be clearly shown on the network diagram;
- (11) Unless otherwise authorized by the Owner Representative, all activities on the schedule representing construction on the site may not have a duration longer than 14 days. Construction items that require more than 14 days to complete must be broken into identifiable activities on the schedule with durations less than 14 days. The sum of these activities represents the total length required to complete that construction item; and
- (12) Additional requirements as specified in the Supplemental General Conditions.

4.7.6 OWNER'S RIGHT TO TAKE EXCEPTIONS. The Owner reserves the right to take reasonable exception to activity duration, activity

exception to activity duration, activity placement, construction logic or time frame for any element of the Work to be scheduled.

- **4.7.7 FLOAT TIME.** Float or slack time is defined as the amount of time between the earliest start date and the latest start date or between the earliest finish date and the latest finish date of a chain of activities on the Schedule. By a proposal request or modification delivered to the Contractor, the Owner has the right to use the float time for non-critical path activities until the Contractor has reallocated such time on a newly submitted schedule.
- **4.7.8 INITIAL SCHEDULE SUBMISSION.** No progress payments will be approved until the Contractor has submitted a Project detailed CPM schedule covering the first 90 days of the Work with a general CPM schedule for the entire

project as required by the Owner. The detailed schedule for the entire project is to be completed prior to the second pay request unless otherwise authorized in writing by the Owner Representative.

4.7.9 **UPDATES.** Prior to any approval of a pay request, the Owner, A/E and Contractor shall review the Contractor's schedule compared to the Work completed. The Owner approves the amount of Work completed as supported by the schedule of values and as verified by the determination of Work completed. If necessary, the Contractor shall then update and submit to the Owner the schedule with the pay request; all of which in accordance with the Owner's approval. All updates shall be provided in electronic and hard copy formats. At each with scheduled meeting the Owner Representative, the Contractor shall provide a "three week look ahead" with long lead items identified.

4.7.10 SCHEDULE OF SUBMITTALS. The Contractor shall prepare and keep current, for the A/E's and Owner's review, a schedule of submittals required under the Contract Documents which is coordinated with the Contractor's construction schedule and allows the A/E a reasonable time to review the submittals. This submittal schedule is to be included as part of the construction schedule. Submittals requiring expedited review must be clearly identified as such in the schedule of submittals.

4.7.11 SCHEDULE RECOVERY. If the Work represented by the critical path falls behind more than 7 days, the project schedule shall be redone within 14 days showing how the Contractor shall recover the time. A narrative that addresses the changes in the schedule from the previously submitted schedule shall be submitted along with the schedule in both hard copy (appropriate report formats to be determined by the Owner Representative) and electronic copy. The Contractor shall comply with the most recent schedules.

4.7.12 SCHEDULE CHANGES AND MODIFICATIONS.

(1) CONTRACT TIME CHANGE REQUIRES MODIFICATION. The Contract Time may only be shortened or extended by a written modification fully executed by the Owner.

(2) CONTRACTOR REORDERING, RESEQUENCING AND CHANGING ACTIVITY **DURATIONS**. Should the Contractor, after approval of the complete detailed construction schedule, desire to change his plan of construction, he shall submit his requested revisions to the Owner and the A/E along with a written statement of the revisions including a description of the sequence and duration changes for rescheduling the work, methods of maintaining adherence to intermediate milestones and the contract completion date and the reasons for the revisions. If the requested changes are acceptable to the Owner, which acceptance shall not be unreasonably withheld, they will be incorporated into the Schedule in the next reporting period. If after submitting a request for change in the Contract Schedule, the Owner does not agree with the request, the Owner will schedule a meeting with the Contractor to discuss the differences.

(3) CHANGES IN CONTRACT TIME. The critical path schedule as the term is used in the provisions herein shall be based on the current version of the Contractor's schedule for the Project and accepted by the Owner just prior to the commencement of the modification, asserted delay, suspension or interruption. If the Contractor believes it is entitled to an extension of Contract Time under the Contract Documents, the Contractor shall submit a PCO in accordance with Article 7.2 to the A/E and the Owner Representative accompanied by an analysis ("Requested Time Adjustment Schedule") in accordance with the Contract Documents for time extensions. The "Requested Time Adjustment Schedule" shall include "fragnets" that represent the added or changed

work to the Schedule. The impact on unchanged activities caused by the changes and/or delays being analyzed shall be included in these fragnets.

A "fragnet" as used in these General Conditions and when used in the context of project scheduling is a subset of project activities that are inter-related by predecessor and successor relationships that are tied into the main schedule with identified start and completion points. Each fragnet may or may not be on the critical path. An entire schedule consists of a series of interrelated fragnets.

4.7.13 EXCUSABLE DELAY.

(1) IN GENERAL. If the Contractor is delayed at any time in the progress of the Work on the critical path schedule by an act or neglect of the Owner or other causes beyond the Contractor's control or by other causes which the Owner determines may justify delay, then the Contract Time shall be extended by Change Order. The Contractor shall immediately take all steps reasonably possible to lessen the adverse impact of such delay. Notwithstanding the above, to the extent any of the causes for delay were caused by the Contractor, reasonably foreseeable by the Contractor or avoidable by the Contractor, then to such extent the delay shall not be cause for extension of the Contract Time. For purposes of this paragraph, Contractors shall include all subcontractors and others under the responsibility of the Contractor.

The determination of the total number of days extension will be based upon the current construction schedule in effect at the inception of the change and/or delay and upon all data relevant to the extension as it exists in the project record. Once approved, such data shall be incorporated in the next monthly update of the schedule.

Contractor acknowledges and agrees that delays in work items which, according to the schedule analysis, do not affect any milestone dates or the Contract completion dates shown on the CPM at the time of the delay, will not be the basis for a contract extension.

- (2) WEATHER-RELATED EXCUSABLE DELAYS. Completion time will not be extended for normal bad weather or any weather that is reasonably foreseeable at the time of entering into the contract. The time for completion as stated in the contract documents includes due allowance for calendar days on which Work cannot be performed out of doors. The Contractor acknowledges that it may lose days due to weather conditions. Contract time may be extended at no cost to the Owner if all of the following are met which must be established by the Contractor:
- (a) That the weather prevented Work from occurring that is on the critical path for the project based upon a critical path schedule previously submitted to the Owner and to the extent accepted by the Owner;
- (b) There are no concurrent delays attributed to the Contractor;
- (c) The Contractor took all reasonable steps to alleviate the impact of the weather and took reasonable attempts to prevent the delay and despite such reasonable actions of Contractor, the weather impacted the critical path as described above; and
- (d) One of the following occurred:
- 1. The weather was catastrophic, such as a tornado, hurricane, severe wind storm, severe hail storm; or
- 2. Based on the full history of information published from the closest station as indicated from the Western Regional Climate Center, one or more of the following occurred:
- a. For any day between November 1 and March 31, the minimum temperature fell below the average minimum temperature plus the extreme

low temperature recorded for the month divided by 2.

- b. For any day between November 1 and March 31, the maximum temperature fell below the monthly average for the minimum temperature.
- c. The daily precipitation exceeded 75% of the historical one day maximum for the month.
- d. The snowfall for the month exceeded 175% of the historical average snow fall for the month.

4.7.14 COMPENSABLE DELAY, SUSPENSION OR INTERRUPTION.

- (1) BASIC CONDITIONS. In addition to the other requirements of the Contract Documents, a compensable delay, suspension or interruption of the work occurs only when the following are met:
- (a) Is wholly unanticipated by the parties at the time of execution of the Contractor's Agreement or is caused by the breach of a fundamental obligation of the Contract Documents attributable to the Owner; and
- (b) The Contractor delivers a written notice to A/E and Owner within seven (7) days that the Contractor knows or should have known of the condition giving rise to the purported compensable delay, disruption, suspension or interruption, and said continuation affects the Contract Time as indicated by the last submitted and reasonable critical path schedule.
- (2) COMPENSABLE DELAY FORMULA. To the extent of the compensable delay, the Contractor's total entitlement for all compensable delay damages is the computed result of the following formula: Contract Sum divided by Contract Time (in calendar days); the result of which is then multiplied by 0.05; and the result of which is multiplied by the number of calendar days of compensable days allowed under these General Conditions that are beyond the Contract Time. Notwithstanding any other

provision of these General Conditions or the Contract Documents, to the extent the Contractor is entitled to receive the 10% or 15% markup under Article 7.4, this provision shall be inapplicable and the markup shall be deemed to include all the compensable delay damages provided by this paragraph.

- (3) PERIOD OF COMPENSABLE DELAY, **SUSPENSION OR INTERRUPTION.** The length and extent of compensable delay, shall be determined, with the use of the Project's critical path schedule, by ascertaining the number of additional days to the Contract Time that are needed in order to perform the Work in accordance with the Contract Documents as a result of the continuation of the aforesaid delay, disruption, suspension or interruption after receipt of the written notice received by the A/E and Owner under Section 4.7.14(1)(b) above.
- (4) CONCURRENT DELAY. Notwithstanding any other provision of these General Conditions, to the extent a non- compensable delay occurs at the same time as a compensable delay, the Owner shall not be responsible for any compensation for the period of the non-compensable delay.
- **4.7.15 TIME EXTENSION REQUESTS.** Any time extension shall be requested within 21 days after the Contractor knew or should have known about the delay and shall be supported by the critical path schedule analysis.

4.7.16 LIQUIDATED DAMAGES.

(1) IN GENERAL. Should the Contractor fail to complete the Work within the Contract Time, there shall be deducted from any amount due or that may become due the Contractor, the sum as stated in the Contractor's Agreement. Such sum is fixed and agreed upon by the Owner and Contractor as liquidated damages due the Owner by reason of the inconvenience and added costs of administration, engineering,

supervision and other costs resulting from the Contractor's default, and not as a penalty. Actual damages related to delay cannot be ascertained at the time of execution of the Contract. To the extent that the liquidated damages exceed any amounts that would otherwise be due the Contractor, the Contractor shall be liable for such excess to the Owner. Owner may seek enforcement of such obligation by legal action, and if such is necessary, shall recover the related costs and attorney fees. Notwithstanding any other provision of these General Conditions, the availability of liquidated damages to the Owner shall not limit the Owner's right to seek damages or other remedies available under law or equity to the extent such damages or remedies are not based upon delay.

(2) NO WAIVER OF OWNER'S RIGHTS. Permitting the Contractor to continue any part of the Work after the time fixed for completion or beyond any authorized extension thereof, shall in no way operate as a waiver or estoppel on the part of the Owner of any of its rights under the Contract Documents, including the right to liquidated damages or any other remedies or compensation.

4.8 DOCUMENTS AND SAMPLES AT THE SITE. **CERTIFYING "AS-BUILTS"**. The Contractor shall maintain at the site for the Owner, one record copy of the Drawings, Specifications, Addenda, Change Orders and other Modifications, in good order and marked weekly to record changes and selections made during construction, as well as approved Shop Drawings, Product Data, Samples and similar submittals. These aforesaid items shall be available to the A/E and shall be delivered to the A/E for submittal to the Owner upon completion of the Work, signed by the Contractor, certifying that they show complete and exact "as-built" conditions, stating sizes, kind of materials, vital piping, conduit locations and similar matters. All notes of encountered or changed conditions shall be included.

4.9 SHOP DRAWINGS, PRODUCT DATA AND SAMPLES.

4.9.1 NOT CONTRACT DOCUMENTS. Shop Drawings, Product Data, Samples and similar submittals are not Contract Documents. The submittal shall demonstrate, for those portions of the Work for which the submittal is required, the way the Contractor proposes to conform to the information given and the design concept expressed in the Contract Documents.

4.9.2 PROMPTNESS. The Contractor shall review, approve and submit to the A/E, Shop Drawings, Product Data, Samples and similar submittals required by the Contract Documents with reasonable promptness and in such sequence as to cause no delay in the Work, or the activities of the Owner or separate contractors.

4.9.3 NOT PERFORM UNTIL A/E APPROVES.

The Contractor shall perform no portion of the Work requiring submittal and review of Shop Drawings, Product Data, Samples or similar submittals until the respective submittal has been approved in writing by the A/E or approved Owner representative. Such Work shall be in accordance with the approved submittals.

4.9.4 REPRESENTATIONS BY CONTRACTOR. By approving and submitting Shop Drawings, Product Data, Samples and similar submittals, the Contractor represents that the Contractor has determined and verified materials, field measurements and field construction criteria related thereto, and has checked and coordinated the information contained within such submittals with the requirements of the Work and of the Contract Documents.

4.9.5 CONTRACTOR'S LIABILITY. The Contractor shall not be relieved of responsibility for deviations from the requirements of the Contract Documents by the A/E's approval of Shop Drawings, Product Data, Samples or similar submittals unless the Contractor has specifically informed the A/E in writing of such deviation at the time of the submittal and the A/E has given written approval to the specific deviation. The

Contractor shall not be relieved of responsibility for errors or omissions in Shop Drawings, Product Data, Samples or similar submittals by the A/E's review and comment.

4.9.6 DIRECT SPECIFIC ATTENTION TO REVISIONS. The Contractor shall direct specific attention in writing to all revisions on resubmitted Shop Drawings, Product Data, Samples or similar submittals, except those requested by the A/E and indicated on previous submittals.

4.9.7 INFORMATIONAL SUBMITTALS.

Informational submittals upon which the A/E is not expected to take responsive action may be so identified in the Contract Documents.

4.9.8 RELIANCE ON PROFESSIONAL

CERTIFICATION. When professional certification of performance criteria of materials, systems or equipment is required by the Contract Documents, the Owner and A/E shall be entitled to rely upon the accuracy and completeness of such calculations and certifications. If a professional stamp is required, the professional shall be licensed in the State of Utah unless otherwise approved by the Owner in writing. Likewise, the Contractor is entitled to rely upon the accuracy and completeness of the calculations made by the A/E in developing the Contract Documents, unless a Contractor of ordinary skill and expertise for the type of Work involved would know that such is inaccurate or incomplete and therefore must immediately notify the Owner in writing.

4.10 USE OF SITE.

4.10.1 IN GENERAL. The Contractor shall confine operations at the site to areas permitted by the Contract Documents, law, ordinances, resolutions, rules and regulations, and permits and shall not unreasonably encumber the site with materials or equipment. Contractor shall take all reasonable means to secure the site, protect the site and protect the Work from any damage. The site shall be left free and clear of

refuse, equipment, materials, etc. and the site shall not be subject to spilled liquids and chemicals, toxic or otherwise. Should such an incident occur while the Contractor has control of the site, the Contractor shall be responsible to clean the site and pay all associated costs, fines and penalties. Notwithstanding this, Contractor is not responsible for any damage to the site or the Work to the extent caused by the Owner or the Owner's agents.

4.10.2 ACCESS TO NEIGHBORING PROPERTIES.

The Contractor shall not, except as provided in the Contract Documents or with the Owner's advance written consent when necessary to perform the Work, interfere with access to properties neighboring the Project site by the owners of such properties and their respective tenants, agents, invitees and guests.

4.11 ACCESS TO WORK. The Contractor shall provide the Owner and A/E access to the Work in preparation and progress, wherever located.

4.12 ROYALTIES AND PATENTS. The Contractor shall pay all royalties and license fees. The Contractor shall defend suits or claims for infringement of patent rights and shall hold the Owner and A/E harmless from loss on account thereof, but shall not be responsible for such defense or loss when a particular design, process or product of a particular manufacturer or manufacturers is required by the Contract Documents. However, if the Contractor has reason to believe that the required design, process or product is an infringement of a patent, the Contractor shall be responsible for such loss unless such information is promptly furnished to the Owner in writing.

4.13 INDEMNIFICATION.

4.13.1 IN GENERAL.

(1) To the fullest extent permitted by law, the Contractor shall indemnify and hold harmless the Owner, it's institutions, agencies, departments, divisions, authorities, and

instrumentalities, boards, commissions, elected officers, employees, agents, or appointed authorized volunteers (hereinafter the above listing of entities and persons is referred to an "indemnities") from and against every kind and character of claims, damages, losses and expenses, including but not limited to attorneys' fees, and including those events covered under the blanket Contractual Liability Coverage required under the Contract Documents, arising out of or resulting from any act or omission in the performance of the Work including the work of all the Subcontractors and their employees, provided that any such claim, damage, loss or expense is caused in whole or in part by the negligent or wrongful act or omission of the Contractor, any Subcontractor, and their employees, provided that any such claim, damage loss or expense is caused in whole or in part by the negligent or intentional act or omission of the Contractor, any Subcontractor, or anyone directly or indirectly employed or the agent of any of them or anyone for whose acts any of them may be liable, regardless of whether or not it is caused in part by a party indemnified hereunder. The Contractor shall defend all actions brought upon such matters to be indemnified hereunder and pay all costs and expenses incidental thereto, but the Owner shall have the right, at its option, to participate in the defense of any such action without relieving the Contractor of any obligation hereunder. Notwithstanding any of the above, to the extent the Contractor is complying with a written directive from the Owner, that is not based on Contractor's recommendation, the Contractor shall not be held liable under the indemnification provision of this Agreement if the Contractor has promptly disagreed with the written directive by delivering such objection to the Owner in writing.

(2) Such obligation shall not be construed to negate, abridge, or otherwise reduce any other right or obligation of indemnity which would otherwise exist as to any party or person under Contract Documents.

(3) In claims against any person or entity indemnified under this Article 4.13 by an employee of the Contractor, Subcontractor, anyone directly or indirectly employed by them or anyone for whose acts they may be liable, the indemnification obligation under this Article 3.13 shall not be limited by a limitation on amount or type of damages, compensation or benefits payable by or for the Contractor or Subcontractor under workers' or workmen's compensation acts, disability benefits acts or other employee benefit acts.

ARTICLE 5. SUBCONTRACTORS.

5.1 AWARD OF SUBCONTRACTS AND OTHER CONTRACTS FOR PORTIONS OF THE WORK.

5.1.1 APPROVAL REQUIRED.

- (1) Listing of Subcontractors shall be as required by U.C.A. 63A-5-208 as amended and/or as stated in the Contract Documents, including but not limited to the "Owner Subcontractors List Form".
- (2) The Contractor shall not contract with a proposed person or entity to whom the Owner has made a reasonable and timely objection. The Contractor shall not be required to contract with anyone to whom the Contractor has made reasonable objection.

5.1.2 BUSINESS AND LICENSING

REQUIREMENTS. All Subcontractors used by the Contractor shall comply with all applicable business and licensing requirements.

- **5.1.3 SUBSEQUENT CHANGES**. After the lapse of twenty-four (24) hours from the bid opening, the Contractor may change its listed Subcontractors only with written approval of the Owner.
- (1) Owner will pay the additional costs for an Owner requested change in subcontractor if all of the following are met:

- (a) If the Owner in writing requests the change of a subcontractor;
- (b) The original subcontractor is a responsible subcontractor that meets the requirements of the Contract Documents; and
- (c) The original subcontractor did not withdraw as a subcontractor on the project.
- (2) In all other circumstances, the Contractor shall pay the additional cost for a change in a subcontractor.
- **5.1.4 BONDING OF SUBCONTRACTORS.** Subcontractors as identified by Owner in the procurement documents, may be required to submit performance and payment bonds to cover the full extent of their portion of the Work. This provision does not in any way limit the right of the Contractor to have subcontractors at any tier be required to have a performance and/or payment bond.

5.2 SUBCONTRACTUAL RELATIONS.

- **5.2.1 COMPLY WITH CONTRACT DOCUMENTS**. By appropriate enforceable agreement, and to the extent it can be practically applied, the Contractor shall require each Subcontractor to be bound to the Contractor by the terms of the Contract Documents, and to assume toward the Contractor all the obligations and responsibilities which the Contractor, by these Documents, assumes towards the Owner and A/E.
- **5.2.2 RIGHTS**. Each Subcontractor agreement shall preserve and protect the rights of the Owner and A/E under the Contract Documents with respect to the Work to be performed by the Subcontractor so that subcontracting thereof will not prejudice such rights, and shall allow to the Subcontractor, unless specifically provided otherwise in the Subcontractor agreement, the benefit of all rights and remedies against the

Contractor that the Contractor, by the Contract Documents, has against the Owner.

- **5.2.3 SUB-SUBCONTRACTORS.** The Contractor shall require each Subcontractor to enter into similar agreements with its Subcontractors which complies with the requirements of Paragraphs 5.2.1 and 5.2.2 hereinabove.
- **5.2.4 DOCUMENT COPIES.** The Contractor shall make available to each proposed Subcontractor, prior to execution of the subcontract agreement, copies of the Contract Documents to which the Subcontractor will be bound. Subcontractors shall similarly make copies of applicable portions of the Contract Documents available to their respective proposed Subcontractors.
- 5.3 CONTINGENT ASSIGNMENT OF SUBCONTRACTS.
- 5.3.1 CONDITIONS FOR ASSIGNMENT TO Each subcontract agreement for a OWNER. subcontractor at any tier for a portion of the Work is assigned by the Contractor to the Owner provided that the assignment is effective only after termination of the Contract by the Owner for cause pursuant to Article 12.2 or stoppage of the Work by Owner pursuant to Article 12.5, and only for those subcontract agreements which the Owner accepts by notifying Subcontractor in writing. The subcontract shall be equitably adjusted to meet the new conditions of the work.

ARTICLE 6. PROTECTION OF PERSONS AND PROPERTY.

- **6.1 SAFETY OF PERSONS AND PROPERTY.**
- **6.1.1 CONTRACTOR RESPONSIBILITY**. The Contractor shall be solely responsible for initiating, maintaining and supervising all safety precautions and programs in connection with the performance of the Contract. The Contractor

shall take all reasonable precautions for the safety of, and shall provide reasonable protection to prevent damage, injury or loss to:

- (1) Employees on the Work and other persons who may be affected thereby;
- (2) The Work and materials and equipment to be incorporated therein, whether in storage on or off the site, under care, custody or control of the Contractor or a Subcontractor: and
- (3) Other property at the site or adjacent thereto, such as trees, shrubs, lawns, walks, pavements, roadways, structures and utilities not designated for removal, relocation or replacement in the course of construction.
- 6.1.2 SAFETY PROGRAM, PRECAUTIONS. The Contractor shall institute a safety program at the start of construction to minimize accidents. Said program shall continue to the final completion of the Project and conform to applicable laws and regulations including the Utah Occupational Safety and Health Rules and Regulations as published by the Utah Industrial Commission -UOSH Division. The Contractor shall post signs, erect barriers, and provide those items necessary to implement the safety program. As soon as the Contractor proceeds with the Work, the Contractor shall have all workers and all visitors on the site wear safety hard hats, as well as all other appropriate safety apparel such as safety glasses and shoes, and obey all safety rules and regulations and statutes. The Contractor shall post a sign in a conspicuous location indicating the necessity of wearing hard hats and the Contractor shall loan such hats to visitors.
- **6.1.3 COMPLIANCE WITH LAWS.** The Contractor shall give notices and comply with applicable laws, ordinances, rules, regulations and lawful orders of public authorities bearing on safety of persons or property or their protection from damage, injury or loss. In particular, the Contractor shall comply with all applicable provisions of Federal, State and

municipal safety laws, rules and regulations as well as building codes to prevent accidents or injury to persons on, about, or adjacent to the premises where the Work is being performed.

6.1.4 ERECT AND MAINTAIN SAFEGUARDS.

The Contractor shall erect and maintain, as required by existing conditions and performance of the Contract, reasonable safeguards for safety and protection, including effective fences, posting danger signs and other warnings against hazards, promulgating safety regulations and notifying owners and users of adjacent sites and utilities.

- **6.1.5 UTMOST CARE**. When use or storage of explosives or other hazardous materials or equipment or unusual methods are necessary for execution of the Work, the Contractor shall exercise utmost care and carry on such activities under supervision of properly qualified personnel.
- **6.1.6 PROMPT REMEDY.** The Contractor shall promptly remedy damage and loss (other than damage or loss insured under property insurance required by the Contract Documents) to property referred to in Paragraph 6.1.1 of these General Conditions caused in whole or in part by the Contractor, a Subcontractor, or anyone directly or indirectly employed by any of them, or by anyone for whose acts they may be liable and for which the Contractor is responsible under said Paragraph 6.1.1, except to the extent such damage or loss is directly due to errors in the Contract Documents or caused by agents or employees of the A/E or Owner. The foregoing obligations of the Contractor are in addition to the Contractor's obligations under the Contract Documents.
- **6.1.7 SAFETY DESIGNEE**. The Contractor shall designate a responsible member of the Contractor's organization at the site whose duty shall be the prevention of accidents, damage, injury or loss. This person shall be the Contractor's superintendent unless otherwise

designated by the Contractor in writing to the Owner and A/E.

6.1.8 LOAD SAFETY.

The Contractor shall not load or permit any part of the construction or site to be loaded so as to endanger its safety.

- **6.1.9 OFF-SITE RESPONSIBILITY**. In addition to its other obligations under this Article 6, the Contractor shall, at its sole cost and expense, promptly repair any damage or disturbance to walls, utilities, streets, ways, sidewalks, curbs and the property of the Owner and third parties (including other governmental agencies) resulting from the performance of the Work, whether by it or by its Subcontractors at any tier. The Contractor shall not cause materials, including soil and debris, to be placed or left on streets or ways.
- **6.1.10 EMERGENCIES**. In an emergency affecting safety of persons or property, the Contractor shall act, at the Contractor's discretion, to prevent threatened damage, injury or loss. Contractor shall promptly notify the Owner Representative of the action taken.
- 6.2 HAZARDOUS MATERIALS. In the event the Contractor encounters on the site material reasonably believed to be asbestos or polychlorinated biphenyl (PCB) or any other hazardous waste or substance which may endanger the health of those persons performing the Work or being on the site, the Contractor shall immediately stop Work in the area affected and immediately report the condition to the Owner Representative and A/E by phone with a follow-up document in writing. The Work in the affected area shall be resumed when written direction is provided by the Owner Representative. Except to the extent provided otherwise in the Contract Documents or if the presence of hazardous materials is due to the fault of the Contractor, the Contractor shall not be required to perform without the Contractor's consent, any Work relating to asbestos,

polychlorinated biphenyl (PCB) or any other hazardous waste or substance.

Owner shall procure a licensed abatement contractor qualified to remove the hazardous material. The abatement contractor shall submit notification of demolition to the Utah Division of Air Quality. Abatement contractor shall pay the notification fee. A copy of the hazardous material survey report shall be available to all persons who have access to the construction site.

- 6.3 HISTORICAL AND **ARCHEOLOGICAL CONSIDERATIONS**. In the event the Contractor knows or should have known of any cultural, historical or archeological material that is either recognized as an item to be protected under Federal, State, or local law or regulation, or is an item of obvious value to the local community or State of Utah, the Contractor shall cease any work that would interfere with such discovery and immediately report the condition to the Owner Representative and A/E by phone with a follow-up document in writing. Work shall resume based upon the direction of the Owner Representative. Contractor cooperation with any Owner recognized archaeologist or other cultural/historical expert is required.
- **6.4 CONTRACTOR LIABILITY**. If the Contractor fails in any of its obligations in Articles 6.1 through 6.3 above, the Contractor shall be liable to any damages to Owner or any third party resulting from such noncompliance. The Contractor shall also be liable for any mitigation or restoration effort resulting from such noncompliance. To the extent all the following is met, the Contractor may treat the discovery of such material similarly to an unforeseen condition:
- **6.4.1** The discovery of such material is reasonably unforeseeable given the site conditions that the Contractor should have been aware;

- **6.4.2** The presence of such material was not identified in any part of the Contract Documents;
- **6.4.3** The Contractor has undertaken all proper action to mitigate any impact of such discovery on the critical path or monies related to the Project;
- **6.4.4** The discovery affects the critical path or contract price from that which was contemplated by the Contract Documents; and
- **6.4.5** The requirements of 7.1.5 and the Contract documents are met.

ARTICLE 7. MODIFICATIONS, REQUEST FOR INFORMATION, PROPOSED CHANGE ORDER, PRELIMINARY RESOLUTION EFFORTS AND CLAIMS PROCESS.

7.1 MODIFICATIONS: IN GENERAL.

- **7.1.1 TYPES OF MODIFICATIONS AND LIMITATIONS.** Changes in the Work may be accomplished after execution of the Contract, and without invalidating the Contract, by Change Order, Construction Change Directive or ASI, subject to the limitations stated in this Article 7 and elsewhere in the Contract Documents. The Contractor must have a written Modification executed by Owner under this Article 7 prior to proceeding with any Work sought to be an extra.
- **7.1.2 BY WHOM ISSUED**. A Change Order or Construction Change Directive shall be issued by the Owner Representative. An ASI is issued by the A/E or by the Owner Representative.

The A/E shall prepare Change Orders and Construction Change Directives with specific documentation and data for the Owner's approval and execution in accordance with the Contract Documents, and may issue ASIs not involving an adjustment in the contract sum or an extension of the Contract Time which are

not inconsistent with the intent of the Contract Documents.

- **7.1.3 CONTRACTOR TO PROCEED UNLESS OTHERWISE STATED.** Changes in the Work shall be performed under applicable provisions of the Contract Documents, and the Contractor shall proceed promptly, unless otherwise provided in the Change Order, Construction Change Directive or ASI.
- **7.1.4 ADJUSTING UNIT PRICES**. If unit prices are stated in the Contract Documents or subsequently agreed upon, and if quantities originally contemplated are so changed in a proposed Change Order or Construction Change Directive that application of such unit prices to quantities of Work proposed will cause a substantial inequity to the Owner or Contractor, the applicable unit prices may be equitably adjusted.
- 7.1.5 SPECIAL NOTICES REQUIRED IN ORDER TO BE ELIGIBLE FOR ANY CONTRACT MODIFICATION. In order to be eligible for any Modification under this Article 7, the Contractor must have met the following special notice requirements:
- The Contractor must file a written notice with the Owner Representative within seven (7) calendar days of that the Contractor knew or should have known of a site condition described below or the Contractor shall be deemed to waive any right to file any PCO, PRE or Claim for additional monies or time related to such

(1) CONCEALED OR UNKNOWN CONDITIONS.

(a) If the Contractor encounters unknown and reasonably unforeseeable subsurface or otherwise concealed physical conditions, including hazardous or historical/cultural materials under Article 6, which differ materially from those indicated by the Contract Documents or a site inspection; or

condition:

- (b) If the Contractor encounters unknown physical conditions of an unusual nature which differ materially from those ordinarily found to exist and generally recognized as inherent in construction activities of the character provided for in the Contract Documents.
- (2) INCREASE IN CONTRACT TIME. If the Contractor encounters a situation in which the Contractor knows or should have known that such situation would cause a delay, disruption, interruption, suspension or the like to the Project, the Contractor must file a notice with the Owner Representative within seven (7) working days of when the Contractor knew or should have known of such circumstance or the Contractor shall be deemed to waive any right to file any PCO, PRE or Claim for additional monies or time related to such circumstance. To the extent the Owner Utah is damaged by the failure of the Contractor to provide such notice after the Contractor knows or should have known of such circumstance, the Contractor shall be liable for liquidated damages attributable thereto, as well as any damages to the Owner that are allowable in addition to liquidated damages.

7.2 CONTRACTOR INITIATED REQUESTS.

- **7.2.1 THE REQUEST FOR INFORMATION, RFI, PROCESS AND TIME TO FILE.** The Contractor may file an RFI with the A/E regarding any concern which will assist the Contractor in the proper completion of the Work including, but not limited to issues related to the Contract Documents, plans and specifications. The RFI shall be filed with the A/E in a timely manner so as not to prejudice the Owner as to the quality, time or money related to the Work.
- **7.2.2 PROPOSED CHANGE ORDER ("PCO").** Within twenty-one (21) days after the Contractor knows or should have known of a situation or concern where the Contractor is going to request additional monies or time, the Contractor must file a Proposed Change Order ("PCO") with the Owner Representative, or the

- Contractor shall be deemed to waive any right to claim additional monies or time related to such situation or concern. The PCO shall include all available documentation supporting the PCO available to the Contractor at the time of filing and the Contractor shall thereafter diligently pursue the supplementation(s) of such documentation and promptly deliver such supplementation(s) to the Owner Representative.
- (1) OWNER REPRESENTATIVE RESPONSE. One of the following may occur after a PCO is filed with the Owner Representative:
- (a) The Owner Representative, after considering any input by the A/E, may reach an agreement with the Contractor and issue a Change Order.
- (b) The Owner, after considering any input by the A/E, may issue a Construction Change Directive.
- (c) If the Owner Representative, after considering any input by the A/E, disagrees with the Contractor's PCO, the Owner representative may seek additional information or verification from the Contractor, the A/E or other sources, may negotiate with the Contractor, may issue a Change Order upon such later agreement, may retract the PR, or may issue a Construction Change Directive.
- (d) If a Construction Change Directive is issued which identifies the Owner Representative's position in regard to the subject contract sum and/or time adjustment or if the PCO is denied by the Owner Representative, the Contractor must file a PRE under Article 7.7 below no later than twenty-one (21) days after the Contractor's receipt of the Construction Change Directive or such denial of the PCO. Failure to file a PRE in these instances shall be deemed to waive any right to additional time or money related to the PCO, Construction Change Directive or denial of the PCO. Such waiver shall entitle the Owner to convert the Construction Change Directive into a

Change Order, whether or not executed by the Contractor.

If the Construction Change Directive leaves open the determination of additional time or money related to the directed change, then the time period for commencement of filing the PRE shall not accrue until such time as the Owner has conveyed to the Contractor a position as to the time and money owing as a result of the directed change.

The A/E must continually work with the Owner in providing data, documentation and efforts to resolve the issues related to the PR.

7.3 PROPOSAL REQUEST INITIATED BY OWNER.

Owner may file a Proposal Request with the Contractor seeking information, data and/or pricing relating to a change in the contract time and or monies owing for particular scope changes or other modifications to the Contract Documents. The PR shall provide a time limit for the Contractor to file a response with the A/E and the Owner Representative. If a proposal is not timely provided by the Contractor, Owner may calculate the Change Order under Article 7.4.2 below. Upon such timely receipt of the proposal, one of the following shall occur:

7.3.1 IF AGREEMENT, CHANGE ORDER ISSUED. The Owner Representative, after considering any input by the A/E, may reach an agreement with the Contractor and issue a Change Order.

7.3.2 IF **DISAGREEMENT**. If the Owner Representative disagrees with the Contractor's proposal, after considering any input from the A/E, the Owner representative may seek additional information or verification from the Contractor or other sources, may negotiate with the Contractor, may issue a Change Order upon such later agreement, may retract the PR, or may issue a Construction Change Directive. If a Construction Change Directive is issued which identifies the Owner representative's position in regard to the subject contract sum and/or time

adjustment, the Contractor must file a PRE within twenty-one (21) days of the Contractor's receipt of the Construction Change Directive, or the Contractor shall be deemed to waive any such request for additional time or money as a result of the issuance of the Construction Change Directive. Such waiver shall entitle the Owner to convert the Construction Change Directive into a Change Order, whether or not executed by the Contractor. If the Construction Change Directive leaves open the determination of additional time or money related to the directed change, then the time period for commencement of filing the PRE shall not accrue until such time as the Owner has conveyed to the Contractor a position as to the time and money owing as a result of the directed change.

7.4 EVALUATION OF PROPOSAL FOR ISSUING CHANGE ORDERS.

7.4.1 ADJUSTING SUM BASED UPON AGREEMENT. If the Change Order provides for an adjustment to the Contract Sum, the adjustment shall be based on the mutual agreement of the Contractor and Owner, including any terms mandated by unit price agreements or other terms of the Contract Documents.

7.4.2 OWNER RESOLUTION OF SUM AND STANDARDS IN THE ABSENCE OF AN AGREEMENT UNDER PARAGRAPH 7.4.1. In the absence of an agreement under Paragraph 7.4.1 above, the adjustment shall be based on an itemized accounting of costs and savings supported by appropriate data. Unless otherwise provided in the Contract Documents, costs for the purposes of this Paragraph shall be limited to the following:

(1) All direct and indirect costs of labor; including workers compensation insurance, social security and other federal and state payroll based taxes, and payroll based fringe benefits paid by Contractor so long as they are reasonable and no higher than that charged to other clients;

- (2) Costs of materials, on-site temporary facilities, supplies and equipment (except hand tools) required for or incorporated into the work;
- (3) Rental costs of machinery, equipment, tools (except hand tools), and on-site temporary facilities, whether rented from the Contractor or others;
- (4) Costs of permits and other fees, sales, use or similar taxes related to the Work;
- **(5)** Additional costs of field supervision and field office personnel directly attributable to the change; and
- (6) Overhead and profit by the following liquidated formula which is not a penalty but a reasonable calculation agreed upon at the time of execution of the Contractor's Agreement, and provided by formula herein due to the fact that the actual amount due for said overhead and profit cannot easily be ascertained at the time of such execution. The markups in 7.4.2(6)(a) and (b) below are to cover the Contractor's additional payment and performance bond premiums, insurance premiums not specified under Paragraph 7.4.2(1), home office and onsite overhead and profit. Overhead and profit includes, but is not limited to the Contractor's Project Manager and Cost Estimator. Each request for pricing shall stand on its own and not be combined with other requests for pricing in determining the allowed markup described below. A particular request for pricing shall include all items reasonably related together and determinable at the time of the request. If several unrelated requests for pricing are grouped together in a single Change Order, each request for pricing will be considered separately for purposes of calculating the markup under the following formula:
- (a) A markup of 15% shall be applied to the cost of each individual charge up to \$20,000 in cost, but in no case shall the markup be less than \$150;

- (b) A markup of 10% shall be applied to the portion of the cost of each individual charge in excess of \$20,000;
- (c) Subcontractors at any tier shall be entitled to markup their costs related to a Change Order with the same percentages as specified in Paragraphs 7.4.2(6)(a) and (b) above, except that the minimum markup shall be \$50 for any individual change.
- **7.4.3 CREDITS**. The amount of credit to be allowed by the Contractor to the Owner for a deletion or change which results in a net decrease in the Contract Sum shall be actual net cost as confirmed to Owner based upon corroboration by an appropriate source.

7.5 CONSTRUCTION CHANGE DIRECTIVES.

- 7.5.1 WHEN USED AND CONTRACTOR'S RIGHT TO CHALLENGE. A Construction Change Directive may be issued by the Owner Representative in the case of a need for the Work to commence. If the Construction Change Directive leaves open the determination of additional time or money related to the directed change, then the Construction Change Directive shall indicate the timeframe(s) in which further information is to be provided to resolve the matter. At any time that the Owner and the Contractor agree upon the time and money related to a Construction Change Directive, a Change Order shall be executed by the parties. Additionally, the Construction Change Directive may be converted to a Change Order under Paragraph 7.2.2 or Article 7.3 above.
- **7.5.2 PROCEED WITH WORK AND NOTIFY Owner ABOUT ADJUSTMENT METHOD.** Upon receipt of a Construction Change Directive, the Contractor shall promptly proceed with the change in the Work involved.
- **7.5.3 INTERIM PAYMENTS BY OWNER.** Pending the final determination of the total cost of the Construction Change Directive, Owner

shall pay any undisputed amount to the Contractor.

7.6 A/E'S SUPPLEMENTAL INSTRUCTION (Commonly referred to as an "ASI"). The A/E may at any time that is consistent with maintaining the quality, safety, time, budget and function of the Work, issue to the Contractor a supplemental instruction ("ASI") after approval from the Owner Representative is obtained.

The Contractor must file with the Owner Representative a PCO under Paragraph 7.2.2 above, within 21 calendar days of the Contractor's receipt of the ASI, or the Contactor shall be deemed to have waived any right to additional time or monies as a result of such ASI.

7.7 PROCEDURE FOR PRELIMINARY RESOLUTION EFFORTS.

- **7.7.1 REQUEST FOR PRELIMINARY RESOLUTION EFFORT (PRE)**. A Contractor raising an issue related to a breach of contract or an issue concerning time or money shall file a PRE as a prerequisite for any consideration of the issue by the Owner. The labeling of the notice or request shall not preclude the consideration of the issue by the Owner.
- **7.7.2 TIME FOR FILING.** The PRE must be filed in writing with the Owner Representative within twenty-one (21) days of any of the following:
- (1) Issuance of a Construction Change Directive that defines the time and sum due the Contractor but the Contractor disagrees with such assessment;
- (2) Issuance of Owner's position in regard to a Construction Change Directive that originally left open the time and/or sum due to the Contractor;
- (3) Issuance of a denial of a PCO by Owner;
- (4) In the case of a Subcontractor, after the expiration of the time period for the Contractor/Subcontractor PRE process under Paragraph 7.7.5 below; or

- (5) When the Contractor knows or should have known about any other issue where the Contractor seeks additional monies, time or other relief from the Owner.
- **7.7.3 CONTENT REQUIREMENT.** The PRE shall be required to include in writing to the extent information is reasonably available at the time of such filing:
- (1) A description of the issue;
- (2) The potential impact on cost and time or other breach of contract; and
- (3) An indication of the relief sought.
- **7.7.4 SUPPLEMENTATION**. Additional detail of the content requirement under Paragraph 7.7.3 above shall be provided later if the detail is not yet available at the initial filing as follows:
- (1) While the issue is continuing or the impact is being determined, the Contractor shall provide a written updated status report every 30 days or as otherwise reasonably requested by the Owner Representative; and
- (2) After the scope of work or other factors addressing the issue are completed, the complete information, including any impacts on time, cost or other relief requested, must be provided to the Owner Representative within twenty-one (21) days of such completion.

7.7.5 SUBCONTRACTORS.

- (1) Under no circumstances shall any provision of these Contract Documents be intended or construed to create any contractual relationship between the Owner and any Subcontractor.
- (2) The Contractor must include the provisions of this Paragraph 7.7.5 in its contract with the first tier Subcontractor, and each Subcontractor must do likewise. At the Contractor's discretion, the Contractor may allow a Subcontractor at the

2nd tier and beyond to submit the PRE directly with the Contractor.

- (3) In order for a Subcontractor at any tier to be involved with the PRE of the Owner, the following conditions and process shall apply:
- (a) The Subcontractor must have attempted to resolve the issue with the Contractor including the submission of a PRE with the Contractor.
- (b) The Subcontractor must file a copy of the PRE with the Owner Representative;
- (c) The PRE to the Contractor must meet the time, content and supplementation requirements of Paragraphs 7.7.2,
- 7.7.3 and 7.7.4. The triggering event for a Subcontractor to file a PRE shall be the time at which the issue cannot be resolved through the normal business practices associated with the contract, excluding arbitration and litigation;
- (d) The PRE submitted to the Contractor shall only be eligible for consideration in the Owner's PRE process to the extent the issue is reasonably related to the performance of the Owner or an entity for which the Owner is liable;
- (e) The Contractor shall resolve the PRE to the satisfaction of the Subcontractor within sixty (60) days of its submittal to the Contractor or such other time period as subsequently agreed to by the Subcontractor in writing. If the Contractor fails to resolve the PRE with the Subcontractor within such required time period, the Subcontractor may submit in writing the PRE with the Contractor and the Owner. In order to be eligible for Owner consideration of the PRE, the Subcontractor must submit the PRE within twenty-one (21) days of the expiration of the time period for the Contractor/Subcontractor PRE process. The Owner shall consider the PRE as being submitted by the Contractor on behalf of the Subcontractor;

- (f) Upon such PRE being submitted, the Contractor shall cooperate with the Owner Representative in reviewing the issue;
- (g) The Owner shall not be obligated to consider any submission which is not in accordance with any provision of this Article 7.7;
- (h) The Subcontractor may accompany the Contractor in participating with the Owner regarding the PRE raised by the Subcontractor. The Owner is not precluded from meeting with the Contractor separately and it shall be the responsibility of the Contractor to keep the Subcontractor informed of any such meetings; and
- (i) Notwithstanding any provision of this Paragraph 7.7.5, a Subcontractor shall be entitled to pursue a payment bond claim.
- **7.7.6 PRE RESOLUTION PROCEDURE**. The Owner Representative may request additional information and may meet with the parties involved with the issue.

7.7.7 CONTRACTOR REQUIRED TO

CONTINUE PERFORMANCE. Pending the final resolution of the issue, unless otherwise agreed upon in writing by the Owner Representative, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents.

- **7.7.8 DECISION.** The Owner shall issue to the Contractor, and any other party brought into the process by the Owner Representative as being liable to the Owner, a written decision providing the basis for the decision on the issues presented by all of the parties within thirty (30) days of receipt of all the information required under Paragraphs 7.7.3 and 7.7.4.
- **7.7.9 DECISION FINAL UNLESS CLAIM SUBMITTED.** The decision by the Owner shall be final, and not subject to any further administrative or judicial review (not including

judicial enforcement) unless a Claim is submitted in accordance with these General Conditions.

7.7.10 EXTENSION REQUIRES MUTUAL AGREEMENT. Any time period specified in this Article 7.7 may be extended by mutual agreement of the Contractor and the Owner.

7.7.11 IF DECISION NOT ISSUED. If the decision is not issued within the thirty (30) day period, including any agreed to extensions, the issue may be pursued as a Claim.

7.7.12 PAYMENT FOR PERFORMANCE.

- (1) Except as otherwise provided in the Contract Documents, any final decision where the Owner is to pay additional monies to the Contractor, shall not be delayed by any PRE, Claim or appeal by another party.
- (2) Payment to the Contractor of any final decision shall be made by the Owner in accordance with the contract for the completed work.
- (3) Notwithstanding any other provision of the Contract Documents, payment to the Contractor shall be subject to any set-off, claims or counterclaims of the Owner.
- (4) Payment to the Contractor for a Subcontractor issue submitted by the Contractor shall be paid by the Contractor to the Subcontractor in accordance with the contract between the Contractor and the Subcontractor.
- (5) Any payment or performance determined owing by the Contractor to the Owner shall be made in accordance with the Contract Documents.

7.8. RESOLUTION OF CLAIM.

7.8.1 CLAIM. If the decision on the PRE is not issued within the required timeframe or if the Contractor is not satisfied with the decision, the Contractor or other party brought into the

process by the Owner, may submit a Claim in accordance with this Article 7.8 as a prerequisite for any further consideration by the Owner or the right to any judicial review of the issue giving rise to the claim.

- 7.8.2 SUBCONTRACTORS. In order for a Subcontractor to have its issue considered in the Claim process by the Owner, the Subcontractor that had its issue considered under Paragraph 7.7.5 may submit the issue as a Claim by filing it with the Contractor and the Owner within the same timeframe and with the same content requirements as required of a Claim submitted by the Contractor under this rule. The Owner shall consider the Claim as being submitted by the Contractor on behalf of the Subcontractor. Under no circumstances shall any provision of these General Conditions or the Contract Documents be intended or construed so as to create any contractual relationship between the Owner and any Subcontractor.
- (1) Upon such Claim being submitted, the Contractor shall fully cooperate with the Director, the person(s) evaluating the claim and any subsequent reviewing authority.
- (2) The Director shall not be obligated to consider any submission which is not in accordance with this Paragraph 7.8.2.
- (3) The Subcontractor may accompany the Contractor in participating with the Director, the person(s) evaluating the Claim and any subsequent reviewing authority regarding the Claim. The Director, the person(s) evaluating the Claim and any subsequent reviewing authority is not precluded from meeting with the Contractor separately, and it shall be the responsibility of the Contractor to keep the Subcontractor informed of any such meetings and matters discussed.
- **(4)** Notwithstanding any provision of this Article 7.8, a Subcontractor shall be entitled to pursue a payment bond claim.

7.8.3 TIME FOR FILING. The Claim must be filed in writing promptly with the Director, but in no case more than twenty-one (21) days after the decision is issued on the PRE under Paragraph 7.7.8 or no more than twenty-one (21) days after the thirty (30) day period under Paragraph 7.7.11 has expired with a decision not issued, whichever is later.

7.8.4 CONTENT REQUIREMENT. The written Claim shall include:

- (1) A description of the issues in dispute;
- (2) The basis for the Claim, including documentation and analysis required by the contract and applicable law and rules that allow for the proper determination of the Claim;
- (3) A detailed cost estimate for any amount sought, including copies of any related invoices; and
- (4) A specific identification of the relief sought.
- **7.8.5 EXTENSION OF TIME TO SUBMIT DOCUMENTATION.** The time period for submitting documentation and any analysis to support a Claim may be extended by the Director upon written request of the claimant showing just cause for such extension, which request must be included in the initial Claim submittal.
- **7.8.6 CONTRACTOR REQUIRED TO CONTINUE PERFORMANCE.** Pending the final determination of the Claim, including any judicial review or appeal process, and unless otherwise agreed upon in writing by the Director, the Contractor shall proceed diligently with performance of the Contract and the Owner shall continue to make payments in accordance with the Contract Documents.
- **7.8.7** AGREEMENT OF CLAIMANT ON METHOD AND PERSON(S) EVALUATING THE CLAIM. The Director shall first attempt to reach agreement with the claimant on the method and person(s) to evaluate the Claim. If such

agreement cannot be made within fourteen (14) days of filing of the Claim, the Director shall select the method and person(s), considering the purposes described in as per Utah Rule R23-26-1. Unless agreed to by the Director and the claimant, any selected person shall not have a conflict of interest or appearance of impropriety. Any party and the person(s) evaluating the Claim has a duty to promptly raise any circumstances regarding a conflict of interest or appearance of impropriety. If such a reasonable objection is raised, and unless otherwise agreed to by the Director and the claimant, the Director shall take appropriate action to eliminate the conflict of interest or appearance of impropriety. The dispute resolution methods and person(s) may include any of the following:

- (1) A single expert and/or hearing officer qualified in the field that is the subject of the Claim;
- (2) An expert panel, consisting of members that are qualified in a field that is the subject of the Claim;
- (3) An arbitration process which may be binding if agreed to by the parties to the Claim;
- (4) A mediator; or
- (5) Any other method that best accomplishes the purposes set forth in Utah Rule R23-26-1.
- 7.8.8 THE EVALUATION PROCESS, TIMEFRAMES OF EVALUATOR(S), DIRECTOR'S DETERMINATION, ADMINISTRATIVE APPEAL TO THE EXECUTIVE DIRECTOR AND JUDICIAL REVIEW. The Claim shall be evaluated, the timeframe for specific events related to the person(s) evaluating the Claim, the Director's determination, and any appeal to the Executive Director.
- **7.8.9** APPEAL PROCESS PREREQUISITE FOR FURTHER CONSIDERATION OR JUDICIAL REVIEW. The administrative appeal to the Executive Director is a prerequisite for any further consideration by the Owner, or to judicial

review of the issue giving rise to the Claim. It shall be considered that the Contractor, or another party brought into the process by the Owner, has not exhausted its administrative remedies if such an administrative appeal is not undertaken.

7.9 PAYMENT OF CLAIM.

- **7.9.1** When a standalone component of a Claim has received a final determination, and is no longer subject to review or appeal, that amount shall be paid in accordance with the payment provisions of the Contract Documents or judicial order.
- **7.9.2** When the entire Claim has received a final determination, and is no longer subject to review or appeal, the full amount shall be paid within fourteen (14) days of the date of the final determination unless the work or services has not been completed, in which case the amount shall be paid in accordance with the payment provisions of the Contract Documents to the point that the work or services is completed.
- **7.9.3** The final determination date is the earlier of the date upon which the claimant accepted the settlement in writing with an executed customary release document and waived its rights of appeal, or the expiration of the appeal period, with no appeal filed, or the determination made resulting from the final appeal.
- **7.9.4** Any final determination where the Owner is to pay additional monies to the Contractor shall not be delayed by any appeal or request for judicial review by another party brought into the process by the Owner as being liable to the Owner.
- **7.9.5** Notwithstanding any other provision of the Contract Documents, payment of all or part of a Claim is subject to any set-off, claims or counterclaims of the Owner.

- **7.9.6** Payment to the Contractor for a Subcontractor issue (Claim) deemed filed by the Contractor, shall be paid by the Contractor to the Subcontractor in accordance with the contract between the Contractor and the Subcontractor.
- **7.9.7** The execution of a customary release document related to any payment may be required as a condition of making the payment.

7.10 ALLOCATION OF COSTS OF CLAIM RESOLUTION PROCESS.

- **7.10.1** In order to file a Claim, a claimant must pay a \$1500 filing fee to the Owner. When the Claim is a pass-through from a Subcontractor in accordance with Paragraph 7.7.5, the payment of the fee shall be made by the Subcontractor.
- **7.10.2** Unless otherwise agreed to by the parties to the Claim, the costs of resolving the Claim shall be allocated among the parties on the same proportionate basis as the determination of financial responsibility for the Claim.
- **7.10.3** The costs of resolving the Claim that are subject to allocation include the claimant's filing fee, the costs of any person(s) evaluating the Claim, the costs of making any required record of the process, and any additional testing or inspection procured to investigate and/or evaluate the Claim.
- **7.10.4** Each party is responsible for its own attorney fees.
- **7.11 ALTERNATIVE PROCEDURES.** To the extent otherwise permitted by law, if all parties to a Claim agree in writing, a protocol for resolving a Claim may be used that differs from the process described in this Article 7.

7.12 IMPACT ON FUTURE SELECTIONS.

7.12.1 The presentation of a good faith and non-frivolous issue or Claim shall not be considered by the Owner's selection process for a future award of contract; and

- **7.12.2** The submission of a bad faith and frivolous issue or Claim, or the failure by a Contractor to facilitate resolution of a Claim, may be considered in the Owner's evaluation of performance.
- 7.13 OWNER'S RIGHT TO HAVE ISSUES, DISPUTES OR CLAIMS CONSIDERED. Articles 7.7 through 7.13 above do not limit the right of Owner to have any of its issues, disputes or claims considered. Owner reserves all rights to pursue its issues, disputes or claims in law or equity including, but not limited to, any or all of the following: damages, delay damages and impacts, losses, liability, patent or latent defects, or failure to perform under the Contract Documents. If the Director appoints an expert or a panel to consider any such issue(s), dispute(s) or claim(s) of Owner, the Contractor shall cooperate with such expert or panel process.

ARTICLE 8. PAYMENTS AND COMPLETION.

8.1 SCHEDULE OF VALUES. With the first Application for Payment, the Contractor shall submit to the A/E and the Owner Representative a schedule of values allocated to all the various portions of the Work. The Schedule of Values shall be submitted on the form approved and provided by Owner. The A/E shall make recommendations to the Owner Representative regarding the Schedule of Values including any suggested modifications. When approved, including any approved modifications, by the Owner Representative, it shall be the basis for future Contractor Applications for Payments. The Contractor shall not be entitled to payment until receipt and acceptance of the Schedule of Values.

8.2 APPLICATIONS FOR PAYMENT.

8.2.1 IN GENERAL. The following general requirements shall be met:

- (1) The Contractor shall submit to the A/E an itemized Application for Payment for Work completed in accordance with the schedule of values and that reflects retainage as provided for in the Contractor's Agreement. The Application for Payment shall be on a special form approved and provided by Owner.
- (2) Such application shall be supported by such data substantiating the Contractor's right to payment as the Owner or A/E may require. Said data may include, but is not limited to, copies of requisitions from Subcontractors.
- (3) Such applications may include requests for payment pursuant to approved Change Orders or Construction Change Directives.
- (4) Such applications may not include requests for payment for portions of the Work performed by a subcontractor when the Contractor does not intend to pay to a Subcontractor because of a dispute or other reason.
- (5) In executing the Application for Payment, the Contractor shall attest that subcontractors involved with prior applications for payment have been paid, unless the Contractor provides a detailed explanation why such payment may not have occurred. Owner reserves the right to require the Contractor to submit a payment waiver from one or more subcontractors.
- 8.2.2 PAYMENT **FOR** MATERIAL **AND EQUIPMENT**. Unless otherwise provided in the Contract Documents, payments shall be made on account of materials and equipment delivered and suitably stored at the site for subsequent incorporation in the Work. approved in advance by the Owner and A/E, payment may similarly be made for materials and equipment suitably stored off the site at a location agreed upon in writing. Payment for materials and equipment stored on or off the site shall be conditioned upon compliance by the Contractor with procedures satisfactory to the Owner to establish the Owner's title to such materials and equipment or otherwise protect

the Owner's interest, and shall include applicable insurance, storage and transportation to the site for such materials and equipment stored off the site. The Owner may require copies of invoices or other suitable documentation.

8.2.3 WARRANTY OF TITLE. The Contractor warrants that title to all Work covered by an Application for Payment will pass to the Owner no later than the time for payment. The Contractor further warrants that upon submittal of an Application for Payment, all Work for which Certificates for Payment have been previously issued and payments received from the Owner shall, to the best of the Contractor's knowledge, information and belief, be free and clear of liens, claims, security interests or encumbrances in favor of the Contractor, Subcontractors, or other persons or entities making a claim by reason of having provided labor, materials and/or equipment relating to the Work.

8.2.4 HOLDBACK BY OWNER.

Notwithstanding anything to the contrary contained in the Contract Documents, the Owner may, as a result of the claims resolution process, withhold any payment to the Contractor hereunder if and for so long as the Contractor fails to perform any of its obligations hereunder or otherwise is in default under any of the Contract Documents.

8.3 CERTIFICATES FOR PAYMENT.

8.3.1 ISSUED BY A/E. The A/E shall within ten (10) days after receipt of the Contractor's Application for Payment, either issue to the Owner a Certificate for Payment, with a copy to the Contractor, for such amount as the A/E determines due, or notify the Contractor and Owner in writing of the A/E's reasons for withholding certification in whole or in part as provided in Paragraph 8.4.1. If the A/E fails to act within said ten (10) day period, the Contractor may file the Application for Payment directly with the Owner Representative and the Owner will thereafter have twenty (20) days from the date of the Owner's receipt to resolve the

amount to be paid and to pay the undisputed amount. The accuracy of the Contractor's Applications for Payment shall be Contractor's responsibility, not A/E's.

8.3.2 A/E'S REPRESENTATIONS. The A/E's issuance of a Certificate for Payment shall constitute a representation to the Owner that to the best of the A/E's knowledge, information and belief, based upon the A/E's observations at the site, the data comprising the Application for Payment, and what is reasonably inferable from the observations and data, that the Work has progressed to the point indicated in the Application and that the quality of the work is in accordance with the Contract Documents. The foregoing representations are subject to minor deviations from the Contract Documents correctable prior to completion and to specific qualifications expressed by the A/E. The issuance of a Certificate for Payment will further constitute a representation that the Contractor is entitled to payment in the amount certified. However, the issuance of a Certificate for Payment shall not be a representation that the A/E has (a) made exhaustive or continuous onsite inspections to check the quality or quantity of the Work, (b) reviewed construction means, methods, techniques, sequences or procedures, (c) reviewed copies of requisitions received from Subcontractors and material suppliers and other data requested by the Owner to substantiate the Contractor's right to payment, (d) ascertained how or for what purpose the Contractor used money previously paid on account of Contract Sum, or (e) any duty to make such inquiries.

8.4 DECISIONS TO WITHHOLD CERTIFICATION.

8.4.1 WHEN WITHHELD. The A/E may decide not to certify payment and may withhold a Certificate for Payment in whole or in part, to the extent reasonably necessary to protect the Owner, if in the A/E's judgment the representations to the Owner required in Paragraph 8.3.2 above cannot be made. If the A/E is unable to certify payment in the amount of the Application, the A/E shall notify the

Contractor and Owner as provided in Paragraph 8.3.1 above. If the Contractor and A/E cannot agree on a revised amount, the A/E shall promptly issue a Certificate for Payment for the amount to which the A/E makes such representations to the Owner. The A/E may also decide not to certify payment or, because of subsequently discovered evidence or observations, may nullify the whole or part of a Certificate for Payment previously issued, to such extent as may be necessary in the A/E's opinion to protect the Owner from loss because of:

- (1) Defective Work not remedied:
- (2) Third party claims filed or reasonable evidence indicating probable filing of such claims;
- (3) Failure of the Contractor to make payments properly to Subcontractors or for labor, materials or equipment;
- (4) Reasonable evidence that the Work cannot be completed for the unpaid balance of the Contract Sum;
- (5) Damage to the Owner or another contractor;
- (6) Reasonable evidence that the Work will not be completed within the Contract Time, and that the unpaid balance would not be adequate to cover actual or liquidated damages for the anticipated delay; or
- (7) Failure to carry out the Work in accordance with the Contract Documents.
- **8.4.2 CERTIFICATION ISSUED WHEN REASONS FOR WITHHOLDING REMOVED.** When the reasons stated in Paragraph 8.4.1 for withholding certification are removed, certification will be made for such related amounts.
- 8.4.3 CONTINUE WORK EVEN IF CONTRACTOR DISPUTES A/E'S

DETERMINATION. If the Contractor disputes any determination by the A/E or the result of the claims resolution process with regard to any Certification of Payment, the Contractor nevertheless shall expeditiously continue to prosecute the Work.

8.4.4 OWNER NOT IN BREACH. The Owner shall not be deemed to be in breach of this Contract by reason of the withholding of any payment pursuant to any provision of the Contract Documents provided the Owner's action or such withholding is consistent with the results of the dispute resolution process.

8.5 PROGRESS PAYMENTS.

8.5.1 IN GENERAL, INTEREST OR LATE PAYMENTS.

- (1) Except as provided in Paragraph 8.3.1, the Owner shall pay any undisputed amount within thirty (30) days of the date that the application for payment was submitted to the A/E. In no event shall Owner be required to pay any disputed amount.
- (2) Except as otherwise provided by law, if any payment is late based upon the provisions of the Contract Documents, the Contractor shall be paid interest in an amount equal to the published Wall Street Journal prime rate plus 2%. The published Wall Street Journal Prime Rate shall be determined using such rate that is published closest to the 1st of the month for each month of the late period. The amount of payment of interest shall be apportioned using such rate(s) for the late period.
- **8.5.2 CONTRACTOR AND SUBCONTRACTOR RESPONSIBILITY.** The Contractor shall promptly pay each Subcontractor, upon receipt of payment from the Owner, out of the amount paid to the Contractor on account of such Subcontractor's portion of the Work, the amount to which said Subcontractor is entitled. The Contractor shall, by appropriate agreement with each Subcontractor, require each

Subcontractor to make payment to its Subcontractors in a similar manner.

- **8.5.3 INFORMATION FURNISHED BY A/E OR OWNER TO SUBCONTRACTOR.** The A/E or Owner shall, on request, furnish to the Subcontractor, if practicable, information regarding percentages of completion or amounts applied for by the Contractor and action taken thereon by the A/E and Owner on account of portions of the Work done by such Subcontractor.
- **8.5.4 OWNER AND A/E NOT LIABLE.** Neither the Owner or A/E shall have an obligation to pay, monitor or enforce the payment of money to a Subcontractor, except to the extent as may otherwise be required by law.
- **8.5.5** CERTIFICATE, PAYMENT OR USE NOT ACCEPTANCE OF IMPROPER WORK. A Certificate for Payment, a progress payment, or partial or entire use or occupancy of the Project by the Owner shall not constitute acceptance of Work that is not in accordance with the Contract Documents.

8.6 PAYMENT UPON SUBSTANTIAL

COMPLETION. Upon Substantial Completion of the Work or designated portion thereof and upon application by the Contractor and certification by the A/E, the Owner shall make payment, reflecting adjustment in retainage, if any, for such Work or portion thereof as provided in the Contract Documents. To the extent allowed by law, the Owner may retain up to 200% of the fair market value of the work that has not been completed in accordance with the Contract Documents.

8.7 PARTIAL OCCUPANCY OR USE.

8.7.1 IN GENERAL. The Owner may occupy or use any completed or partially completed portion of the Work at any stage when such portion is designated by separate agreement with the Contractor, and authorized by public authorities having jurisdiction over the Work. Such partial occupancy or use may commence

whether or not the portion is Substantially Complete, provided the Owner and Contractor have accepted in writing the responsibilities assigned to each of them for payments, retainage if any, security, maintenance, heat, utilities, damage to the Work and insurance, and have agreed in writing concerning the period for correction of the Work and commencement of the warranties required by the Contract Documents. When the Contractor considers a portion to be substantially complete, the Contractor shall prepare and submit a list to the A/E as previously provided for herein. Consent of the Contractor to partial occupancy or use shall not be unreasonably withheld. Contractor shall have continuing responsibility to protect the site and the Work during such partial occupancy and shall be responsible for damage except to the extent caused solely by the Owner during such partial occupancy or use.

The stage of progress of the Work shall be determined by written agreement between the Owner and Contractor.

- **8.7.2 INSPECTION**. Immediately prior to such partial occupancy or use, the Owner, Contractor and A/E shall jointly inspect the area to be occupied or portion of the Work to be used in order to determine and record the condition of the Work.
- **8.7.3 NOT CONSTITUTE ACCEPTANCE.** Except to the extent it is agreed upon in writing by the Owner, partial occupancy or use of a portion or portion of the Work shall not constitute acceptance of Work not complying with the requirement of the Contract Documents.

8.8 FINAL PAYMENT.

8.8.1 CERTIFICATE FOR PAYMENT. The A/E's Final Certificate for Payment shall constitute a further representation that the conditions listed in Paragraph 8.8.2 as precedent to the Contractor's being entitled to final payment have been fulfilled.

- **8.8.2 CONDITIONS FOR FINAL PAYMENT.** Neither final payment nor any remaining retained percentage shall become due until the Contractor submits to the A/E the following to the extent required by the Owner Representative:
- (1) An affidavit that payrolls, bills for material and equipment, and other indebtedness connected with the Work for which the Owner's property might be responsible or encumbered (less amounts withheld by Owner) have been paid or otherwise satisfied;
- (2) A current or additional certificate evidencing that insurance required by the Contract Documents to remain in force after final payment is currently in effect and will not be canceled or allowed to expire until at least 30 days prior written notice, by certified mail, return receipt requested, has been given to the Owner;
- (3) A written statement that the Contractor knows of no substantial reason that the insurance will not be renewable to cover the period required by the Contract Documents;
- (4) If requested by surety in a timely manner or by Owner, consent of surety, to final payment;
- (5) Receipt of Record Drawings, Specifications, Addenda, Change Orders and other Modifications maintained at the site; the warranties, instructions, operation and maintenance manuals, and training videos required to be furnished by the Contract Documents:
- **(6)** Other data establishing payment or satisfaction of obligations, such as receipts, releases and waivers of liens, claims, security interests or encumbrances arising out of the Contract, to the extent and in such form as may be designated by the Owner. If a Subcontractor refuses to furnish a release or waiver required by the Owner, the Owner may require consent of Surety to the final payment. If such liens, claims,

- security interests or encumbrances remain unsatisfied after payments are made, the Contractor shall refund to the Owner all money that the Owner may be compelled to pay in discharging such lien, including all costs and reasonable attorneys' fees; and
- (7) A written statement demonstrating how the Contractor will distribute interest earned on retention to Subcontractors as required by Section 13.8.5, U.C.A.
- **8.8.3 WAIVER OF CLAIMS: FINAL PAYMENT.** The making of final payment shall constitute a waiver of Claims by the Owner except those arising from:
- (1) Liens, Claims, security interests or encumbrances arising out of the Contract and unsettled:
- (2) Failure of the Work to comply with the requirements of the Contract Documents;
- (3) Terms of warranties required by the Contract Documents; or
- (4) The one-year guaranty period and any corrected Work.
- **DELAYS NOT CONTRACTOR'S FAULT. If,** 8.8.4 after Substantial Completion of the Work, final completion thereof is materially delayed through no fault of the Contractor or by issuance of Change Orders affecting final completion, the Owner shall, upon application by the Contractor and certification by the A/E, and without terminating the Contract, make payment of the balance due for that portion of the Work fully completed and accepted. Such payment shall be made under terms and conditions governing final payment, except that it shall not constitute a waiver of claims. Unless otherwise stated by the Owner in writing, the making of final payment shall constitute a waiver of claims by the Owner as provided in Paragraph 8.8.3 for that portion of that Work fully completed and accepted by the Owner.

8.8.5 WAIVER BY ACCEPTING FINAL PAYMENT. Acceptance of final payment by the Contractor or a Subcontractor shall constitute a waiver of Claims by that payee except those Claims previously made in writing and identified by that payee as unsettled at the time of final Application for Payment. Such waivers shall be in addition to the waiver described in Paragraph 8.8.3.

ARTICLE 9. TESTS AND INSPECTIONS, SUBSTANTIAL AND FINAL COMPLETION, UNCOVERING, CORRECTION OF WORK AND GUARANTY PERIOD.

9.1 TESTS AND INSPECTIONS.

IN GENERAL. Tests, inspections and 9.1.1 approvals of portions of the Work required by the Contract Documents or by laws, ordinances, rules, regulations, resolutions or orders of public authorities having jurisdiction shall be made at appropriate time. Unless otherwise specifically set forth in the Contract Documents or agreed to by the Owner in writing, the Owner shall contract for such tests, inspections and approvals with an independent entity, or with the appropriate public authority, and the Owner shall bear all related costs of tests, inspections and approvals except as provided below. If any of the Work is required to be inspected or approved by the terms of the Contract Documents or by any public authority, the Contractor shall, at least two working days prior to the time of the desired inspection, and following the procedures established by the Owner, request such inspection or approval to be performed. The Contractor shall give the A/E timely notice of when and where tests and inspections are to be made so that the A/E may observe such procedures.

9.1.2 FAILURE OF AN INSPECTOR TO APPEAR. Work shall not proceed without any required inspection and the associated authorization by

Owner to proceed unless the following procedures and requirements have been met:

- (1) The inspection or approval was requested in a timely manner as provided in Paragraph 9.1.1;
- (2) The Contractor received written confirmation from the inspection entity that the inspection was scheduled;
- (3) The Contractor has contacted or attempted to contact the inspector to confirm that the inspector is unable to perform the inspection as scheduled;
- (4) If the inspector has confirmed that it is unable to perform the inspection as scheduled or if the Contractor is unable to contact the inspector, the contractor shall attempt to contact the City Building Official or Owner Representative for instruction; and
- **(5)** The Contractor has documented the condition of the work prior to being covered through photos or other means.
- **9.1.3 NONCONFORMING WORK**. If such procedures for testing, inspection or approval under Paragraph 9.1.1 reveal failure of portions of the Work to comply with the requirements established by the Contract Documents, the Contractor shall bear all costs made necessary by such failure including those of repeated procedures and compensation for the Owner's expenses, including the cost of retesting for verification of compliance if necessary, until the Owner accepts the Work in question as complying with the requirements of the Contract Documents.
- **9.1.4 CERTIFICATES**. Required certificates of testing, inspection or approval shall, unless otherwise required by the Contract Documents, be secured by the Contractor and promptly delivered to the A/E.
- **9.1.5 A/E OBSERVING**. If the A/E is to observe tests, inspections or approvals required by the

Contract Documents, the A/E shall do so with reasonable promptness and, where practicable, at the normal place of testing.

9.1.6 PROMPTNESS. Tests, inspections and arrangements for approvals conducted pursuant to the Contract Documents shall be made promptly to avoid unreasonable delay in the Work.

9.2 INSPECTIONS: SUBSTANTIAL AND FINAL.

9.2.1 SUBSTANTIAL COMPLETION

INSPECTION. Prior to requesting a substantial completion inspection, the Contractor prepare a comprehensive initial punch list, including unresolved items from inspections, for review by the Owner and A/E to determine if the Project is ready for a substantial completion inspection. If the Owner determines that the initial punch list indicates that the Project is not substantially complete, the initial punch list will be returned to the Contractor with written comments. If the Owner determines that the initial punch list indicates that the Project may be substantially complete, the A/E shall promptly organize and perform a Substantial Completion inspection in the presence of the Owner and all appropriate authorities.

- (1) If the A/E reasonably determines that the initial punch list prepared by the Contractor substantially understates the amount of the Work remaining to be completed and the Project is not substantially complete, the A/E shall report this promptly to the Owner, and upon concurrence of the Owner, the Contractor will be assessed the costs of the inspection and punch list preparation incurred by the A/E and the Owner.
- (2) When the Work or designated portion thereof is Substantially Complete, the A/E shall prepare a Certificate of Substantial Completion which shall establish the date of Substantial Completion; shall establish responsibilities of the Owner and Contractor for security, maintenance, heat, utilities, damage to the work

and insurance; and shall fix the time within which the Contractor shall finish all items on the punch list accompanying the Certificate. The Certificate of Substantial Completion shall require approval by the Owner Representative. If there is a punch list, the Contractor shall proceed promptly to complete and correct items on the list. Failure to include an item on the punch list does not alter the responsibility of the Contractor to complete all Work in accordance with the Contract Documents.

- (3) Warranties required by the Contract Documents shall commence on the date of Substantial Completion of the Work or designated portion thereof except to the extent as provided otherwise in the Contract Documents or if such warranty is related to an item where the work is not complete. Such warranty documents shall state the length of the warranty, which must comply with the Contract Documents.
- (4) The Certificate of Substantial Completion shall be submitted by the A/E to the Owner and Contractor for their written acceptance of responsibilities assigned to them in such Certificate.
- the Owner (5) Except to the extent Representative otherwise approves advance and in writing, the Contractor shall submit the following documents in order to achieve Substantial Completion: written warranties, guarantees, operation and maintenance manuals, and all complete as-built drawings. The Contractor must also provide or obtain any required approvals for occupancy. The Contractor is responsible for the guaranty of all Work, whether performed by it or by its Subcontractors at any tier.
- **9.2.2 FINAL COMPLETION INSPECTION.** Prior to requesting a final inspection, the Contractor shall verify all punch list items are corrected/completed. Once all punch list items are corrected/completed the Contractor shall notify the Owner and request a final inspection.

The Owner shall notify the A/E and perform a final inspection. Two final inspections may be allowed due to required weather changes required to complete some items. When all punch list items are completed a final pay request will be provided by the Contractor, authorized by the A/E and processed by the Owner.

9.3 UNCOVERING OF WORK.

9.3.1 UNCOVER UNINSPECTED WORK. Except as provided in Paragraph 9.3.3, if a portion of the Work is covered prior to an Inspector's approval to proceed, it must, be uncovered for the Inspector's inspection and be replaced at the Contractor's expense without change in the Contract Time.

9.3.2 OBSERVATION PRIOR TO COVERING. Except as provided in Paragraph 9.3.3, if the Owner or the A/E has requested in writing to observe conditions prior to any Work being covered or if such observation is specified in the Contract Documents, and the Work is covered without such observation, the Contractor shall be required to uncover and appropriately replace the Work at the Contractor's expense without change in the Contract Time. If the Contractor requests an inspection and the Owner or A/E, including any inspector of each, does not appear, the Contractor shall immediately notify the Owner of such lack of appearance, but shall not cover the Work without such inspection.

9.3.3 WHEN AN INSPECTOR FAILS TO APPEAR OR A/E OR Owner DID NOT MAKE PRIOR REQUEST. If Work is performed by the Contractor without an inspection as provided in Paragraph 9.1.2 or if a portion of the Work has been covered which the A/E or Owner has not specifically requested to observe prior to its being covered or such observation is not specified by the Contract Documents, the A/E or Owner may request to see such Work and it shall be uncovered by the Contractor. If such Work is in accordance with the Contract Documents,

costs of uncovering and replacement, shall, by appropriate Change Order, be charged to the Owner. If such Work is not in accordance with the Contract Documents, the Contractor shall pay such costs unless the condition was caused by the Owner or a separate contractor in which event the Owner shall be responsible for payment of such costs.

9.4 CORRECTION OF WORK AND GUARANTY PERIOD.

9.4.1 CONTRACTOR CORRECT THE WORK. The Contractor shall correct Work rejected by the A/E, Inspector or Owner, or failing to conform to the requirements of the Contract Documents, whether observed before or after Substantial Completion and whether or not fabricated, installed or completed. The Contractor shall bear the costs of correcting such rejected Work, including additional testing and inspections and compensation for the A/E's and Inspector's services and expenses made necessary thereby.

9.4.2 GUARANTY AND CORRECTION AFTER SUBSTANTIAL COMPLETION. If within one year after the date of Substantial Completion of the Work or designated portion thereof, or after the date for commencement of warranties established under Paragraph 9.2.1 or by terms of an applicable special warranty or guaranty required by the Contract Documents, any of the Work is found to be not in accordance with the requirements of the Contract Documents, including failure to perform for its intended purpose, the Contractor shall correct it promptly after receipt of written notice from the Owner to do so unless the Owner has previously given

the Contractor a written acceptance of such condition. The period of one year shall be extended with respect to portions of the Work first performed after Substantial Completion by the period of time between Substantial Completion and the actual performance of the Work. This obligation of the Contractor under this Paragraph 9.4.2 shall be operative notwithstanding the acceptance of the Work

under the Contract, the final certificate of payment, partial or total occupancy and/or termination of the Contract. The Owner shall give notice of observed defects with reasonable promptness; however, failure to give such notice shall not relieve the Contractor of its obligation to correct the Work at the cost that the Contractor would have incurred if the Owner did so report with reasonable promptness. All corrected Work shall be subject to a one-year guaranty period the same in all respects as the original Work, except that such guaranty period shall commence from the time of Substantial Completion of the corrected Work. This guaranty period does not affect the Owner's right to available remedies against pursue any Contractor.

9.4.3 REMOVAL OF WORK.

- (1) The Contractor shall promptly remove from the premises all Work that the Owner and/or the A/E determines as being in nonconformance with the Contract Documents, whether incorporated or not.
- (2) The Contractor shall promptly replace and re-execute the Work in accordance with the Contract Documents and without expense to the Owner.
- (3) The Contractor shall bear the expense of correcting destroyed or damaged construction, whether completed or partially completed, of the Owner or of other contractors destroyed or damaged by such removal or replacement.
- (4) If the Contractor does not remove such rejected Work within a reasonable time, fixed by written notice, the Owner may have the materials removed and stored at the expense of the Contractor.
- (5) If the Contractor does not correct the nonconforming Work within a reasonable time, fixed by written notice, the Owner may correct it in accordance with Paragraph 12.2.2 of these General Conditions.

9.4.4 NOT LIMIT OTHER OBLIGATIONS. Nothing contained in this Article 9.4 shall be construed to establish a period of limitation with respect to other obligations which the Contractor may have under the Contract Documents. Establishment of the time period of one year as described in Paragraph 9.4.2 relates only to the specific obligation of the Contractor to correct the Work, and has no relationship to the time within which the obligation to comply with the Contract Documents may be sought to be enforced, nor to the time within which proceedings may be commenced to establish the Contractor's liability with respect to the Contractor's obligations other than specifically to correct the Work.

9.5 ADDITIONAL WARRANTIES.

- **9.5.1 IN GENERAL**. In addition to any other provisions of this Article 9, the following warranties shall apply:
- (1) The Contractor warrants to the Owner that materials and equipment furnished under the Contract will be of good quality and new, except to the extent otherwise required or expressly permitted by the Contract Documents.
- (2) The Contractor also warrants to the Owner that the Work will be free from defects not inherent in the quality required or permitted and that the Work will conform with the requirements of the Contract Documents. Work not conforming to said requirements, including substitutions not properly approved and authorized, may be considered defective at the Owner's option.
- **9.5.2 EXCLUSION**. Unless due to the negligent or intentional act or omission of the Contractor or those under the Contractor's control, or as otherwise stated in the Contract Documents, the Contractor's guaranty excludes remedy for damage or defect caused by abuse, modifications not executed by the Contractor, improper or insufficient maintenance, improper

operation, or normal wear and tear under normal usage.

9.5.3 FURNISH EVIDENCE ON REQUEST. If requested by the A/E or Owner, the Contractor shall furnish satisfactory evidence as to the type and quality of materials and equipment.

9.6 ACCEPTANCE OF NONCONFORMING WORK. If the Owner prefers to accept Work which is not in accordance with the requirements of the Contract Documents, the Owner may do so in writing instead of requiring its removal and correction, in which case the Contract Sum shall be reduced as appropriate and equitable. Such adjustment shall be effected whether or not final payment has been made.

ARTICLE 10. INSURANCE AND BONDS.

10.1 LIABILITY INSURANCE.

10.1.1 IN GENERAL. To protect against liability, loss and/or expense arising from damage to property or injury or death of any person or persons incurred in any way out of, in connection with or resulting from the Work provided hereunder, Contractor shall obtain and maintain in force during the entire period of this Contract without interruption, at its own expense, the following insurance from insurance companies authorized to do business in the State of Utah in a form and content satisfactory to the Owner and rated "A-"or better with a financial size category of (a) Class X or larger where the Contract Sum is \$1,000,000 or greater or (b) Class VII or larger where the Contract Sum is under \$1,000,000. Said rating and financial size category shall be as published by A.M. Best Company at the time the Contract is executed.

(1) Workers' Compensation Insurance and Employers' Liability Insurance. Worker's Compensation Insurance shall cover full liability under the Worker's Compensation Laws of the jurisdiction in which the Project is located at the

statutory limits required by said jurisdiction's laws. Employer's Liability Insurance shall provide the following limits of liability: \$100,000 for each accident; \$500,000 for Disease-Policy Limit; and \$100,000 for Disease-Each Employee. The Contractor shall require all Subcontractors to take and maintain similar policies of Workers' Compensation Insurance.

(2) Commercial General Liability Insurance.

a. Commercial General Liability Insurance, on an "occurrence basis," including insurance for independent contractors. operations, subcontractors at any tier, products/completed operations and contractual liability specifically designating the Indemnity provisions of these General Conditions as an insured contract on the Certificate of Insurance. Such Commercial General Liability Insurance must be endorsed with a Broad Form Property Damage **Endorsement (including Completed Operations)** and afford coverage for explosion, collapse and underground hazards. Such Commercial General Liability Insurance shall be in limits not less than the following:

\$1,000,000 General Aggregate, plus:

- i. if the Construction Value is \$25,000,000 or more, an additional \$5,000,000 umbrella policy (which covers aggregate and per occurrence) is required; or
- ii. if the Construction Value is \$10,000,000 or more but less than \$25,000,000, an additional \$2,000,000 umbrella policy (which covers aggregate and per occurrence) is required.
- \$1,000,000 Products-Completed Operations Aggregate \$1,000,000 Personal and Advertising Injury \$1,000,000 Each Occurrence
- b. For purposes of this subparagraph 2(a), Construction Value means:

- i. the Contract Sum if the work is being performed under a Standard Construction Contractor's Agreement;
- ii. the Fixed Limit of Construction Costs if the work is being performed under a Construction Manager/General Contractor Agreement; or
- iii. the Guaranteed Fixed contract Amount if the work is to be performed under a Design/Build Agreement.
- (3) Automobile liability insurance for claims arising from the ownership, maintenance, or use of a motor vehicle. The insurance shall cover all owned, non-owned, and hired automobiles used in connection with the Work, with the following minimum limits of liability:
- \$1,000,000 Combined Single Limit Bodily Injury and Property Damage Per Occurrence
- (4) Aircraft Use. Contractor using its own aircraft, or employing aircraft in connection with the Work performed under this Agreement shall maintain Aircraft Liability Insurance with a combined single limit of not less than \$1,000,000 per occurrence. Said certificate shall state that the policy required by this paragraph has been endorsed to name the Owner as Additional Insureds.
- (5) Unless otherwise provided by the procurement documents, the insurance requirements in 10.1.1(1) through (4) above do not apply to subcontractors or suppliers at any tier under the Contractor and any insurance requirements of subcontractors and suppliers at any tier is a matter between the General Contractor and such subcontractor or supplier.
- **10.1.2 CONFIGURATIONS.** Any policy required by this Article may be arranged under a single policy for the full limit required, or by a combination of underlying policies with the balance provided by an Excess or Umbrella Liability Policy.

10.1.3 CONTRACTOR LIABILITY. Irrespective of the requirements as to insurance to be carried by Contractor as provided herein; insolvency, bankruptcy or failure of any insurance company to pay all claims accruing, shall not be held to relieve Contractor of any obligations hereunder.

10.1.4 CERTIFICATE, NOTICE REQUIREMENTS, ADDITIONAL INSURED. Before the Contract Agreement is executed, certificates evidencing coverages as specified above are in effect, shall be furnished to the Owner. Such insurance certificates shall contain provisions that no cancellation, material change therein or nonrenewal shall become effective except upon thirty (30) days prior written notice to the Owner as evidenced by return receipt, certified mail sent to Owner. The Contractor shall notify the Owner within thirty (30) days of any claims(s) against the Contractor, and if such claim(s) exceed 20% of the applicable required insured limits, the Owner may require the Contractor to re-instate the policy to provide full protection at the original limits. For any risk not covered by the Worker's Compensation Policy, the Owner shall be named as additional insured parties. All insurance policies provided shall be primary and non-contributing with, and not in excess of, any other insurance or self-insurance available to the Owner.

10.1.5 DEDUCTIBLE LIABILITY. Any and all deductibles in the above described policies shall be assumed by, for the account of, and at sole risk of Contractor.

10.1.6 ADDITIONAL REQUIREMENTS:

- (1) Any type of insurance or any increase of limits of liability not described in this Agreement which the Contractor requires for its own protection or on account of any statute, rule or regulation, shall be its own responsibility and at its own expense.
- (2) The carrying of any insurance required by this Agreement shall in no way be interpreted as relieving the Contractor or Subcontractors of any

other responsibility or liability under this Agreement or any applicable law, statute, rule, regulation or order.

(3) Contractor shall not violate or knowingly permit to be violated any of the provisions of the policies on insurance required under these General Conditions.

10.2 "BUILDER'S RISK" PROPERTY INSURANCE.

10.2.1 IN GENERAL. The Owner shall provide "Builder's Risk" property insurance to protect the Owner, as well as all Contractors and Subcontractors, and include them as insureds, with respect to Work performed hereunder at the Owner's own cost and expense, according to the policies and forms currently in force with insurance carriers selected by the Owner's Risk Manager. The Owner's Risk Manager shall furnish, upon request, all parties in interest with copies of said policies authenticated by authorized agents of the insurers.

10.2.2 INSPECTIONS, RECOMMENDATIONS. Owner shall have the right to inspect the Work. The Contractor shall comply with reasonable risk control recommendations made by insurers or the Owner. Such inspections or recommendations do not relieve the Contractor of any of its responsibilities under the Contract

Documents.

10.2.3 DEDUCTIBLE. The above described "Builders Risk" policies shall be subject to a total deductible of \$5,000 per loss occurrence, which shall be assumed by all Contractors or Subcontractors, in proportion to their share of the total amount of an insured loss occurrence.

10.2.4 ADJUSTED WITH AND PAYABLE TO RISK MANAGER AS TRUSTEE. Any insured property loss is to be adjusted with the Owner, and made payable to the Owner as trustee for the Contractor and Subcontractors, as their interests may appear, subject to the requirements of any applicable loss payable clause.

10.2.5 WAIVER. Contractor, including all Subcontractors, and Owner hereby waive all rights against each other for damages caused by perils insured against under the "Builder's Risk" insurance provided by Owner, except such rights as Contractor may have to the proceeds of such insurance held by the Owner as trustee. The Owner and the Contractor each shall require similar waivers from their contractors, subcontractors, subconsultants and agents, at any tier.

10.2.6 SPECIAL HAZARDS. Owner shall bear the risk of loss, delay and/or damage due to earthquake and/or flood and may either insure or self-insure that risk. If the Contractor requests in writing that insurance for other special hazards be included in the "Builder's Risk" policy, the Owner's Risk Manager shall, if possible, include such insurance in the policy and the cost thereof shall be charged to the Contractor by Change Order.

10.3 PERFORMANCE BOND AND PAYMENT **BOND**. The Contractor shall submit and maintain in full force and effect as required by law and the Contract Documents, at its own expense, on forms provided by the Owner, and include as part of the quoted total all costs involved in securing and furnishing, the bonds listed below, based on the completed cost of the Contract and effective upon execution of the Contract. Said bonds shall be from surety companies which are authorized to do business in the State of Utah, listed in the U. S. Department of Treasury Circular 570, Companies Holding Certificates of Authority as Acceptable Sureties on Federal Bonds and as Acceptable Reinsuring Companies, and acting within the limitation listed therein.

10.3.1 A full 100 percent performance bond covering the faithful execution of the Contract in accordance with the Contract Documents; and 10.3.2 A full 100 percent payment bond covering payment of all obligations arising under the Contract Documents, for the protection of

each person supplying labor, service, equipment, or material for the performance of the Work.

10.3.3 Any required insurance required under the U.S. Terrorism Risk Insurance Act of 2002, any similar applicable law, or as such Act may be amended.

ARTICLE 11. MISCELLANEOUS PROVISIONS.

- 11.1 A/E'S RESPONSIBILITIES. These General Conditions are not intended to provide an exhaustive or complete list of the A/E's responsibilities. A separate agreement between the Owner and A/E incorporates these General Conditions by reference and includes additional Design responsibilities.
- 11.2 **SUCCESSORS AND ASSIGNS.** The Owner and Contractor respectively bind themselves, their partners, successors, assigns and legal representatives to the other party hereto and to partners, successors, assigns and legal representatives of such other party in respect to covenants, agreements and obligations contained in the Contract Documents. The Contractor shall not assign the Contract without the prior written consent of the Owner, nor shall the Contractor assign any amount due or to become due as well as any rights under the Contract, without prior written consent of the Owner.

11.3 WRITTEN NOTICE.

11.3.1 PERSONAL DELIVERY AND REGISTERED OR CERTIFIED MAIL. Written notice shall be deemed to have been duly served if delivered in person to the individual or a member of the firm or entity or to an officer of the corporation for which it was intended, or if delivered at or sent by registered or certified mail, return receipt requested, to the last business address known to the party giving notice.

11.3.2 FAX. **Notwithstanding** any other provision of these General Conditions, written notice shall also be deemed to have been duly served by verified use of a FAX system by using the known and operative calling number. Service by use of the FAX system is encouraged when timely notice will benefit the Owner, A/E or Contractor. Notice shall be considered complete and verified upon the sending and confirmation of delivery using the FAX system, if on the same day notice is also sent by registered or certified mail, return receipt requested, to the last business address known to the party giving notice, confirming the FAX delivery.

11.4 RIGHTS AND REMEDIES.

- **11.4.1 NOT LIMIT.** Duties and obligations imposed by the Contract Documents and rights and remedies available thereunder shall be in addition to and not a limitation of duties, obligations, rights and remedies otherwise imposed or available by law.
- 11.4.2 NOT WAIVER. Except as expressly provided elsewhere in the Contract Documents, no action or failure to act by the Owner, A/E or Contractor shall constitute a waiver of a right or duty afforded them under the Contract Documents, nor shall such action or failure to act constitute approval or acquiescence in a breach thereunder, except as any of the above may be specifically agreed to in writing. In no case shall the Contractor or any Subcontractors be entitled to rely upon any waiver of any of these General Conditions unless agreed to in writing by the Owner.

11.5 COMMENCEMENT OF STATUTORY LIMITATION PERIOD.

11.5.1 BEFORE SUBSTANTIAL COMPLETION. Except as provided in 11.5.4 below, as to acts or failures to act occurring prior to the relevant date of Substantial Completion, any applicable statute of limitations shall commence to run and any alleged cause of action shall be deemed to

have accrued in any and all events not later than such date of Substantial Completion.

11.5.2 BETWEEN SUBSTANTIAL COMPLETION AND FINAL CERTIFICATION FOR PAYMENT.

Except as provided in Paragraph 11.5.4 below, as to acts or failures to act occurring subsequent to the relevant date of Substantial Completion and prior to issuance of the final Certification for Payment, any applicable statute of limitations shall commence to run and any alleged cause of action shall be deemed to have accrued in any and all events not later than the date of issuance of the final Certification for Payment.

- 11.5.3 AFTER FINAL CERTIFICATION FOR PAYMENT. Except as provided in Paragraph 11.5.4 below, as to acts or failures to act occurring after the relevant date of issuance of the final Certificate for Payment, any applicable statute of limitations shall commence to run and any alleged cause of action shall be deemed to have accrued in any and all events not later than the date of any act or failure to act by the Contractor pursuant to any guaranty provided under Article 9 the date of any correction of the Work or failure to correct the Work by the Contractor under Paragraph 9.4.2, or the date of actual commission of any other act or failure to perform any duty or obligation by the Contractor or Owner, whichever occurs last.
- 11.5.4 EXCEPTION. Notwithstanding any other provision of this Article 11.5 to the contrary, no applicable statute of limitations shall be deemed to have commenced with respect to any portion of the Work which is not in accordance with the requirements of the Contract Documents, which would not be visible or apparent upon conducting a reasonable investigation, and which is not discovered by the Owner until after the date which, but for this Paragraph 11.5.4, would be the date of commencement of the applicable statute of limitations; the applicable statute of limitations instead shall be deemed to have commenced on the date of such discovery by the Owner.

- NOT DISCRIMINATE, NO SEXUAL 11.6 **HARASSMENT**. Pursuant to the laws of the State of Utah, the Contractor, Subcontractors, or anyone for whose act any of them may be liable, will take affirmative action to not discriminate against any employee or applicant for employment because of race, creed, color, sex, religion, ancestry or national origin. To the extent applicable, said persons will comply with all provisions of Executive Order No. 11246 dated September 24, 1965 and rules, regulations, orders, instructions, designations and other directives promulgated pursuant thereto. Contractor, Subcontractors, or anyone for whose act any of them may be liable, shall not act in any manner as would violate the laws, regulations and policies of the United States or the State of Utah prohibiting sexual harassment.
- 11.7 APPLICABLE LAWS. The applicable laws and regulations of the State of Utah, as well as any applicable local laws and regulations not superseded or exempted by State law, shall govern the execution of the Work embodied in the Contract Documents as well as the interpretation of the Contract Documents.
- **11.8 INTERPRETATION**. In the interest of brevity, the Contract Documents frequently omit modifying words such as "all" and "any" and articles such as "the" and "an", but the fact that a modification or an article is absent from the statement and appears in another is not intended to affect the interpretation of either statement.
- **11.9 VENUE**. In case of any dispute, which may arise under the Contract Documents, the place of venue shall be in the County of Salt Lake, Utah, unless otherwise agreed to by all the parties in writing.
- **11.10 SEVERABILITY**. The invalidity of any part, paragraph, subparagraph, phase, provision or aspect of the Contract documents shall not impair or affect in any manner the validity, enforceability or effect of the remainder of the Contract Documents.

- 11.11 CONSTRUCTION OF WORDS. Unless otherwise stated in the Contract Documents, words, which have well-known technical or construction industry meanings, shall be construed as having such recognized meanings. Unless the context requires otherwise, all other technical words shall be construed in accordance with the meaning normally established by the particular, applicable profession or industry. All other words, unless the context requires otherwise, shall be construed with an ordinary, plain meaning.
- **11.12 NO THIRD PARTY RIGHTS.** These General Conditions create rights and duties only as between Owner and Contractor, and Owner and A/E. Nothing contained herein shall be deemed as creating third party beneficiary contract rights or other actionable rights or duties as between Contractor and A/E, or as between Owner, Contractor, or A/E on the one hand, and any other person or entity.

ARTICLE 12. TERMINATION OR SUSPENSION OF THE CONTRACT.

12.1 TERMINATION BY CONTRACTOR.

- **12.1.1 IN GENERAL**. If the Work is stopped for a period of sixty (60) days through no act or fault of the Contractor or a Subcontractor, or their agents or employees or any other persons performing portions of the Work under contract with any of the above, the Contractor, may terminate the Contract in accordance with 12.1.2 hereinbelow for any of the following reasons:
- (1) Because the Owner has persistently failed to fulfill fundamental Owner's obligations under the Contract Documents with respect to matters important to the progress of the Work;
- (2) Issuance of an order of a court or other public authority having jurisdiction which necessitates such termination, except that

- where the Contractor has standing, the Contractor must cooperate in efforts to stay and/or appeal such order;
- (3) An act of government, such as a declaration of national emergency, making material unavailable; or
- (4) Unavoidable casualties or other similar causes as listed in Paragraph 12.2.2(2) hereinbelow.
- **12.1.2 NOTICE**. If one of the reasons for termination in Paragraph 12.1.1 hereinabove exist, the Contractor may, upon ten (10) additional days' written notice to the Owner and A/E, and such condition giving cause for termination still not cured, terminate the Contract and recover from the Owner payment for Work executed and for proven loss with respect to materials, equipment, tools, and construction equipment and machinery, including reasonable overhead, profit and damages associated only with work completed prior to the notice of termination.

12.2 TERMINATION BY THE Owner FOR CAUSE.

- 12.2.1 IN GENERAL. The Owner Director or Designee may terminate the Contract if the Contractor fails to cure any of the following within a period of ten (10) days (or longer if the Owner so approves in writing) after receipt of notice from the Owner specifying the cause for termination:
- (1) The Contractor persistently or repeatedly refuses or fails to supply enough properly skilled workers or proper materials;
- (2) The Contractor fails to make payment to Subcontractors for materials or labor in accordance with the respective agreements between the Contractor and the Subcontractors;
- (3) The Contractor persistently disregards laws, ordinances, or rules, regulations, resolutions or

orders of a public authority having jurisdiction;

- (4) The Contractor fails to perform the Work within the time specified in the Contract Documents or any authorized extension thereof or the Contractor fails to make progress with the Work as to endanger such compliance;
- (5) The Contractor fails to perform the Work or is otherwise in breach of a material provision of the Contract Documents;
- **(6)** The Contractor fails to respond promptly to the financial responsibility inquiry under the Contractor's Agreement;
- (7) As permissible by law for a reason to terminate, the Contractor is adjudged bankrupt;
- **(8)** As permissible by law for a reason to terminate, the Contractor should make a general assignment for the benefit to creditors;
- **(9)** As permissible by law for a reason to terminate, the Contractor should have a receiver appointed on account of the Contractor's insolvency; or
- (10) The Contractor fails to follow the material safety requirements and precautions either as expressly provided in the Contract Documents or as consistent with the customary practices in the industry.

12.2.2 OWNER'S RIGHT TO CARRY OUT THE WORK.

(1) If the Contractor defaults or neglects to carry out the Work in accordance with the Contract Documents and fails within a ten (10) day period (or longer if approved by the Owner in writing) after receipt of written notice from the Owner to cure such default or neglect, the Owner may without prejudice to other remedies the Owner may have, correct such deficiencies, including taking over the Work and prosecuting the same to completion, by contract or

otherwise, and may take possession of, and utilize in completing the Work, such materials, appliances, and facilities as may be on the site of the Work as well as the site as necessary for its proper completion. In such case, the Owner shall offset from payments then or thereafter due the Contractor the cost of correcting such deficiencies, including compensation for the A/E, Owner's staff and legal counsel's additional services and expenses made necessary by such default, neglect or failure. If payments then or thereafter due the Contractor are not sufficient to cover such amounts, the Contractor shall pay the difference to the Owner. The Contractor shall continue performance of the Contract to the extent not terminated.

(2) Except with respect to defaults of Subcontractors, the Contractor shall not be liable for any excess costs if the failure to perform the Contract arises out of causes beyond the control and without the fault or negligence of the Contractor or anyone for whom the Contractor may be liable. Such causes may include, but are not limited to, acts of God or of the public enemy, acts of the local municipality, State of Utah or federal government in either their sovereign or contractual capacity, fires, floods, epidemics, quarantine restrictions, strikes, freight embargoes, and unusually severe weather; but in every case the failure to perform must be beyond the control and without the fault or negligence of the Contractor or anyone for whom the Contractor may be liable. If the failure to perform is caused by the default of a Subcontractor, and if such default arises out of causes beyond the control of both the Contractor and the Subcontractor, and without the fault or negligence of either of them or anyone for whom either may be liable, the Contractor shall not be liable for any excess costs for failure to perform unless the supplies or services to be furnished by the Subcontractor were obtainable from other sources in sufficient time to permit the Contractor to meet the required delivery or completion schedule.

- **12.2.3 ITEMS REQUIRED TO BE TRANSFERRED OR DELIVERED.** The Owner may require the Contractor to transfer title and deliver to the Owner, in the manner and to the extent directed by the Owner:
- (1) Any completed portion of the Work; and
- (2) Any partially completed portion of the Work and any parts, tools, dies, jigs, fixtures, drawings, information, and contract rights (hereinafter called "construction materials") as the Contractor has specifically produced or specifically acquired for the performance of such part of this Contract as has been terminated; and the Contractor shall, upon direction of the Owner, protect and preserve property in the possession of the Contractor in which the Owner has an interest.
- **12.2.4 PAYMENT.** When the Owner terminates the Contract for one or more of the reasons stated in Paragraph 12.2.1, the Owner may withhold payment and/or pursue all available remedies.
- **12.2.5 OWNER PROTECTION IF LIENABLE.** When the subject property is lienable, the Owner may withhold from amounts otherwise due the Contractor for such completed Work or construction materials such sum as the Owner determines to be necessary to protect the Owner against loss because of outstanding liens or claims for former lien holders.
- **12.2.6 CREDITS AND DEFICITS**. If the unpaid balance of the Contract Sum exceeds the full cost of finishing the Work, including compensation for the A/E's services and expenses made necessary thereby, such excess shall be paid to the Contractor. If such cost exceeds the unpaid balance, the Contractor shall pay the difference to the Owner this obligation for payment shall survive the termination of the Contract.
- **12.2.7 IF CONTRACTOR FOUND NOT IN DEFAULT OR EXCUSABLE**. If, after notice of termination of the Contract under the provisions

of this Article, it is determined for any reason that the Contractor was not in default under the provisions of this Article, or that the default was excusable under the provisions of this Article, the rights and obligations of the parties shall be the same as if the notice of termination had been issued pursuant to the termination for convenience provisions.

- **12.2.8 RIGHTS** AND REMEDIES NOT EXCLUSIVE. The rights and remedies of the Owner provided in this Article 12.2 shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.
- 12.3 SUSPENSION, DELAY OR INTERRUPTION OF WORK BY THE OWNER FOR CONVENIENCE.
- 12.3.1 BY OWNER IN WRITING. The Owner may in writing and without cause, order the Contractor to suspend, delay or interrupt the Work in whole or in part for such period of time as the Owner may determine to be appropriate for the convenience of the Owner.
- **12.3.2 TIME PERIOD FOR CLAIMS**. Any PRE by the Contractor for adjustment under this Article 12.3 must be asserted by the Contractor, in writing, within twenty-one (21) days from the date of termination of such suspension, delay or interruption; provided that the Owner may, in its sole discretion, receive and act upon any such PRE asserted at any time prior to final payment under this Contract.

12.3.3 ADJUSTMENTS.

Any adjustment in Contract Sum and Time shall be in accordance with Articles 3, 4, and 7.

12.4 TERMINATION FOR CONVENIENCE OF THE OWNER.

12.4.1 IN GENERAL.

The performance of Work under this Contract may be terminated by the Owner in accordance

with this Article 12.4 in whole, or from time to time, in part, whenever the Owner shall determine that such termination is in the best interest of the Owner or any person for whom the Owner is acting under this Contract. Any such termination shall be effected by delivery to the Contractor of a notice of termination specifying the extent to which performance of Work under the Contract is terminated, and the date upon which such termination becomes effective.

- **12.4.2 CONTRACTOR OBLIGATIONS.** After receipt of a notice of termination, and except as otherwise directed by the Owner in writing, the Contractor shall:
- (1) Stop work under the Contract on the date and to the extent specified in the notice of termination;
- (2) Place no further orders or subcontracts for materials, services or facilities, except as may be necessary for completion of such portion of the Work under the Contract as is not terminated;
- (3) Terminate all orders and subcontracts to the extent that they relate to performance of Work terminated by the notice of termination;
- (4) Assign to the Owner in the manner, at the times, and to the extent directed by the Owner, all of the right, title and interest of the Contractor under the orders and subcontracts so terminated, in which case the Owner shall have the right, in its discretion, to settle or pay any or all claims arising out of the termination of such orders and subcontracts;
- (5) Settle all outstanding liabilities and all claims arising out of such termination of orders and subcontracts, with the approval or ratification of the Owner, which approval or ratification shall be final for all the purposes of this Article 12.4;
- **(6)** Transfer title and deliver to the Owner in the manner, at the times, and to the extent, if any, directed by the Owner:

- (a) The fabricated or unfabricated parts, work in process, completed work, supplies, and other material produced as a part of, or acquired in connection with the performance of the Work terminated by the notice of termination; and
- (b) The completed or partially completed drawings, information, and other property which, if the Contract had been completed, would have been required to be furnished to the Owner;
- (7) Use best efforts to sell, in the manner, at the times, to the extent, and at the price or prices directed or authorized by the Owner, any property of the types referred to in Paragraph 12.4.2(6) above; provided, however, that the Contractor:
- (a) Shall not be required to extend credit to any purchaser; and
- (b) May acquire any such property under the conditions prescribed by and at a price or prices approved by the Owner; and provided further that the proceeds of any such transfer of or disposition shall be applied in reduction of any payments to be made by the Owner to the Contractor under this Contract or shall otherwise be credited to the price or cost of the Work covered by this Contract or paid in such other manner as the Owner may direct;
- (8) Complete performance of such part of the Work as shall not have been terminated by the notice of termination; and
- **(9)** Take such action as may be necessary, or as the Owner may direct, for the protection and preservation of the property related to this Contract which is in the possession of the Contractor in which the Owner has or may acquire an interest.

12.4.3 TERMINATION CLAIM.

After receipt of a notice of termination, the Contractor may submit to the Owner a PRE, in the form and with certification prescribed by the Owner. Such PRE shall be submitted promptly but in no event not later than sixty (60) days from the effective date of termination.

- **12.4.4 AGREED UPON PAYMENT.** Subject to the provisions of Paragraph 12.4.3 above, the Contractor and the Owner may agree upon the amount to be paid to the Contractor by reason of the total or partial termination of Work pursuant to this Article 12.4.
- **12.4.5 PAYMENT NOT AGREED UPON.** In the event of the failure of the Contractor and the Owner to agree, as provided in Paragraph 12.4.4, upon the whole amount to be paid to the Contractor by reason of the termination of Work pursuant to this Article 12.4, the Owner shall pay to the Contractor the amounts determined by the Owner as follows, but without duplication of any amounts agreed upon in accordance with Paragraph 12.4.4:
- (1) With respect to all Contract Work performed prior to effective date of the notice of termination, the total (without duplication of any items) of:
- (a) The cost of such Work including undisputed Claim amounts:
- (b) The cost of terminating, settling and paying claims arising out of the termination of Work under subcontracts or orders as provided in Paragraph 12.4.2(5) above, exclusive of the amounts paid or payable on account of supplies or materials delivered or services furnished by Subcontractors prior to the effective date of the notice of termination under this Contract, which amounts shall be included in the cost on account of which payment is made under Paragraph 12.4.5(1)(a) above;
- (c) A sum, as overhead and profit on Paragraph 12.4.5(1)(a) above, determined by the Owner to be fair and reasonable;
- (d) The reasonable cost of the preservation and protection of property incurred pursuant to

- Paragraph 12.4.2(9); and any other reasonable cost incidental to termination of Work under this Contract, including expenses incidental to the determination of the amount due to the Contractor as the result of the termination of Work under this Contract.
- (2) The total sum to be paid to the Contractor under Paragraph 12.4.5(1) above shall not exceed the total Contract Sum as reduced by the amount of payments otherwise made and as further reduced by the Contract price of work not terminated. Except for normal spoilage, and except to the extent that the Owner shall have otherwise expressly assumed the risk of loss in writing, there shall be excluded from the amounts payable to the Contractor under Paragraph 12.4.5(1) above, the fair value of property which is destroyed, lost, stolen, or damaged so as to become undeliverable to the Owner, or to a buyer pursuant to Paragraph 12.4.2(7).
- **12.4.6 DEDUCTIONS**. In arriving at the amount due the Contractor under this Article 12.4, there shall be deducted:
- (1) All unliquidated advance or other payments on account theretofore made to the Contractor, applicable to the terminated portion of this Contract;
- (2) Any Claim which the Owner may have against the Contractor in connection with this Contract; and
- (3) The agreed price for, or the proceeds of sale of, any materials, supplies, or other things acquired by the Contractor or sold, pursuant to the provisions of this Article 13.4, and not otherwise recovered by or credited to the Owner.
- **12.4.7 PARTIAL TERMINATION**. If the termination is partial, the Contractor may file with the Owner a PRE for the amounts specified in the Contract relating to the continued portion of the Contract and such equitable adjustment

as may be agreed upon shall be made in such amounts. Any PRE under this Paragraph 12.4.7 must be filed within twenty-one (21) days from the effective date of the notice of termination.

12.4.8 PARTIAL PAYMENTS. The Owner may, from time to time, under such terms and conditions as it may prescribe, make partial payments and payments on account against costs incurred by the Contractor in connection with the terminated portion of this Contract whenever, in the opinion of the Owner the aggregate of such payments shall be within the amount to which the Contractor will be entitled hereunder. If the total of such payments is in excess of the amount finally agreed or determined to be due under this Article 12.4. such excess shall be payable by the Contractor to the Owner upon demand, together with interest at a rate equal to the average rate at the time being received from the investment of local funds, as determined by the City finance representative, for the period until the date such excess is repaid to the Owner; provided, however, that no interest shall be charged with respect to any such excess payment attributable to a reduction in the Contractor's claim by reason of retention or other disposition of termination inventory until ten (10) days after the date of such retention or disposition, or such later date as determined by the Owner by reason of the circumstances.

12.4.9 PRESERVE AND MAKE AVAILABLE **RECORDS**. Unless otherwise provided for in this Contract, or by applicable law, the Contractor shall, from the effective date of termination until the expiration of three years after final settlement under this Contract, preserve and make available to the Owner at all reasonable times at the office of the Contractor, but without direct charge to the Owner, all books, records, documents and other evidence bearing on the costs and expenses of the Contractor under this Contract and relating to the Work terminated hereunder, or, to the extent approved by the Owner Representative, photographs,

micrographs, or other authentic reproductions thereof.

12.5 OWNER'S RIGHT TO STOP THE WORK. If the Contractor fails to correct Work or fails to carry our Work, as required by the Contract Documents or fails to comply with all required and customary safety precautions; the Owner, by written order signed personally or by an agent specifically so empowered by the Owner in writing, may order the Contractor to stop the Work, or any portion thereof, until the cause for such order has been eliminated; however, the right of the Owner to stop the Work shall not give rise to a duty on the part of the Owner to exercise this right for the benefit of the Contractor or any other person or entity.

ORDINANCE NO. 543

AN ORDINANCE OF BOX ELDER COUNTY AMENDING THE BOX ELDER COUNTY ZONING MAP BY REZONING APPROXIMATELY 8.96 ACRES LOCATED AT APPROXIMATELY 11200 NORTH 3600 WEST FROM A-20 (AGRICULTURE 20 ACRES) TO RR-1 (RURAL RESIDENTIAL 1 ACRE) ZONE.

WHEREAS, the applicant is requesting that the property described herein be zoned from A-20 (Agriculture 20 Acres) to RR-1 (Rural Residential 1 Acre) zone; and

WHEREAS, the Box Elder County Planning Commission scheduled a public hearing on the petition to rezone the property and amend the Box Elder County zoning map and provided notice of the public hearing by mailing notice to each affected property owner and each adjacent property owner, and each affected entity at least 10 calendar days before the public hearing, and by posting it in at least 3 public locations within the county or on the county's official website; and by publishing it in a newspaper of general circulation in the area and on the Utah Public Notice Website at least 10 calendar days before the public hearing; and

WHEREAS, the Box Elder County Planning Commission, after appropriate notice, held a public hearing on October 21, 2021, to allow the general public to comment on this proposed rezone and amendment of the zoning map; and

WHEREAS, after providing for public comment from the general public, the Box Elder County Planning Commission has found and determined that the proposed rezone and amendment to the zoning map is in conformity with the General Plan of Box Elder County, that the uses allowed by the proposed change are harmonious with the overall character of the existing development in the vicinity of the property, that it will not adversely affect adjacent properties, that the facilities and services intended to serve the property to be rezoned are adequate, and will provide for the health, safety, and general welfare of the public and protect the environment; and

WHEREAS, based upon these findings, the Box Elder County Planning Commission has recommended that the Box Elder County Commission amend the zoning map as has been requested; and

WHEREAS, the Box Elder County Commission, after appropriate notice, held a public meeting on January 5, 2022 to review and discuss this proposed amendment; and

WHEREAS, after reviewing and discussing, the Board of County Commissioners of Box Elder County, Utah finds that the amendment to the zoning map as set forth below is in conformity with the General Plan of Box Elder County, that the uses allowed by the proposed change are harmonious with the overall character of the existing development in the vicinity of the property, that it will not adversely affect adjacent properties, that the facilities and services intended to serve the property to be rezoned are adequate, and that it will be in the best interest of and promote the health, safety and general welfare of the residents of Box Elder County;

NOW THEREFORE, the County Legislative body of Box Elder County ordains as follows:

SECTION 1: Zoning Map Amendment. The Zoning Map of Unincorporated Box Elder County is hereby amended by classifying the following described parcels in unincorporated Box Elder County

from A-20 (Agriculture 20 Acres) to RR-1 (Rural Residential 1 Acre) zone:

PARCEL 05-023-0011

THE SOUTH 800 FEET OF THE FOLLOWING DESCRIBED PROPERTY LYING NORTH OF STATE HWY SR 102: BEG AT POINT 66 FT NORTH & 66 FT WEST OF SE CORNER OF SEC 06, T 11N, R 02W, SLM. THENCE NORTH ALONG WEST SIDE OF COUNTY ROAD 790.6 FT; EAST 71 FT; NORTH 1787 FT TO E/4 CORNER OF SD SEC; WEST 1014 FT TO BEAR RIVER; THENCE FOLLOWING EAST BANK OF BEAR RIVER TO A POINT 1476 FT WEST & 790.6 FT NORTH OF POB; S 05*30`00" E 794.3 FT; EAST 1400 FT TO BEG

SECTION 2: Effective Date. This ordinance shall become effective fifteen (15) days after its passage.

18		
PASSED, ADOPTED AND A SYNO	PSIS ORDERED PUBLISHED County Commissioners of Box	this <u>5⁴B</u> day of Elder County, Utah,
Attest: SEAL Marla Young Box Elder County Clerk	Commissioner Summers Commissioner Scott Commissioner Hadfield Box Rider County Commission	Voting Aye Voting Aye Voting Aye The Chair
State of Utah) .ss) County of Box Elder)		
On this _54h day of	d who by me duly sworn (or affill document was signed by him in	rm), did say he is the
My Commission Expires: Oct 15, 203	9)

MARLA YOUNG
Notary Public
State of Utah
Commission Expires October 15, 2025
Commission Number 721029

Notary Public

ORDINANCE NO. 548

AN ORDINANCE OF BOX ELDER COUNTY AMENDING THE BOX ELDER COUNTY ZONING MAP BY REZONING 5.11 ACRES LOCATED AT APPROXIMAETLY 9000 South HWY 89 FROM C-G (GENERAL COMMERCIAL) TO C-E (COMMERCIAL ENTERPRISE) ZONE.

WHEREAS, the applicant is requesting that the property described herein be zoned from C-G (General Commercial) to C-E (Commercial Enterprise) zone; and

WHEREAS, the Box Elder County Planning Commission scheduled a public hearing on the petition to rezone the property and amend the Box Elder County zoning map and provided notice of the public hearing by mailing notice to each affected property owner and each adjacent property owner, and each affected entity at least 10 calendar days before the public hearing, and by posting it in at least 3 public locations within the county or on the county's official website; and by publishing it in a newspaper of general circulation in the area and on the Utah Public Notice Website at least 10 calendar days before the public hearing; and

WHEREAS, the Box Elder County Planning Commission, after appropriate notice, held a public hearing on December 16, 2021, to allow the general public to comment on this proposed rezone and amendment of the zoning map; and

WHEREAS, after providing for public comment from the general public, the Box Elder County Planning Commission has found and determined that the proposed rezone and amendment to the zoning map is in conformity with the General Plan of Box Elder County, that the uses allowed by the proposed change are harmonious with the overall character of the existing development in the vicinity of the property, that it will not adversely affect adjacent properties, that the facilities and services intended to serve the property to be rezoned are adequate, and will provide for the health, safety, and general welfare of the public and protect the environment; and

WHEREAS, based upon these findings, the Box Elder County Planning Commission has recommended that the Box Elder County Commission amend the zoning map as has been requested; and

WHEREAS, the Box Elder County Commission, after appropriate notice, held a public meeting on January 5, 2022 to review and discuss this proposed amendment; and

WHEREAS, after reviewing and discussing, the Board of County Commissioners of Box Elder County, Utah finds that the amendment to the zoning map as set forth below is in conformity with the General Plan of Box Elder County, that the uses allowed by the proposed change are harmonious with the overall character of the existing development in the vicinity of the property, that it will not adversely affect adjacent properties, that the facilities and services intended to serve the property to be rezoned are adequate, and that it will be in the best interest of and promote the health, safety and general welfare of the residents of Box Elder County;

NOW THEREFORE, the County Legislative Body of Box Elder County, ordains as follows:

SECTION 1: Zoning Map Amendment. The Zoning Map of Unincorporated Box Elder County is hereby amended by classifying the following described parcels in unincorporated Box Elder County from C-G (General Commercial) to C-E (Commercial Enterprise) zone:

PARCEL 01-047-0013

SE/4 OF NE/4 OF SEC 14 T07N R02W SLM LYING W OF STATE HWY. EXC OF RES.

PARCEL 01-047-0044

PART OF NE/4 OF SEC 14, T 07N, R 02W, SLBM. BEGINNING AT A POINT WHICH LIES SOUTH 1294.1 FT & EAST 437.1 FT FROM N/4 CORNER OF SAID SEC S 01*08`00" E 27.5 FT, N 84*41'00" W 287.0 FT, N 36*00'00" W 7.0 FT, S 89*03'00" E 289.4 FT TO POB.

SECTION 2: Effective Date. T passage.	This ordinance shall become effective fifteen (15) days after its
Ganuary, 2022, by the Bo	SYNOPSIS ORDERED PUBLISHED thisday of ard of County Commissioners of Box Elder County, Utah,
OLEAK MANAGER	Commissioner Summers Commissioner Scott Commissioner Hadfield Voting Voting Voting
Attest: Million OF OF OF OTHER PORTION OF THE PROPERTY OF THE	Box Elder County Commission - Chair
Marla Young Box Elder County Clerk	
State of Utah) .ss) County of Box Elder)	
notary public, <i>Veff Velott</i> proved on the basis of satisfactory eviden	, 2022, personally appeared before me, the undersigned , whose identity is personally known to me (or nce) and who by me duly sworn (or affirm), did say he is the nd said document was signed by him in behalf of said at said Corporation executed the same.
My Commission Expires: Oct . 15, c	1035
MARLA YOUNG Notary Public State of Utah My Commission Expires Octo Commission Number 7	Notary Public

RESOLUTION NO. 22-01

RESOLUTION OF THE BOX ELDER COUNTY COMMISSION ADOPTING A POLICY TO COMPLY WITH THE OCCUPATIONAL SAFETY AND HEALTH ADMINISTRATION'S (OSHA'S) EMERGENCY TEMPORARY STANDARD (ETS) FOR EMPLOYERS WITH 100 OR MORE EMPLOYEES (29 CFR 1910.501).

WHEREAS Title 29 Chapter XVII Part 1970 Subpart U of the United States Code of Federal Regulations (29 CFR 1910.501) requires employers of 100 employees or more to adopt a policy to comply with the provisions of this code; and

WHEREAS, Box Elder County employs over 100 people; and

WHEREAS, Box Elder County is required to comply with federal and state OSHA rules; and

WHEREAS, Box Elder County Commission feels that the OSHA COVID-19 ETS is an example of gross overreach by the federal government, leaving the Commission with no choice but to unwillingly comply to avoid possible penalties for non-compliance; and

WHEREAS, Box Elder County Commission has determined that it is in the best interests of the residents of Box Elder County and will promote the safety, health, morals and general welfare of the employees, employees' families, customers, visitors, and residents of Box Elder County to comply by implementing this mandated policy.

NOW THEREFORE, BE IT RESOLVED that the Box Elder County adopts a personnel policy complying with OSHA's COVID-19 ETS as follows:

19-6. COVID-19 ETS Vaccination and Testing

All Box Elder County employees are required to be either fully vaccinated against COVID-19 or to submit a weekly negative COVID-19 test result. Effective January 10, 2022, all Box Elder County employees who are not fully vaccinated against COVID-19 must wear a face covering while at work. Employees hired after January 10, 2022, are required to be fully vaccinated by their first day of work or submit a negative test result dated no more than three days prior to their first day of work, with subsequent weekly testing as described below.

Exception: As more fully explained before, this policy does not apply to employees who do not report to a workplace where other individuals (such as coworkers or customers) are present; or employees while working from home.

A. Employees have two options under this policy:

1. Vaccination

- a. An employee is considered fully vaccinated two weeks after receiving the required number of doses of a COVID-19 vaccine.
- b. Employees who choose to receive the COVID-19 vaccine will need to receive the last dose of their vaccine no later than December 27, 2021, to meet the January 10, 2022, deadline to be fully vaccinated.
- c. COVID-19 vaccinations are free, whether an individual has health insurance or not. While a provider may bill a patient's health insurance for administering the vaccine, there is no out-of-pocket cost to an individual.
- d. An employee may take up to four hours of ETS time while on duty per dose to travel to the vaccination site, receive a vaccination, and return to work. This would mean a maximum of eight hours of ETS time for employees receiving two doses while on duty. If an employee spends less time getting the vaccine, only the necessary amount of duty time will be granted. If an employee is vaccinated outside of their approved duty time they will not be compensated.
- e. Employees may utilize up to two workdays of sick leave immediately following each dose if they have side effects from the COVID-19 vaccination that prevent them from working. Employees who have no sick leave will be granted up to two days of ETS leave immediately following each dose if necessary. Employees are to work with their managers to schedule necessary time off.
- f. Official documentation of vaccination status must be provided to human resources by January 9, 2022, and must include the following:
 - i. The type of vaccine administered.
 - ii. The date(s) of administration.
 - iii. The name of the health care professional or clinic site administering the vaccine.
- g. Acceptable proof of vaccination status is:
 - i. The record of immunization from a health care provider or pharmacy;
 - ii. A copy of the COVID-19 Vaccination Record Card;
 - iii. A copy of medical records documenting the vaccination;
 - iv. A copy of immunization records from a public health, state, or tribal immunization information system; or
 - v. A copy of any other official documentation that contains the type of vaccine administered, date(s) of administration, and the name of the health care professional(s) or clinic site(s) administering the vaccine(s).
- h. If an employee is unable to produce one of these acceptable forms of proof of vaccination, despite attempts to do so (e.g., by trying to contact the vaccine administrator or state health department), the employee can provide a signed

and dated statement attesting to their vaccination status (fully vaccinated or partially vaccinated); attesting that they have lost and are otherwise unable to produce one of the other forms of acceptable proof; and including the following language:

- i. "I declare (or certify, verify, or state) that this statement about my vaccination status is true and accurate. I understand that knowingly providing false information regarding my vaccination status on this form may subject me to criminal penalties."
- i. Employees must certify that the documentation they are submitting is true and correct. Employees found to have provided false documentation will be subject to discipline up to and including termination of employment.

2. Weekly Testing and Face Covering

- a. Employees who are not fully vaccinated must inform Human Resources of their vaccination status by January 6, 2022. They must provide a statement indicating whether they are partially, or unvaccinated and not planning to get vaccinated.
- b. Starting February 9, 2022, employees who report to the workplace at least once every seven days;
 - i. Must be tested for COVID-19 at least once every seven days; and
 - ii. Must provide documentation of the most recent COVID-19 test results to Human Resources no later than the seventh day following the date on which the employee last provided a test result.
- c. Any employee who does not report to the workplace during a period of seven or more days (e.g., if they were teleworking for two weeks prior to reporting to the workplace):
 - Must be tested for COVID-19 within seven days prior to returning to the workplace; and
 - ii. Must provide documentation of that test result to Human Resources upon return to the workplace.
- d. If an employee does not provide documentation of a COVID-19 test result as required by this policy, they will be removed from the workplace until they provide a test result.
- e. Employees who have received a positive COVID-19 test, or have been diagnosed with COVID-19 by a licensed healthcare provider, are not required to undergo COVID-19 testing for 90 days following the date of their positive test or diagnosis.
- f. Employees are responsible for obtaining the COVID-19 test. In the event free testing is not available, the County will pay for the test if required by law; otherwise, the employee must pay for the test. At-home testing is not currently permitted.

- g. Acceptable tests include: rapid or PCR.
- h. Employees found to have provided false documentation will be subject to discipline up to and including termination of employment.
- i. Effective January 10, 2022, all employees who are not fully vaccinated against COVID-19 must wear an OSHA-approved face covering while at work, both when indoors and when occupying a vehicle with another person for work purposes.
- j. Exceptions to the face covering requirement are as follows:
 - i. When an employee is alone in a room with floor to ceiling walls and a closed door.
 - ii. For a limited time, while an employee is eating or drinking at the workplace or for identification purposes in compliance with safety and security requirements.
 - iii. When an employee is wearing a respirator of facemask.
 - iv. Where Human Resources has determined that the use of face coverings is infeasible or creates a greater hazard (e.g., when it is important to see the employee's mouth for reasons related to their job duties, or when the use of a face covering presents a risk of serious injury or death to the employee).

B. Employees Who Test Positive for COVID-19

- 1. Employees must promptly notify their supervisor and Human Resources when they have tested positive for COVID-19 or have been diagnosed with COVID-19 by a licensed healthcare provider.
- 2. The supervisor must immediately remove an employee from the workplace if they have received a positive COVID-19 test or have been diagnosed with COVID-19 by a licensed healthcare provider (i.e., immediately send them home or to seek medical care, as appropriate).
- 3. An employee removed from the workplace because they are COVID-19 positive must remain away from the workplace until the employee receives a negative result on a COVID-19 nucleic acid amplification test (NAAT) following a positive result on a COVID-19 antigen test if the employee chooses to seek a NAAT test for confirmatory testing; meets the return to work criteria in CDC's "Isolation Guidance"; or receives a recommendation to return to work from an licensed healthcare provider.
- **4.** Under CDC's "isolation Guidance," asymptomatic employees may return to work once 10 days have passed since the positive test, and symptomatic employees may return to work after all the following are true:
 - a. At least 10 days have passed since symptoms first appeared, and
 - b. At least 24 hours have passed with no fever without fever-reducing medication, and

c. Other symptoms of COVID-19 are improving (loss of taste and smell may persist for weeks or months and need not delay the end of isolation).

C. Reasonable Accommodation

1. Employees in need of an exemption from this policy due to a medical reason, or because of a sincerely held religious belief, or because of any other reason allowed by law, must submit a request for accommodation to the human resources department to begin the interactive accommodation process. All such requests will be handled in accordance with applicable laws and regulations.

D. Violations of this Policy

1. Employees who fail to comply with this policy may be subject to discipline, in accordance with Box Elder Employee Policies Section 5, and in accordance with the law.

E. Confidentiality of Records

1. All medical information collected from individuals, including vaccination information, test results, and any other information obtained as a result of testing, will be treated in accordance with applicable laws and policies on confidentiality and privacy. These records are to destroyed as soon as possible once this policy is no longer required, in accordance with applicable laws related to the ETS and to records control and management.

ADOPTED AND APPROVED by the County Legislative Body of Box Elder County, State of Utah, with 3 members present and 3 members voting in favor thereof in regular session of the Box Elder County Commission this 5th day of January, 2022. This policy shall be effective immediately until such time that OSHA no long requires compliance with Subpart U, or when the law is otherwise made invalid.

ATTEST:

Marla Young

COUNTY CLERK

Teff Scott

COUNTY COMMISSION CHAIRMAN

5 De Koll